# Review Neuroplasticity in Alzheimer's Disease

## Bruce Teter<sup>1\*</sup> and J. Wesson Ashford<sup>2</sup>

<sup>1</sup>Department of Medicine, University of California Los Angeles, California and Veteran's Affairs–Greater Los Angeles Healthcare System, Sepulveda, California

<sup>2</sup>Departments of Psychiatry and Neurology, and the Sanders-Brown Center on Aging, University of Kentucky, Lexington, Kentucky and Veteran's Affairs Medical Center, Lexington, Kentucky

Index

10.5

Neurogenesis

Ramon y Cajal proclaimed in 1928 that "once development was ended, the founts of growth and regeneration of the axons and dendrites dried up irrevocably. In the adult centers the nerve paths are something fixed, ended and immutable. Everything must die, nothing may be regenerated. It is for the science of the future to change, if possible, this harsh decree." (Ramon y Cajal, 1928). In large part, despite the extensive knowledge gained since then, the latter directive has not yet been achieved by 'modern' science. Although we know now that Ramon y Cajal's observation on CNS plasticity is largely true (for lower brain and primary cortical structures), there are mechanisms for recovery from CNS injury. These mechanisms, however, may contribute to the vulnerability to neurodegenerative disease. They may also be exploited therapeutically to help alleviate the suffering from neurodegenerative conditions. Published 2002 Wiley-Liss, Inc.<sup>+</sup>

#### **Key words:** neuroplasticity; Alzheimer's; genetics; apolipoprotein E; therapeutics; pharmacogenetics

Abbreviations: AB, amyloid ß protein; ADDLs, AB-derived diffusible ligands; AD, Alzheimer's disease; apoE, apolipoprotein E (gene or protein); APP, amyloid precursor protein; ChAT, choline acetyl transferase; CNS, central nervous system; CREB, cAMP response element binding protein; DRG, dorsal root ganglion; E4, E3, apoE isotypes epsilon 4, epsilon 3; EC, entorhinal cortex; ECL, entorhinal cortex lesion; ERT, estrogen replacement therapy; GFAP, glial fibrillary acidic protein; GT1-1, hypothalamic cell line; HC, hippocampus; HDL, high density lipoprotein; HNE, hydroxy-nonenol; IML, inner molecular layer of the hippocampus; ko, gene knockout mice; LDLR, low density lipoprotein receptor; LRP, LDLRrelated protein; LTP, long term potentiation; MAP, microtubule associated protein; NCAM, neural cell adhesion molecule; NF-kB, nuclear factor kappa B; NFT, neurofibrillary tangles; NGF, nerve growth factor; NOS, nitric oxide synthetase; NO, nitric oxide; NSAIDs, nonsteroidal antiinflammatory drugs; NSE, neuron-specific enolase; OML, outer molecular layer; OHSC, organotypic hippocampal slice culture; PNS, peripheral nervous system; PS, presenilin; PSD-95, post-synaptic density protein; TNFa, tumor necrosis factor a; VLDL, very low-density lipoprotein

\*Correspondence to: Bruce Teter, Dept. of Medicine, UCLA, VA Medical Center, MC151, 16111 Plummer St., Sepulveda, CA 91343.

Received 23 May 2002; Revised 4 July 2002; Accepted 8 July 2002

1 Introduction 2.1 Neuroplasticity-An Overview 2.2 Synapses 2.3 Adhesion Molecules 2.4 Glia 2.5 Age 3 Alzheimer's Disease 3.1 Temporal, Spatial Course 32 Development, Differentiation Recapitulation 3.3 Synaptic Loss 3.4 Axonal, Dendritic Remodelling Aberrant Sprouting, Distrophic Neurites 35 3.6 Entorhinal Cortex, Hippocampal Pathway and Lesion Models (ECL, OHSC) 3.7 Cholesterol in the CNS and in AD 4.1 Apolipoprotein E (apoE) ApoE-dependent Sprouting 4.2 4.3 Alzheimer's Disease 4.4 Sprouting Mechanisms and the Lipid Metabolism Model 4.5 Model Systems 4.6 ApoE-knockout Mice 4.7 Human ApoE Isotype Transgenics 4.8 In vitro Sprouting systems 4.9 ApoE Isotype-dependent Granule Cell Mossy Fiber Sprouting. 4.10 ApoE4 Gain-of-function Defect in Sprouting 4 1 1 ApoE, Gender, Estrogen 4.12 ApoE Therapeutic Implications: Drug Interactions and Pharmacogenetics 5.1 APP and AB 5.2 APP Trophic Effects and Axonal Transport 5.3 APP Processing Balance 5.4 Amyloid-beta (AB) and Amyloid Plaques 6 Tau 7 Presenilins Neurodegeneration, Neuroregeneration Interactions 8 8.1 Degeneration-Regeneration Cross Talk and Combinatorial Signalling 8.2 Cell Cycle 8.3 Nitric Oxide (NO) 9.1 Gender and Estrogen Estrogen Replacement Therapy (ERT) 9.2 9.3 Estrogen/Plasticity 10.1 Treatments for Plasticity 10.2 Oxidation 10.3 Inflammation/NF-kB 10.4 Growth Factors

#### 1. INTRODUCTION

Alzheimer's disease (AD) displays aspects of mechanisms related to all the major theories of aging: mitochondrial decline in energy production, deregulation of calcium homeostasis, ROS generation and accumulation of its damage products, immune/inflammation dysfunction, hormone deregulation, and loss of regenerative ability (Brewer, 2000). The information storage defect in AD is represented at all levels of systems functions: biological, psychological and sociological (Ashford et al., 1998a). All these factors and levels can be traced to basic mechanisms of memory storage and retrieval. The contribution of neuroplasticity to AD, as a compensatory response or a fundamental defect, is gaining recognition, from the original recognition of the implications of dystrophic neurites by Alzheimer and others to more recent evidence of plasticity at many levels (Fischer, 1907; Simchowicz, 1911; Scheibel and Tomiyasu, 1978; Scheibel, 1982). That AD is a fundamental defect in such mechanisms was first proposed in 1985 (Ashford and Jarvik, 1985) and has been recently reviewed (Neill, 1995; Mesulam, 2000; Arendt, 2001a,b). This common downstream target can explain how numerous genes and factors cause the same clinical and neuropathological phenotype.

AD is characterized by ongoing neurodegeneration, yet in AD and in normal aging neuronal loss is not a prerequisite for functional deficits (reviewed in Morrison and Hof, 1997; Mrak et al., 1997). Synaptic pathology is an early marker of both AD and aging (Greenough et al., 1978; Agnati et al., 1992; Martin et al., 1994). Is AD an inevitable consequence of aging-related processes, simply a faster deterioration of the capacity for plasticity? Even 'normal aging' can change its course at some point: enhanced dendritic growth in early aging (70s) is followed by regression of dendritic arbor in the oldest old (90s) (Flood et al., 1985). Plasticity in AD may be a process of compensatory, albeit futile sprouting in vulnerable neurons. In this scheme, mechanisms of plasticity and their physiological burden are overstimulated in AD, leading to secondary neurodegenerative effects, which then feed a vicious cycle of increasing plasticity burden (Mesulam, 2000; Joseph et al., 2001). The increasing burden of plasticity is initially an adaptive response that also includes upregulation of  $\tau$ phosphorylation and APP turnover, with subsequent formation of neurofibrillary tangles (NFT) and amyloid plaques as consequences that eventually lead to neurodegenerative events including loss of synapses, axons, and dendrites, and eventually cell death (Mesulam, 2000). The two pathologic hallmarks of AD, neuritic plaques and NFT, could be both causative in memory deficits and result from more fundamental failures of memory, where positive feedback in vicious cycles could feed initially minor disturbances (Geddes et al., 1985). AD synaptic degeneration can also be viewed as an adaptive 'rescue program' in response to metabolic fuel deprivation, by pruning of the axonal tree to reduce energy-consuming neuronal activity, as suggested by the decrease in synaptic metabolic activity with age and in AD (Heininger, 2000).

The vulnerability of neurons to the effects of such plasticity-elicited degeneration reflects their capacity for plasticity. A simplistic analogy is found in cancer, where cells with a predilection to divide are the most vulnerable to failure of mitogenic control. Failure of neuroplasticity ultimately unleashes the onset of clinical AD symptomology by disrupting the balance between degenerative and regenerative processes (reviewed in Mesulam, 2000; Arendt, 2001a).

It is remarkable that all genetic causes and risk factors of AD can impinge on neuroplasticity. Instead of causing AD, these genetic mutations can be viewed as interacting with ongoing, age-related impaired plasticity activity to accelerate the events that lead to its failure. Alternatively, they could initiate stress-related repair mechanisms that fail because of downstream defects in or blocks to plasticity. Are genetic factors in AD progeroid genes? For example, ApoE4 is associated with decreased longevity compared to E2 (Corder et al., 1994). What do genetic mutations tell about the distinction between AD and 'normal' aging? Do genetic mutations decrease the natural activity of the wildtype protein, or are they gain-of-function mutations that create altogether new activities? Only parallel analysis of wild-type and mutant forms address such questions. In addition, most of the genes and factors discussed here are pleiotropic and interact at various levels. Such interactions create many secondary, indirect effects that exponentially expand the complexity of AD etiology, and full coverage of these is beyond the scope of this review (extensively reviewed in Arendt, 2001a). Further, because most factors also show effects in neurodegeneration, the interactive relationship between neurodegeneration and the capacity for neuroplasticity adds yet more to this complexity (see Section 8).

#### 2.1 NEUROPLASTICITY: AN OVERVIEW

Neuroplasticity is both a substrate of learning and memory and a mediator of responses to neuronal attrition and injury (compensatory plasticity). It is a continuous process in reaction to neuronal activity and neuron injury, death, and genesis, which involves modulation of structural and functional processes of axons, dendrites, and synapses. The varied structural elements that embody plasticity include LTP, synaptic efficacy, synaptic remodeling, synaptogenesis, neurite extension including axonal sprouting and dendritic remodeling, and neurogenesis and recruitment. In a broader sense, phenomenological processes that manifest plasticity are: synapses (electrical, biochemical, structural), neurite (axon, dendrite), neuron cell bodies, anterograde (toward distal neurites) and retrograde (from distal neurites) transport, cell interactions (neuronglia), neural networks, and behavioral, psychological, and sociological activities.

The rules of synaptic strengthening postulated by Hebb (1949), which require a concerted activation of preand postsynaptic elements (see Sections 2.2, 8), subserve the phenomenon of LTP as a model of memory formation, and which is also associated with synapse dynamics including formation and removal of synapses and changes in synapse morphology (Chang and Greenough, 1984; Toni et al., 1999; Martin et al., 2000) (see Section 2.2). Signals of plasticity include intraneuronal (anterograde and retrograde), interneuronal (transsynaptic and extra/ parasynaptic) as well as intercellular signaling through glia (Cotman and Nieto-Sampedro, 1984; Neill, 1995). They include many molecules in the following families: extracellular matrix molecules, semaphorins/collapsins, immunoglobulins, myelin-associated inhibitors, tyrosine kinase receptors, netrins, neurotrophic factors, growth factors, inflammatory cytokines, and neurotransmitters; furthermore, many inhibitory molecules also come from the same classes (reviewed in Horner and Gage, 2000). Many mutant and transgenic mice have helped elucidate aspects of plasticity (reviewed in Chen and Tonegawa, 1997).

The adult central nervous system responds to injury with limited yet sometimes effective restoration of synaptic circuitry. Whether compensatory growth is widespread and whether it reverses cognitive deficits has been debated (Cotman et al., 1991; Poirier, 1994; Masliah et al., 1995b). Functional recovery requires that reactive synaptogenesis not exacerbate circuitry dysfunction, as has been proposed (Cotman et al., 1991; Masliah et al., 1991c). If reactive plasticity leads to aberrant misconnection by innervating the wrong target, there may be intrinsic, inhibitory or limiting mechanisms to attenuate such misguided synaptogenesis. Clearly, brain self-organization continuously balances synapse formation and removal as well as neurite sprouting and retraction, and in some conditions, inhibition of sprouting may actually be protective by sequestering dysfunctional neurons. Such inhibition of distal plasticity events could signal plasticity-related events in the perikaryon (Mesulam, 2000). Chronic stimulation, however, may become unsustainable resulting in a plasticity 'burden' that leads to degenerative events.

#### 2.2 Synapses

The balance between dynamic stabilization and destabilization of synapses may provide the basis for failure of plasticity with age and disease. Aspects of LTP are mediated by rapid generation of new spines, presumably guided by actin-mediated shape changes (Engert and Bonhoeffer, 1999; Maletic-Savatic et al., 1999; reviewed in Luscher et al., 2000). The shape of dendrites, as well as cell survival, can be modified by neurotrophins (McAllister et al., 1999; reviewed in Huang and Reichardt, 2001). The cytoskeleton also mediates aspects of signal transduction, as shown by microtubule involvement with effector molecules in the hedgehog, Wnt, JNK, and ERK pathways (reviewed in Gundersen and Cook, 1999). Actin controls the generation and motility of growth cones, spines and dendrites. F-actin assembly at the leading edge of growth cones is regulated by many factors, especially those of the substrate (Suter and Forscher, 1998; Hynes, 1999) and by small receptor-activated GTPases including rac, rho and Cdc42 (Lanier and Gertler, 2000). Dendritic spines are enriched in actin (reviewed in Matus, 1999). Not only are they highly motile structures covered with presynaptic structures, they may coordinate with the postsynaptic complex,

moving together, mechanically stabilizing the synapse (Barres and Smith, 2001). Synapse formation during development may be a collaborative process involving growth of a presynaptic element on a site where a postsynaptic spine is either present or ready to form (Horner, 1993) (see Section 9.1). Further, the cadherin/catenin systems play an important role in the recognition between presynaptic growth cones and its postsynaptic dendritic target (Brose, 1999). Subsequent actions of immediate early genes like Narp, Arc, and synaptotagmin recruit and localize synaptic protein components. Arc stimulates both activity and plasticity of synapses and is modulated by the insulin receptor signal cascade. The cellular sorting, directional transport, and specific accumulation of axonal and dendritic components (including certain mRNAs) (Schuman, 1999; Winckler et al., 1999; Wells et al., 2000) are affected by AD-related pathology like NFTs (see Section 6) and APP (see Section 5.2).

Interestingly, mRNAs for GAP-43 and Arc have been found in growth cones, and NR1 and Arc in dendrites, implicating the important need for their activities at these sites and their synapse-specific regulation (Crino and Eberwine, 1996; Gazzaley et al., 1997; reviewed in Huang, 1999; Martin et al., 2000; Campenot and Eng, 2000; Steward and Schuman, 2001). Translationdependent synapse formation can occur even in the absence of cell bodies (Schacher and Wu, 2002).

Presynaptic markers include GAP-43, SNAP25, syntaxin, synaptotagmin, synaptoporin, synaptophysin, and the synapsins. GAP-43 is highly expressed in neural development, axon regeneration and neuritic sprouting (Neve et al., 1988; Masliah et al., 1991a ; de la Monte et al., 1995; Benowitz and Routtenberg, 1997). Postsynaptic markers include MAP-2, PSD-95, NR1, spinophillin, and dendritic actin (reviewed in McEwen, 2001).

## 2.3 Adhesion Molecules

Optimal cell adhesion is required for synaptic plasticity (Schubert, 1991). Presynaptic differentiation is triggered by molecules associated with the synaptic basal lamina (reviewed in McGowan and Marinkovich, 2000). Adhesion molecules also communicate directly with signaling cascades regulating cell proliferation and differentiation, like FAK and MAP cascade, which are also implicated in AD (Shirazi and Wood, 1993; Zhang et al., 1994; Gartner et al., 1999). L1 and PSA-NCAM are associated with regenerating hippocampal axons (Aubert et al., 1998; Seki and Rutishauser, 1998; Ronn et al., 1999; Weidner et al., 1999). NCAM-I, a marker of plasticity (Ronn et al., 1998), is increased in hippocampal regions, but in a disorganized way in more AD-affected hippocampal areas (Mikkonen et al., 1999). Proteolytic disassembly of the extracellular matrix is regulated by MMP-9 during dendritic remodeling in the adult hippocampus (Szklarczyk et al., 2002). Laminin stimulates neurite outgrowth (Baron-Van Evercooren et al., 1982), is reorganized with estradiol-induced neurite outgrowth (Rozovsky et al., 2002), is found around plaques in AD brain (McKee et al., 1991; Murtomaki et al., 1992), and its mRNA and protein

405

are elevated in AD brain. Laminin interacts with many factors and systems reviewed here (see Sections 3.3, 3.4, 5.3, 5.4, 6, 8, 9.3).

## 2.4 Glia: Astrocytes and Microglia

Astrocytes and microglia play critical roles in CNS response to and recovery from injury (Gage et al., 1988; Frederickson, 1992; Norenberg, 1994; Chao et al., 1996; Bechmann and Nitsch, 1997; Rabchevsky, 2002). Astrocytes have been shown to play important roles in nutrient supply, waste removal, and axonal guidance. More recent work reveals that astrocytes play a more active role in neuronal activity, including regulating ion flux currents, energy production, neurotransmitter release, and synaptogenesis. The latter includes the activity of glia cell apposition to synapses and the regulation of synapse elimination by ensheathment (known as glial swelling) (reviewed in Laming et al., 2000). Ultrastructurally, this is seen as close apposition of GFAP-positive processes (astrocyte end-feet) that undergo rearrangement associated with changes in GFAP expression and localization. This has been observed not only in the hypothalamus during estrus cycledependent synaptogenesis, but also in hippocampus and visual cortex, and may mediate the astrocyte control of synapse number in the developing cerebellum (Lino et al., 2001). Age-related increases in GFAP as an astrocyte activation marker, involved in astrocytic morphologic changes in responses to injury and stress (Nichols et al., 1993; David et al., 1997), may adversely affect their support of synaptogenesis (Vernadakis, 1996); indeed, repression of GFAP is associated with estradiol-induced neurite outgrowth (Rozovsky et al., 2002). Astrocytes can couple directly to neurons and directly regulate synaptic activity (Alvarez-Maubecin et al., 2000). Neurons signal to astrocyte through neuronally-derived glial growth factors (GGF) (Verdi et al., 1996). Glia (astrocytes, microglia and oligodendrocytes) secrete growth-promoting factors like neurotrophins (NT-3) and cytokines, and show agedependent changes in this activity (Sievers et al., 1995; reviewed in Goldberg and Barres, 2000). Other possible mediators and modulators include S100b (Whitaker-Azmitia et al., 1997), taurine, PS-NCAM, tenascin, NT-3, and cytokines. Glia mediate many effects of estrogen on plasticity (Garcia-Segura et al., 2001) (see Section 9), and are major producers of apoE lipoprotein particles (see Section 2.4, 4).

Glia also play roles in failure of plasticity (reviewed in Lemke, 2001). When activated, microglia and astrocytes secrete potent inhibitors of neurite outgrowth (Snow et al., 1990; McKeon et al., 1991; Canning et al., 1996). White matter actively inhibits axon outgrowth through secretion of inhibitory proteins like myelin protein IN-1, proteoglycans, semaphorins and slit proteins; however, responsiveness of neurons to this kind of inhibition can depend on their ability to survive (Davies et al., 1997) (see Section 8). These may contribute to or mimic the astrocyte-induced physiological 'stop' signal to growth cone progression (Reier et al., 1983; Liuzzi and Lasek, 1987).

## 2.5 Age

Age diminishes many aspects of plasticity including LTP induction and maintenance, compensatory synaptogenesis after injury, and reactive synaptogenesis in response to complex experience (Scheff et al., 1980; Mori, 1993; Lanahan et al., 1997). The interplay of many factors contributes to decreased synaptoplastic potential in the aging brain, with resulting delay of axonal sprouting and less effective formation of new connections to replace those lost (McWilliams and Lynch, 1984; Anderson et al., 1986). The capacity of neurons to elaborate neurites is reduced with age but is not lost completely (Brewer, 2000; Brewer et al., 2001). The failure of granule cell axon sprouting is inherent in the age of the sprouting neuron, not the age of its targets (Li et al., 1995).  $Ca^{2+}$  homeostasis is disrupted with aging and can contribute to disrupted neuronal plasticity (Mattson et al., 1992; Teyler et al., 1994; Ghosh and Greenberg, 1995; Foster and Norris, 1997; O'Neill et al., 2001).

Is this age-related loss in plasticity capacity due to reduced intrinsic neuronal capacity, reduced stimulation, or increased inhibition, which can be the same as reduced stimulation in terms of neuronal permissiveness or responsiveness (Tuttle and O'Leary, 1998)? It seems all three are involved, and future experimental directions will focus on determining whether boosting the extrinsic signaling can ameliorate the reduction in intrinsic growth ability (Aubert et al., 1995; Neumann and Woolf, 1999; Cai et al., 1999; reviewed in Goldberg and Barres, 2000). The decreased capacity for plasticity with age might represent a continuous process of which AD is an inevitable endpoint, although there are many differences between normal aging and AD that support AD as a partly age-independent disease (see Section 1).

## 3. ALZHEIMER'S DISEASE

## 3.1 Temporal and Spatial Course

AD pathology progresses over a typical spatial and temporal course of events, with the sequential involvement of basal forebrain, entorhinal cortex, hippocampus, amygdala, and association cortices (Braak and Braak, 1991, 1997). This sequence of events can be understood from a perspective of the functional network through which these areas associate. AD-vulnerable regions like hippocampus and amygdala are related by ancient projections of the olfactory bulb. The entorhinal cortex sits at the evolutionary crossroads between the highly plastic olfactory system, with its distributed representation of information, and the archi-cortex (hippocampus), paleocortex (amygdala), and neocortex (see Section 3.6). These evolutionary relationships may underlie the neural network of initiating and propagating processes in AD (Ashford et al., 1998a).

AD pathology affects CNS regions involved in higher brain functions that are synaptically (structurally and functionally) plastic, and involved in acquisition of new epigenetic information. The limbic system has perhaps the highest potential for neuroplasticity compared to other parts of the cerebral cortex (indicated by high level expression of GAP-43, particularly in the entorhinalhippocampal pathway (see Section 3.6)) (Neve et al., 1988; Lin et al., 1992). Plasticity-related dendritic remodeling (length and branching) is most extensive in limbic and paralimbic regions (entorhinal-hippocampal), less in association cortices, and undetectable in primary sensory and motor areas (Arendt et al., 1998a). The lifelong increased neuroplasticity burden and chronic upregulation of plasticity-related cellular activities of the limbic system could increase its vulnerability to NFT formation. Degeneration in limbic structures could then spread to adjacent limbic and paralimbic neurons in reciprocally connected association cortices to increase their plasticity burden. This would induce reactive synaptogenesis to replace the synapses provided originally by the degenerating axons of NFT-bearing neurons and induce dendritic remodeling to receive synapses once associated with the dendritic trees of adjacent degenerating neurons. If these reactive neurons cannot respond to the challenge of this increased plasticity burden due to barriers to plasticity, they too might then be subjected to similar  $\tau$  events and subsequent NFT formation with cytoskeletal disruption.

In AD there is extensive loss of cholinergic input into the hippocampus (reviewed in Francis et al., 1999). Cholinergic neurotransmission plays an essential role in reactive and experience-induced synaptic reorganization (Baskerville et al., 1997; Kilgard and Merzenich, 1998; Zhu and Waite, 1998), and induces production of neurotrophic secreted APP (Nitsch et al., 1992) (see Sections 5.3, 10.3). Cortical cholinergic depletion in AD (Geula and Mesulam, 1999) arises from loss of neurons that project from nucleus basalis of Meynert, a limbic structure that retains high plasticity in late adulthood (Arendt et al., 1995) and contains some of the first neurons to show NFT pathology (Mesulam, 1996).

## 3.2 Development, Differentiation Recapitulation

Mechanisms of plasticity in adults overlap those used in brain maturation in early development (Cotman et al., 1990; Eriksson et al., 1998; Wheal et al., 1998). Regions with the highest degree of structural plasticity are those that take the longest to mature during childhood (Braak and Braak, 1996) and are the same regions most vulnerable in AD (reviewed in Arendt, 2001b). Although many regions undergo critical periods of intense plasticity, many become relatively quiescent at maturity and the regions that retain high levels of plasticity correlate with AD vulnerability (Ashford et al., 1995, 2000; Alexander et al., 2002). This may allow for the evolutionary acquisition of higher brain functions; regions vulnerable in AD share a common evolutionary foundation in the massive enlargement of the association cortices and functionally linked regions (Rapoport, 1990; Neill, 1995) (see Section 3.1).

The differential susceptibility of AD-specific regions and neurons may be related to the degree of retained capacity for plastic remodeling (Arendt et al., 1998a). In vivo, synaptogenesis rates decline with developmental age (Gall et al., 1979) and there is recapitulation of developmental gene expression responses in adult lesion and aging, including AD (Kondo et al., 1996; Styren et al., 1999). The *Nun Study* indicates that the risk of AD can be determined as early as 20years of age, implicating genetic and developmental factors (Ashford and Mortimer, 2002). If such mechanisms controlling developmental plasticity are defective and are later reactivated (in aging, or AD, or pre-AD), they would contribute to ineffective plasticity responses and exacerbate the plasticity burden of aging and AD.

It has been hypothesized that a 'labile state of differentiation' of neurons allows for neuroplasticity after development but also renders these neurons vulnerable to degenerative effects (Arendt, 2000, 2001a,b). In AD, the differentiation control may be in some way disrupted, involving expression or re-expression of genes (dedifferentiation) that contribute to making new neuronal connections in regenerative plasticity, i.e., genes involved in both growth cones and synaptic connections (Pfenninger et al., 1991) (see Sections 3.1, 3.3), as a necessary component of the ability to maintain a high degree of plasticity throughout life. This retention of plasticity potential leads to or may require re-expression of developmentally regulated genes, alteration of posttranslational modifications and imbalance of gene products, and re-activation of cell cycle genes such as cyclin B and E, as observed in neurons in healthy, elderly individuals (Nagy et al., 1997; Smith et al., 1999) and in phospho-7-expressing neurons (Nagy et al., 1997). This confounds irreversible block of entry into the cell cycle of the neuron, a situation that may trigger cell death (Heintz, 1993) (see Sections 8.2, 8.3). For example, developmentally regulated genes like MAP1B-P, involved in axon growth (Gordon-Weeks and Fischer, 2000; Mack et al., 2000), are downregulated postnatally but remains active in regions of plasticity (Nothias et al., 1996). Its distribution parallels PSA-NCAM, involved in neurite growth and synaptogenesis (Seki and Arai, 1993; Muller et al., 1996; Cremer et al., 1997). The capacities for plasticity may depend on specific kinases, high levels of neurofilaments, and  $\tau$  isoforms (Myoken et al., 1990; Hof and Morrison, 1994; Bahr and Vicente, 1998; Delacourte et al., 1998; Esclaire et al., 1998; Morrison et al., 1998), some of which also mark neurons destined for degeneration in AD (see Section 8).

## 3.3 Synaptic Loss

Synaptic loss is an early event in AD and is a structural correlate of cognitive dysfunction (Gonatas et al., 1967; Gibson, 1983; Davies et al., 1987; Bertoni-Freddari et al., 1989; Hamos et al., 1989; Scheff et al., 1990; Weiler et al., 1990; Brunelli et al., 1991; Terry et al., 1991; Honer et al., 1992; Lassmann et al., 1992, 1993; Zhan et al., 1993; Martin et al., 1994; Masliah et al., 1994, 1995a; Dickson et al., 1995; Heinonen et al., 1995; DeKosky et al., 1996; Sze et al., 1997; Cotman and Anderson, 2000; Mattson et al., 2001; reviewed in Arendt, 2001a). Memory loss in AD may result from synaptic dysfunction that precedes largescale neurodegeneration, where the synapse-to-neuron ratio is decreased by about 50% (Cullen et al., 1997; Lambert et al., 1998; Chapman et al., 1999; Hsia et al., 1999; Chen G et al., 2000; Tezuka et al., 2001; reviewed in Arendt, 2001a). Synapse and dendrite loss in AD exceeds that seen with normal aging (reviewed in Terry et al., 1994; Anderton et al., 1998). Synaptic degeneration, like early AD, progresses slowly at first, perhaps reflecting attempts for compensatory plasticity, and as such could be initially reversible, but eventually becomes irreversible due to marked synapse loss (Rapoport, 1999).

In early AD, a number of growth-associated proteins are upregulated, which may reflect attempts to stimulate plasticity, including GAP-43, MARCKS, spectrin, heparansulfate, laminin (see Sections 2.3, 6), NCAM, various cytokines and neurotrophic factors including NGF (see Section 10.4), bFGF, EGF, IL-1, Il-2, IL-6, IGF-1, IGF-2, PDGF, HGF/SF, and several growth factor receptors (reviewed in Arendt, 2001b: see Section 3.2). Deregulation of proteins involved in structural plasticity of axons and dendrites (Jorgensen et al., 1990; Hatanpaa et al., 1999; Lubec et al., 1999; Mikkonen et al., 1999) and computational studies (Horn et al., 1996; Hasselmo, 1997) indicate a failure of plasticity mechanisms, and support a disruption of synapse turnover as a primary mechanism in AD (Arendt, 2001b) (see Section 3.2). For example, synapsin IIa mRNA is downregulated in early AD, as detected by gene chip microarray analysis (Ho et al., 2001; Pasinetti, 2001). Synaptic remodeling in AD brain is detected also by elevation in the NCAM/SNAP-25 ratio (Jorgensen et al., 1990, 1997; Jorgensen, 1993, 1995). Although these structural and biochemical changes in AD provide understanding of aspects of plasticity, their relation to the properties of LTP in relation to AD are poorly understood (Dawson et al., 1992; Farooqui and Horrocks, 1994; Nalbantoglu et al., 1997).

## 3.4 Axonal and Dendritic Remodeling

Extracts of AD brain increase axonal branching of neurons grown on laminin (Kittur et al., 1992; Jorgensen, 1993), and AD brain and CSF extracts sustain neuron growth and survival (Uchida et al., 1988; Uchida and Tomonaga, 1989; Pauwels et al., 1993; Erickson et al., 1994). Axonal and dendritic remodeling in AD show restricted regional and temporal localization (Arendt et al., 1995, 1997, 1998a). Neocortex and hippocampus exhibit increased sprouting and synaptogenesis in AD (Grady et al., 1989; Masliah et al., 1991a,c; Jobst et al., 1994). Sprouting of commissural and associational fiber axons in AD is indicated by expansion of kainic acid receptor distribution that matches that seen in entorhinal cortex lesions (see Section 3.6); hippocampal sprouting of septal afferents is indicated by the pattern of AchE innervation in the perforant path terminal zone (Geddes et al., 1985; Gertz et al., 1987; Hyman et al., 1987; Masliah et al., 1991c) (see Section 3.6). In AD, axon length correlates with dementia severity suggesting regressive axonal events may be more relevant than dendritic attrition or neuron loss (Anderson, 1996). This is consistent with degeneration of presynaptic termini that then leads to secondary

transneuronal degeneration of postsynaptic dendrites (Su et al., 1997).

Dendritic extent in the hippocampus can increase with age itself, possibly a compensatory response to loss of synaptic connections (Flood and Coleman, 1990). This may not be sustainable, however, because enhanced dendritic growth in the early aging (70s) is followed by regression of dendritic arbor in the oldest old (90s) (Flood et al., 1985). Neocortex and hippocampus in AD also show massive somatodendritic sprouting (Ihara, 1988; Jorgensen et al., 1997), which may reflect unsuccessful remodeling in response to presynaptic or axonal damage (Scott, 1993). Such somatodendritic sprouts, which have filopodium-like structures resembling growth cones, contain  $\tau$  and MAP2, which recapitulates their codistribution in neurite sprouting during development (reviewed in Arendt, 2001a). These dendritic changes therefore may be secondary to deafferentation, signal transduction failures, or cytoskeletal abnormalities (Anderton et al., 1998). As in aging, dendritic sprouting in AD may not be sustainable, as dendritic extent can decline (Flood and Coleman, 1990), particularly dendrites of hippocampal granule cells (Einstein et al., 1994). Some neuropil threads (curly fibers) show preferential development at dendritic branch points, suggesting that blocking dendritic transport could lead to dendrite pruning, and the loss of associated synapses (Ashford et al., 1998b).

## 3.5 Aberrant Sprouting and Dystrophic Neurites as Dendritic Sprouting

Neuronal sprouting in AD can be aberrant based on its localization, morphology, cytoskeletal composition (Arendt et al., 1986, 1998b; Arendt and Zvegintseva, 1987; McKee et al., 1989; Ferrer et al., 1990; Phinney et al., 1999), and synaptic protein expression (Geddes et al., 1985; Ihara, 1988). Aberrant sprouting can be an early feature of AD (Ihara, 1988), preceding detectable tangle formation and extensive neuron loss (Su et al., 1993; Arendt et al., 1998b), and therefore might represent a fundamental defect in AD rather than an overt response to ongoing degeneration (Geddes et al., 1991; Cotman et al., 1993; Masliah et al., 1993a,b). Abnormal neurite growth might be associated with the elevation of NGF receptors (Ernfors et al., 1990; Mufson and Kordower, 1992) that precedes neurofibrillary degeneration (Arendt, 1993). APP transgenic mice also show behavioral and synaptic changes before plaque formation (Holcomb et al., 1998; Hsia et al., 1999; Moechars et al., 1999; Chen G et al., 2000) (see Section 5). Transgenic mice expressing APP show increased hippocampal synaptophysin that correlates with impaired learning and memory (King and Arendash, 2002).

Dystrophic neurites (mainly dendritic) within or near plaques (Gonatas et al., 1967; Probst et al., 1983; Benzing et al., 1993; Su et al., 1993), as a consistent component of AD pathology, were originally regarded as aberrant sprouts by Alzheimer and others (Fischer, 1907; Simchowicz, 1911; Scheibel and Tomiyasu, 1978). This is supported by Golgi (Scheibel and Tomiyasu, 1978; Ferrer et al., 1983, 1990; Arendt et al., 1986; Ihara, 1988), ultrastructure (Paula-Barbosa et al., 1980), and association of GAP-43, MARCKS, spectrin, and synaptic, axonal, and cytoskeletal proteins (Geddes et al., 1985, 1986, 1990; Masliah et al., 1989, 1990, 1991a, Masliah et al.,b,d, 1992, 1993a; Cotman et al., 1990; Kosik, 1991; Saitoh et al., 1993; Phinney et al., 1999) (see Section 5.4). Abnormally dilated synaptic terminals, indicative of a compensatory response, are found in both aged and demented brains (Braak and Braak, 1988; DeKosky and Scheff, 1990; Ferrer et al., 1990) (see Section 5.4).

Mega neurites, which represent a specific subpopulation of dystrophic neurites (>10  $\mu$ M diameter), are often associated with plaques, contain synaptophysin, hyperphosphorylated PHF- $\tau$ , GAP-43, and are modified by sialic acid addition and glycosylation. These characteristics suggest that they are abnormal neuritic sprouts of atrophic dendritic structures (Espinosa et al., 2001). Such modifications may represent early events in neurofibrillary degeneration mediated by microtubule depolymerization at the growth cone and adhesion interactions (Araujo et al., 1997).

## 3.6 Entorhinal Cortex, Hippocampal Pathway, and Lesion Models

In AD, the entorhinal cortex (EC) shows extensive loss of neurons (Hyman et al., 1984; Geddes et al., 1985) and neuronal cytoskeletal disruption (McKee et al., 1991) whereas the hippocampal region that receives EC innervation (the molecular layer) shows plaque-independent granule cell dendritic pathology (Einstein et al., 1994) and loss of synaptophysin immunoreactivity (Heinonen et al., 1995). In response to these degenerative effects, some patients show regenerative changes in the dentate gyrus (Geddes et al., 1985; Arendt et al., 1998a) and apoE4 patients are impaired in this compared to apoE3 (Arendt et al., 1997; reviewed in Arendt, 2001b). Aged human brain shows increased granule cell axon sprouting, suggesting that the molecular layer might be partially deafferented with age (Cassell and Brown, 1984). The dendritic spine density of granule cells is reduced in AD only in distal segments, possibly indicating sprouting of undamaged proximal segments.

In mice and rats, the entorhinal cortex lesion (ECL) is a well-established model of synaptic plasticity (Poirier et al., 1991a,b; Masliah et al., 1995d, 1996; Danik and Poirier, 1998) and behavioral correlates (Miwa and Ueki, 1996; Good and Honey, 1997; Hardman et al., 1997). ECL-induced deafferentation of the EC input models aspects of AD, albeit in an acute model, and has been used to show age-dependent reduction in sprouting in response to ECL (Scheff et al., 1980). The major neuron type that undergoes sprouting is the granule cell of the dentate gyrus, whose dendritic field is the target of EC innervation. Granule cell axons, so-called mossy fibers, sprout and are detected by Timm's stain (for vesicular zinc) (Danscher, 1981; Gaarskjaer, 1986), or other markers of neurite sprouting GAP-43, synaptophysin; these latter markers also detect sprouting of commissural/associational fibers (Cotman et al., 1991). The ECL paradigm recapitulates developmental gene expression responses seen in adult lesion models, aging, and AD (Kondo et al., 1996; Styren et al., 1999) (see Section 3.2). Such similarities to developmental events include expression of synapsin I, eNCAM, and fetal ALZ-50 reactive clone 1 (FAC1) (reviewed in Bulinski et al., 1998; Styren et al., 1999) and partial similarity to changes in dendritic structure, microtubule (MAP2) metabolism, intermediate filament expression (nestin, vimentin), trkB expression, and glutamate and GABA receptor expression.

Organotypic hippocampal slice culture (OHSC) is a semi-simplified yet physiologically and neuro-organorelevant in vitro system of postnatal hippocampal tissue, widely regarded as a bridge between in vivo and in vitro models, powerful in elucidating mechanisms of complex and necessarily emergent CNS phenomena (Zimmer et al., 1999). OHSCs are typically derived from early postnatal rodents although adult OHSC methods are now available (Temple and Malouf, 2000; Xiang et al., 2000). OHSCs continue to develop and retain organotypic features of the intact hippocampus (Bruce et al., 1995), including development of the mossy fiber pathway that arises from dentate granule cells and projects to the CA3 pyramidal cells (Zimmer and Gahwiler, 1987; Sutula et al., 1989), as well as other synaptic development phenomenon that parallel those observed in vivo (Buchs et al., 1993; Muller et al., 1993; Stoppini et al., 1993).

OHSC is an in vitro model of deafferentationinduced hippocampal neuron sprouting that replicates aspects of ECL. In addition to C/A connections (Frotscher, 1992), the preparation of OHSC transects the perforant path and thereby removes the major extrinsic innervation by the entorhinal cortex (EC) to the granule cell dendritic field in the OML, as well as the commissural projection to the IML. Like ECL in vivo, this deafferentation stimulates sprouting of granule cell mossy axon collaterals into the dentate molecular layer, where they are not normally found in abundance (Gaarskjaer, 1986; Sekiguchi et al., 1996). There they make aberrant synapses (Rudling and Angelin, 1993) with dendrites of the deafferented granule cells that are electrophysiologically functional (Wong and Moss, 1992). Granule cell axon sprouting is altered by intrinsic neural excitability in the absence of cell death (Stringer et al., 1997) (see Sections 3.1, 3.4, 4.2, 8.1).

## 3.7 Cholesterol in the CNS and AD

Metabolism of cholesterol in the brain and cross-talk with peripheral lipid metabolism (reviewed in Dietschy and Turley, 2001; Rapoport, 2001) is an emerging consideration for AD etiology and possible therapeutic targets (Roses and Saunders, 1997; Vance et al., 2000). Levels of cholesterol in the brain are critical for synapse formation and maintenance and recent studies identify cholesterol as a limiting factor in synaptogenesis (reviewed in Koudinov and Koudinova, 2001). Reduced cholesterol may place a limit on plastic processes thus reducing the tendency to develop AD. An issue for very long axons is the ability to supply sufficient cholesterol for rapid axonal growth, especially in regeneration. What proportion of axonal membrane phospholipid is synthesized in situ in axons compared to that made in cell bodies and transported to axons? (reviewed in Vance et al., 2000).

AD brain contains less cholesterol, perhaps because of enhanced efflux of derivatized cholesterol from the brain (Koudinov and Koudinova, 2001). This contributes to AD-related alterations in membrane composition (Bertoni-Freddari, 1988; Majocha et al., 1989; Svennerholm and Gottfries, 1994; Gottfries et al., 1996), membrane fluidity (Scott et al., 1994; Fernandes et al., 1999; Zubenko et al., 1999), and lipid bilayer structure and dynamics (Mason et al., 1992; Mulder et al., 1998) (see Section 4.4). Cholesterol also influences the phosphorylation status of  $\tau$  (Distl and Meske, 2000; Fan et al., 2001; Koudinov and Koudinova, 2001; Ohm et al., 2001), MAP2 phosphorylation in the context of dendrite outgrowth (Fan et al., 2002), and amyloid metabolism (including AB production) and its related membrane fluidity effects (Hartmann, 2001; Buxbaum et al., 2002; Ji et al., 2002; Runz et al., 2002; Wahrle et al., 2002) (see Section 5).

Statins, as inhibitors of cholesterol synthesis, may reduce the prevalence of AD (Jick et al., 2000; Wolozin et al., 2000; Buxbaum et al., 2002; Rockwood et al., 2002; Yaffe et al., 2002), possibly by reducing cholesterol turnover in the brain (Locatelli et al., 2002). Axonal growth ceases when cholesterol synthesis is inhibited by pravastatin and could be reactivated by addition of cholesterol to either cell bodies or distal axons (Posse de Chaves et al., 1997). LTP is inhibited by cholesterol biosynthesis inhibitors (Matthies et al., 1997) and LTP induction is associated with pathway-specific increases in lipid production (Koudinov and Koudinova, 2001). An important contribution of glia is their production of apoE-bound lipoprotein particles to deliver rate-limiting cholesterol to neurons, stimulating both synaptogenesis and stable maintenance of synapses, as measured by synapsin and synaptophysin (Kosik, 1992; Pfrieger and Barres, 1997; Barres and Smith, 2001; Mauch et al., 2001; Ullian et al., 2001) (see Section 4.4). Possible mechanisms include cholesterol as a limiting factor in the structural demands of synaptogenesis including membrane formation, synaptic vesicle formation, and clustering of postsynaptic receptors (Gimpl et al., 1997; Martens et al., 2000; Thiele et al., 2000; Bruses et al., 2001; Lang et al., 2001), activation of synaptogenesis by cholesterol signaling through the apoE receptor LRP (see Section 4.1), or other pathways such as hedgehog, Wnt, and reelin (Herz, 2001a; Rice et al., 2001). Future goals should include determining whether these cellular effects can be generalized to synaptogenesis during learning and memory, whether astrocyte-derived cholesterol is a limiting factor in vivo (see Section 2.4), and evaluating the differential effects apoE isotypes in these phenomena.

#### 4.1 APOLIPOPROTEIN E

Apolipoprotein E (apoE) is a component of several classes of lipoproteins regulating lipid metabolism and redistribution (Mahley and Huang, 1999; LaDu et al.,

2000; Mahley and Rall, 2000). ApoE isotype E4 is a risk factor for familial and late-onset AD, showing increased risk particularly in the 60-80 year age group (Breitner et al., 1999), and earlier age of onset (Roses et al., 1995; Blacker et al., 1997; Meyer, 1998) ApoE4 influences the risk of AD through pleiotropic effects on both the pathology of AD and the environmental and developmental factors influencing its etiologies (reviewed in Teter, 2000; Teter et al., 2002). This pleiotropy obscures the mechanism for apoE4, and may involve a balance or interaction between neurodegenerative (Poirier, 1994; Buttini et al., 1999; reviewed in Teter et al., 2002) and neuroregenerative effects (see Section 8). The major epidemiological effect of E4 in AD is to promote an earlier age of onset than E3, typically by  $\sim 5$  years but as much as 15 years (reviewed in Hyman et al., 1996; Blacker et al., 1997; Meyer, 1998; Mesulam, 1999; Arendt, 2001b; Ashford and Mortimer, 2002). Because AD is characterized by ongoing neurodegeneration, accelerated clinical onset could be caused by defects in apoE-related compensatory mechanisms that repair circuitry (reviewed in Mesulam, 1999; Teter, 2000; Arendt, 2001a) This is only one of several mechanisms that could delay the onset of AD.

A great deal of evidence implicates a role for apoE in AD-associated plasticity (Poirier, 1994; Masliah et al., 1995d, 1996), possibly through its isoform-specific functions in cholesterol and phospholipid metabolism and membrane lipid recycling and trafficking, which facilitate neuronal sprouting (Mahley, 1988). ApoE plays a role in both PNS and CNS synaptic remodeling (Poirier et al., 1993a; Poirier, 1994; Laskowitz et al., 1998) although apoE deficiency does not compromise PNS regeneration, perhaps by compensatory overproduction of another apolipoprotein (Popko et al., 1993), it seems to be essential in the CNS (Poirier et al., 1993a; Masliah et al., 1995b). Evolutionary perspectives of apoE allele frequencies are consistent with roles in diet and lipid metabolism (Corbo and Scacchi, 1999).

Differential intracellular trafficking may underlie apoE isotype effects on plasticity. ApoE isotypes localize differentially and accumulate in neurons and astrocytes (Xu et al., 1998). ApoE isotypes may be sorted into late endosomes, escaping lysosomal hydrolysis, where they can then differentially mediate intracellular process like stimulating neurite outgrowth (Mahley and Rall, 2000). E4 may not be able to escape the endocytic pathway to interact with  $\tau$  or contribute other functions (Hardy et al., 1998; Tesseur et al., 2000).

Many of the activities of apoE are dependent on receptor-mediated events, involving any of a number of low-and high-affinity receptors, including the LDL receptor family of lipoprotein receptors (reviewed in Herz, 2001a), like LRP and HSPG. Several of the neurite outgrowth-promoting properties of apoE isotypes have been shown to be dependent on LRP, both in vitro (Table I) and in vivo (Veinbergs et al., 2001). LRP decreases with age (Kang et al., 2000; Herz, 2001a) and is implicated in AD (Rebeck et al., 1993), with LRP and VLDL polymor-

TABLE I. ApoE4 is Defective in Supporting Neurite Sprouting I	(n	Vitro
---------------------------------------------------------------	----	-------

apoE source	Neurite source	apoE4 effect (apoE3 stimulates)	Depends on	References
Pure	DRG and 1° cortical neuron	Inhibit	Lipoprotein, apoE levels	Handelmann et al., 1992; Nathan et al., 1994; Nathan et al., 2002
Pure	N2A	Inhibit	β-VLDL, LDLR/LRP	Nathan et al., 1994; Nathan et al., 1995
Transfected N2A low expressing	N2A	Inhibit	β-VLDL, HSPG/LRP	Bellosta et al., 1995
Transfected N2A high expressing	N2A	Neutral		De Mattos et al., 1998
Pure	GT1-1 (a HT line)	Neutral	β-VLDL, LRP	Holtzman et al., 1995b
Human plasma HDL, CSF lipoproteins	GT1-1 (a HT line)	Neutral	LRP	Fagan et al., 1996
GFAP transgenic astrocyte	1° HC neuron	Neutral	LRP	Sun et al., 1998
Pure (no lipid) + laminin	1° HC neuron	Stimulates (=E3)		Huang et al., 1995
Transfected HEK cells	1° HC neuron	Stimulates (=E3)		Puttfarcken et al., 1997
Human APOE	Granule neurons	Stimulates (=58% E3)		Teter et al., 1999b
transgenic OHSC		"Inhibits" by dose	apoE levels	Teter et al., 2002

phisms increasing AD risk (Kang et al., 1997; Helbecque et al., 1998). LRP is implicated in LTP in OHSC (Zhou et al., 2000). LRP signaling roles may modulate synaptic plasticity because it interacts with NMDA receptors via the multivalent scaffold protein PSD-95 in postsynaptic membranes, among many possible mechanisms (Gotthardt et al., 2000; reviewed in Herz, 2001a,b; Herz and Strickland, 2001). LRP may mediate the effect of E4 but not E3 stimulating the ERK cascade and CREB (Ohkubo et al., 2001). ApoE isotypes show other signaling-dependent effects (reviewed in Ohm et al., 2001).

ApoE expression is increased in early postnatal development (Muller et al., 1997), which correlates with the onset of synaptic development. ApoE is upregulated by estrogen and in association with estrogen-stimulated, apoE-dependent plasticity (Tam et al., 1986; Stone et al., 1997; Srivastava et al., 1997), and in glia (primarily astrocytes) in regions that undergo estrus cycle-dependent synaptic remodeling (Stone et al., 1997) (see Section 4.11). ApoE mRNA is upregulated in AD (Poirier, 1994) and in the entorhinal cortex lesion model (Poirier et al., 1991a; McRae et al., 1997). Besides effects of apoE levels on plaque development (Bales et al., 1997), levels of expression of the apoE protein have a profound effect on the isotype-specific activity in supporting compensatory sprouting in vitro and in lesion responses and behavior effects in vivo (see Section 4.10). The dose-responsiveness of isotypespecific activities also bears on the therapeutic implications of altering apoE expression levels (see Section 4.12).

## 4.2 ApoE-Dependent Sprouting (reviewed in Teter, 2000; Teter et al., 2002)

**4.3 Alzheimer's Disease.** There is epidemiologic evidence for failure of plasticity in E4 patients with Alzheimer's disease. For example, in later stages of AD, E4 brains show reduced dendritic remodeling of pyramidal

and subcortical neurons in addition to more severe degeneration. ApoE E4 copy number also affects the relationship between (and possible coupling between) neuronal loss and dendritic growth (see Section 8), with E4/E4 showing no relationship, and shows a shift toward proximal branching (Arendt et al., 1997; 1998a; reviewed in Arendt, 2001b). Interestingly, basal dendrites do not consolidate LTP unlike apical dendrites (Arai et al., 1994a) (see Sections 3.4, 3.6, 4.9 for proximal branching effects).

4.4 Sprouting mechanisms and the lipid metabolism model. The role of apoE in stimulating neuronal regeneration has received much support. E4 consistently shows defects (reviewed in Poirier, 1994, 1995; Danik and Poirier, 1998; Holtzman and Fagan, 1998; Laskowitz et al., 1998; Kerr and Kraus, 1998); unfortunately, no studies have examined the relative capacity of E2 to support neurite sprouting. A well-established mechanism involves the role of apoE in lipid metabolism. Among the many activities that apoE has demonstrated that could account for its CNS effects (see Sections 4.1, 4.4), its definitive role in cholesterol and phospholipid scavenging, metabolism, and transport has defined its role in CNS and PNS plasticity after injury (Masliah et al., 1995d, 1996). The central model of this latter role has been described (Boyles et al., 1989; Poirier et al., 1993a, 1994; Laskowitz et al., 1998) where glia phagocytosing degenerating terminals esterify cholesterol from scavenged membrane lipid, repackage it with apoE as a lipoprotein particle and deliver it to neurons to supply cholesterol for neurite growth via their apoE receptors, LDLR or LRP. Aspects of this mechanism were demonstrated originally in the PNS (Boyles et al., 1989; Saada et al., 1995). Recently, apoE and the cholesterol it carries was identified as the glial factor that stimulates new synapse formation in cultured neurons (Mauch et al., 2001; Ullian et al., 2001).

Besides lipid metabolism, isotype-specific effects could be mediated by specific association with lipoprotein particles, inter- and intracellular apoE trafficking, and oxidative effects of apoE. The defective ability of E4 to support neurite sprouting could involve the isotype- and cell type-specific differential localization and accumulation of apoE (see Section 4.1). Stimulation of neurite outgrowth by E3 in vitro is associated with greater neuronal apoE accumulation (Nathan et al., 1994) and E3 extends along neurites more than E4 (Nathan et al., 1995). Studies have shown that apoE isotypes experimentally directed to cytoplasmic compartmentalization exhibited the E4 defect in sprouting of N2A cells; that the carboxy terminus determined intracellular distribution whereas the amino terminus mediated neurite sprouting suggests that the E4 defect may be due to differential cytoplasmic compartmentalization (Huang et al., 1999).

ApoE could play a role in lipid metabolism through its oxidative effects (see Section 10.2). ApoE-dependent effects on oxidative stress could modulate its ability to support neurite sprouting and could account for synaptic disruption observed in apoE-ko mice. Lipid peroxidation toxicity could inhibit sprouting by the inability to efflux such toxins. In humans, E4 genotype shows higher plasma lipid peroxide that correlates with apoE levels (Smith et al., 1998) and higher lipid peroxidation in brain (Ramassamy et al., 2000). The lack of apoE in the apoE-ko mice results in oxidative stress in the periphery and CNS, e.g., increased CNS F2-isoprostanes (Montine et al., 1999; Pratico et al., 1999), which are suppressed in the plasma by vitamin E (Pratico et al., 1998). Vitamin E ameliorates cognitive deficits in apoE-ko mice (Veinbergs et al., 2000); apoE-ko animals demonstrate increased susceptibility to oxidative stress conditions including global ischemia where neuronal damage correlates with 4-HNE (Horsburgh et al., 1999). Lack of apoE, however, (in apoE-ko mice) was found to increase CNS lipid peroxidation without neurodegenerative or synaptic changes, perhaps because of an oxidative magnitude issue (Montine et al., 1999) (see Section 10.2).

#### 4.5 Model systems

4.6 ApoE-knockout mice. Studies of the apoE-ko mouse (Piedrahita et al., 1992) reveal insight into functions of apoE, peripherally and centrally. Although neuropathologically normal, apoE-ko mice show numerous CNS defects including impaired memory and learning deficits, some of which are age-dependent (Gordon et al., 1995; Masliah et al., 1995b,c, 1996; Krzywkowski et al., 1997, 1999; Veinbergs et al., 1997; Veinbergs and Masliah, 1999; Keller et al., 2000; Bi et al., 2001), changes in cholinergic responses (Gordon et al., 1995), age-related disruption in the dendritic cytoskeleton, and reduced synaptophysin and MAP2 in the hippocampus (Masliah et al., 1995a; Veinbergs and Masliah, 1999; reviewed in Masliah et al., 1996;). Some effects, however, may be straindependent (Gandy et al., 1995; Masliah et al., 1996). ApoE-ko mice also demonstrate deficits in response to injury, including cerebral ischemia (Connolly et al., 1996; Laskowitz et al., 1997), and impaired synaptic regeneration (recovery of synaptophysin to entorhinal cortex deafferentation) (Masliah et al., 1995b; Chen Y et al., 1997; Laskowitz et al., 1997; Fagan et al., 1998) (see Section 3.6). Neurotrophic compounds like cerebrolysin that ameliorate behavioral and neurodegenerative changes in apoE-ko mice are associated with upregulation of GAP-43 (Masliah et al., 1999). Some evidence suggests, however, that synapse formation in development is normal in apoE-ko mice, and humans who lack apoE are apparently cognitively normal (Feussner et al., 1996), suggesting the existence of redundant pathways replacing some apoE functions. It may be that in the aging brain these redundant pathways are ineffective, increasing the reliance on apoE activity for plasticity. Loss of synaptic and dendritic density seen separately with age or with absence of apoE expression are synergistic in aged apoE-ko mice (Masliah et al., 1995d).

**4.7 Human apoE isotype transgenics.** Several lines of transgenic mice have been developed that express the human apoE isotypes under the transcriptional control of various promoters: the natural human apoE promoter (human apoE); the astrocyte-specific GFAP promoter (GFAP); the neuron-specific NSE promoter (NSE); and recently, the natural mouse apoE promoter (*mouse apoE*), so-called knock-in mice. Clearly, each has strengths and limitations experimentally and in their relevance to AD. For example, the GFAP transgenic mice have provided what is considered a natural source of lipoprotein particles as synthesized by astrocytes. E3 produced by primary astrocyte cultures from transgenic mice (GFAP) is better than E4 at promoting neurite outgrowth in primary cultured neurons (Sun et al., 1998, Table I). This is also an advantage of the human apoE transgenic mice (see Section 3.6); transgenic line also expresses apoE in vivo with cellular specificity like that seen in humans (Xu et al., 1996, 1998, 1999) (see Section 4.1). Sprouting responses and synaptic disruptions in hippocampal pyramidal neurons of aged apoE-ko mice (GAP-43, MAP-2) and behavioral deficits are better ameliorated by E3 than E4 transgene (human apoE) expression (Veinbergs et al., 1999). Similar results were obtained by infusion of apoE isotypes directly into the brain of apoE-ko mice (Masliah et al., 1997). Behavioral and structural alterations are seen in female E4 but not E3 transgenic mice (NSE) (see Section 4.11). E4 transgenic mice (human ApoE) are unable to compensate for age-related neuronal loss by synaptic remodeling of the residual neurons (Hoffman and Chernak, 1994; Cambon et al., 2000). E4 transgenic mice (NSE) show less synaptophysin (a presynaptic marker) and MAP-2 (a dendritic marker) and behavioral deficits (Buttini et al., 1999; Raber et al., 1998, 2000, 2002). The effects of neuronal expression of human ApoE on sprouting have not been addressed adequately.

**4.8 In vitro sprouting systems.** The mechanisms by which apoE facilitates neuronal sprouting have been studied extensively in vitro. In most studies, E4 was defective in supporting neurite sprouting (Table I). In

these studies, the apoE source varied between pure (recombinant) protein, lipoprotein particles produced by transfected neurons or liver cells (HEK-293), particles produced by transgenic astrocytes (GFAP promoterdriven), or by a balanced production by all CNS cells (human ApoE transgenics). Neurite sprouting was measured in neuron cell lines N2A or GT1-1 (a hypothalamic line), primary hippocampal neurons, or hippocampal granule neuron mossy fibers in OHSC (see Section 3.6). In all these studies, E3 stimulated sprouting, whereas E4 showed an inhibitory effect, no effect, or weakly stimulatory effects on sprouting, always less than (or equal to) E3 (reviewed in Teter, 2000).

Important findings include consideration of the lipidation state of the apoE isoforms to reveal isoform-specific activities. For example, pure E4 inhibits N2A sprouting only when reconstituted with b-VLDL or with other lipid sources (also required for the defect in N2A-expressed apoE4) (Nathan et al., 1994, 1995; Bellosta et al., 1995; Holtzman et al., 1995b), and apoE isoforms expressed in lipoprotein particles by transfected HEK cells do not reveal the E4 defect (Puttfarcken et al., 1997) whereas those produced by transgenic astrocytes do (Sun et al., 1998). The lipidation state of apoE is a critical issue yet to be resolved fully because not all apoE isotype-specific effects, including sprouting, depend on lipidation (reviewed in Nathan et al., 1994, 2002; Jordan et al., 1998; Teter, 2000).

Possible mechanisms of isotype-specific sprouting include isotype-specific effects on lipid efflux (see Sections 3.7, 4.9); apoE cellular accumulation (Ji et al., 1998) (see Section 4.1), microtubule depolymerization and destabilization (Nathan et al., 1995; Pitas, 1996; Roses et al., 1996; Pitas et al., 1998), and neurotoxicity (Marques et al., 1996; reviewed in Teter, 2000; Teter et al., 2002). Several studies show a dependence of E3-stimulated sprouting on the LRP receptor or the HSPG/LRP receptor system (Table I) (Bellosta et al., 1995; Holtzman et al., 1995b; Nathan et al., 1995; Fagan et al., 1996; Sun et al., 1998). Interestingly, exogenous E3 does not rescue the E4 defect in stimulating sprouting (Nathan et al., 1994; Holtzman et al., 1995b).

4.9 Isotype-dependent granule cell mossy fiber **sprouting.** In early development and in OHSC in vitro (see Section 3.6), the early postnatal development of the granule cell mossy fiber system (Gaarskjaer, 1986; Slomianka and Geneser, 1997) parallels the large increase in apoE expression at this time (Muller et al., 1997). Mossy fiber sprouting in OHSC is found to be regionally dependent on apoE expression, where only dorsal dentate granule cells fail to sprout in apoE-ko OHSC (Teter et al., 1999a). These studies indicate that apoE-dependent spouting is region-specific, perhaps reflecting a developmental age-dependent difference in the capability of the granule cells to react to deafferentation. Aspects of this regionspecific, apoE-dependent sprouting have been demonstrated independently in adult animals, where apoE-ko mice show deficient sprouting in response to ECL (Masliah et al., 1995b; Stone et al., 1998) (see Section 4.11).

Granule cell sprouting in OHSC derived from E3 and E4 transgenics (human ApoE) showed that E4 induced sprouting to a level only 50% of that induced by E3 (Teter et al., 1999b). This E4 defect in sprouting was demonstrated recently in vivo using the same transgenic mice and the ECL paradigm, where compensatory sprouting measured by GAP-43 and synaptophysin immunoreactivity did not recover as effectively in E4 as in E3, nor did morphometric measures of sprouting extent (White et al., 2001). ApoE4 transgenic mice (NSE) also show poorer recovery from other lesion paradigms, such as excitotoxic injury (Buttini et al., 2000). The reduced distal mossy fiber sprouting measured in E4 OHSC (outer molecular layer sprouting) may be explained by effects on neurite branching. The effect of apoE4 on proximal neurite branching in AD (see Section 4.3) is also seen in sprouting responses in vitro (Nathan et al., 1995).

4.10 ApoE gain-of-function defect in sprouting. Although E4 reduced sprouting activity in most studies, several studies indicate that the E4 activity in the inhibition of neurite sprouting actually represents a gainof-negative function. First, Nathan et al. (1994) found that dorsal root ganglion (DRG) neurons, Neuro2A cells (Bellosta et al., 1995), and primary cortical neurons (Nathan et al., 2002) extend neurites in the presence of E3, but decreased neurite extension with E4 is dose-dependent. Importantly, E4 inhibition dominates over E3 stimulation, an effect seen in other in vitro systems (Holtzman et al., 1995b) and in bigenic mice (Nathan et al., 1995; Buttini et al., 2000) (see below). Second, in the OHSC model of denervation-induced fiber sprouting (see Section 4.9), transgenic (human apoE) expression of E4 is not only defective in supporting neurite sprouting compared to E3, but increased expression of E4 (by doubling the transgene copy number) inhibited sprouting whereas increasing E3 expression stimulated sprouting (Teter et al., 2002). The apparent gain-of-negative activity of apoE4 could be a form of toxicity (reviewed in Teter, 2000; Teter et al., 2002) that, at higher expression levels, dominates its weak sprouting activity. This could be relevant at the apoE levels measured in OHSC media because similar levels are found in human CSF and brain  $(2-6 \mu g/ml)$  (Hesse et al., 2000). Two studies show in vivo evidence consistent with E4 dominant negative inhibition of neurite sprouting. First, in the loss of synaptic markers (synaptophysin, MAP2 and neurofilament) in response to kainate lesioning, whereas E4 transgenics (NSE) show reduced synaptophysin that is equal to apoE-ko mice, doubling the gene dose causes even greater reductions; notably, the E4 effect dominates over E3 (Buttini et al., 2000). Second, E4specifc cognitive impairments in these same mice (NSE) are not present in nontransgenic apoE-ko littermates (this "gain of function" is in comparison to apoE-ko, not a dose response of E4) (Raber et al., 1998, 2000; reviewed in Teter et al., 2002).

**4.11 ApoE, Gender, Estrogen (see Section 9).** Gender has an impact on ApoE4 effects, further increasing AD risk and diminishing ERT treatment response in postmenopausal women (Corder et al., 1993; Poirier et al., 1993b; Farrer et al., 1995, 1997; Schneider and Farlow, 1997; Yaffe et al., 1997, 2000; Bretsky et al., 1999). These results from human studies are paralleled to some extent by results from studies of transgenic animals.

In OHSC, granule cell sprouting is regionally dependent on apoE expression (see Section 4.6). Although sprouting in wild-type, apoE-expressing OHSC is stimulated by physiological levels of estrogen, an effect that is blocked by both progesterone and tamoxifen, as seen in purified neuron cultures (Chawen et al., 1992; Woolley and McEwen, 1993), estrogen does not stimulate sprouting in apoE-ko OHSC, showing that neuronal sprouting is increased by estrogen in the same hippocampal region where sprouting is dependent on apoE. Likewise, whereas apoE-ko animals show compromised compensatory sprouting in response to ECL lesion in vivo (Masliah et al., 1995a; Masliah et al., 1996; Anderson et al., 1998; Stone et al., 1998), estrogen replacement in ovariectomized mice stimulates sprouting only in wild-type but not apoE-ko mice (Stone et al., 1998). Like the region-specific apoE dependency of estrogen-stimulated granule cell sprouting in OHSC, granule cells in the dorsal region are sensitive specifically to estrogen-stimulated increases in spine density (Miranda et al., 1999). Sprouting may be stimulated by estrogen through upregulation of apoE expression (see Section 4.1). Upregulation of apoE synthesis in glia (primarily astrocytes) occurs in CNS regions that undergo estrus cycle-dependent synaptic remodeling (Stone et al., 1997). Estrogen and apoE may therefore interact in their modulation of both AD risk and CNS plasticity. This is consistent with a postmenopausal decline in peripheral apoE levels (Kushwaha et al., 1991; Muesing et al., 1992). Other possible mechanisms of apoE and estrogen interaction include estrogen receptor polymorphism (Mattila et al., 2000) and oxidation (Inestrosa et al., 1998) (see Section 10.2).

Only female E4 transgenic mice (*NSE*) develop agerelated progressive impairments in spatial learning and memory in the water maze and nonspatial novel object recognition memory (Raber et al., 1998; 2000, reviewed in Teter et al., 2002). These cognitive impairments are independent of the cellular source of apoE as they are observed in mice expressing E4 in neurons (*NSE* transgenics) and in astrocytes (*GFAP* transgenics) (see Section 4.7). The findings that the detrimental effects of E4 are greater in female than in male transgenic mice is consistent with the epidemiological interaction of apoE4 and female gender on increased risk to develop AD.

## 4.12 ApoE Therapeutic Implications: Drug Interactions and Pharmacogenetics

ApoE4 plays a major role in the risk and onset of AD for  $\sim$ 50% of AD cases in the United States (Ashford and Mortimer, 2002); therefore, therapies that target the mechanism of increased risk for apoE4 and reduced risk for apoE3 and E2 would greatly impact AD prevalence. Possible targets include apoE expression levels and regulation, apoE protein structure or gene replacement, and primary targets or secondary effects of apoE activity. The

protein structural determinants of apoE4 are known (Weisgraber, 2001); with further understanding of how structure modulates various apoE activities, this avenue holds promise for drugs that convert the E4 protein to a structure that resembles E3 or E2. Gene replacement may capitalize on emerging stem cell technology, or using cell precursors in the bone marrow that can cross the blood brain barrier and differentiate into a variety of CNS cell types (see Section 10.5).

Reducing the expression of the human apoE4 gene could reduce apoE4-related risk, however, the relevance of apoE gain- and loss-of-function effects are not well understood. Further, human apoE gene regulation is very poorly understood, particularly with respect to effects of current and candidate therapeutic drugs. Estrogen is known to regulate apoE expression, and this may mediate, at least in part, the effect of estrogen replacement therapy (ERT) in improving the cognitive deficits in postmenopausal women with AD and the poorer response of apoE4 women (see Section 4.11). These effects of the efficacy of ERT in AD will be better understood with results from several clinical trials currently in progress (WHIMS and others) as apoE genotype is monitored routinely.

Other therapeutic drugs show apoE isotypedependent effects that may interact with primary targets or secondary effects of apoE activity (reviewed in Poirier, 1999). Tacrine (an anti-cholinesterase) therapy has lower efficacy in E4 (Poirier et al., 1995) and in women with E4, no effect by genotype in men (Farlow et al., 1998), and lower efficacy in E4 women on combination tacrine plus ERT (Schneider and Farlow, 1997). There are indications, however, that apoE genotype may affect only longer-term tacrine therapy (MacGowan et al., 1998). The efficacy of a noradrenergic and vasopressinergic activity facilitator is also higher in E4 (Richard et al., 1997). Citicoline, an intermediate of lipid and acetylcholine biosynthesis that increases cerebral blood flow, shows greater efficacy with E4 (Alvarez et al., 1999). Growth hormone therapy is poor in E4 (Johannsson et al., 1995), possibly involving the mechanisms of GH regulation of plasma ApoE levels (Sjoberg et al., 1994). Other drugs that could modulate apoE expression include therapeutic agents that target oxidative mechanisms, such as vitamin E, selegiline (Sano et al., 1997b), and Ginkgo biloba extract (EGb 761) (Le Bars et al., 1997) (see Sections 4.4, 10.2), anti-inflammatory drugs like NSAIDs that could impact apoE expression through glial responses to inflammation (see Section 10.3), and statins that modulate cholesterol levels and may thereby regulate apoE expression (see Section 3.7).

The gain-of-negative function of E4 could have important clinical implications for the pharmacogenomic efficacy of therapeutic drugs that impact or target apoE expression (Poirier, 1999; Saunders et al., 2000) to the extent that E4-defective sprouting contributes to neuroregenerative events in neurodegenerative conditions (or, for that matter, any toxic activity of E4 that prevents neuroregeneration). A drug that increases apoE expression might show efficacy in E3 but not in E4, and may even exacerbate the E4 condition, whereas the opposite is predicted for E4 defects that are simple loss-of-functions. As in the ERT trials, it will be important to evaluate apoE genotype effects in trials of other drugs that can modulate apoE expression. With the implementation of pharmacogenetic approaches to therapeutic drug design and with testing and efficacious matching of treatment protocol to the genetic polymorphism fingerprint of the patient, understanding of genetic influences on neurodegenerative disease will see rational therapeutic application.

Therapeutic strategies must also consider both when and how the drug target contributes to disease etiology. For example, the apoE4 phenotype of accelerating the age of onset requires prevention strategies and may not respond to drugs designed to slow disease progression. The pleiotropic effects of apoE and its isotypes raise the strong possibility that the isotypes differ in the mechanism by which they contribute to AD etiology. Although apoE has emerged as the strongest genetic risk factor for sporadic AD and is implicated in other neurodegenerative diseases, many other genes are also implicated. Identification of these genes (Tanzi, 1999) has been slow methodologically, but the availability of human genomic sequence to reveal other polymorphisms linked to AD will help pharmacogenetic drug design.

## 5.1 APP AND AB

Other than production of AB, the functions of APP relevant to AD etiology include neuronal development, synaptogenesis, synaptic plasticity, and cell signaling (Luo et al., 1992; Moya et al., 1994; Mucke et al., 1994; Muller et al., 1994; Roch et al., 1994; Qiu et al., 1995; reviewed in Neve et al., 2000, 2001; Neve, 2001; Arendt, 2001a). APP and PS are expressed at higher levels in neurons in regions most affected in AD: hippocampal CA fields, amygdala subregions, and neocortex (Lee et al., 1996). APP gene expression and processing is regulated by NF-KB as an injury-responsive cytokine/neurotrophic factor (Mattson and Camandola, 2001; Weggen et al., 2001) (see Section 10.3). Injury and denervation that induces plasticity also upregulate APP (Banati et al., 1993; Wallace et al., 1993; Beeson et al., 1994; Chauvet et al., 1997), as does cholinergic innervation (Nitsch et al., 1995) (see Section 3.1). APP interacts with substrate adhesion in many ways, including association with adhesion patch components, integrins, transglutaminase, glycosaminoglycans, and collagen (reviewed in Arendt, 2001a).

#### 5.2 APP Trophic Effects and Axonal Transport

The superfamily of amyloid precursor proteins (APP, APLP1,2) is associated with axonal outgrowth in several neural systems (Ohta et al., 1993; Arai et al., 1994b; Moya et al., 1994; Thinakaran et al., 1995; Lyckman et al., 1998) and neurite outgrowth that is APP isoform-dependent (Milward et al., 1992; Qiu et al., 1995). APP is secreted from neurons in response to electrical activity and induces neurite outgrowth, synaptogenesis, and LTP (Roch et al., 1994; Huber et al., 1997; Ishida et al., 1997; Mattson,

1997). Transgenic mice expressing various forms of APP exhibit both degenerative and regenerative changes that depend on age and APP genotype. Transgenic mice expressing APP show increased synaptophysin (King and Arendash, 2002), even at low levels of APP (Mucke et al., 1994), and secreted APP promotes dendrite outgrowth at pM concentrations (Mattson, 1994). APP transgenic mice undergo synaptic, electrophysiologic, and behavioral changes before plaque formation and in the absence of overt neurodegeneration (Hsia et al., 1999; Mucke et al., 2000), although some models do show plaque-associated neuron loss (Calhoun et al., 1998). APP transgenic mice have increased numbers of synapses (Mucke et al., 1994) and increased cortical neuron number before plaque formation (Bondolfi et al., 2002) (see Section 3.5). APP may enhance proliferation of neural stem cells (Ohsawa et al., 1999) (see Section 10.5). In contrast, degenerative changes are expressed in older animals, perhaps reflecting accumulated AB/amyloid or soluble AB toxicity (see Section 5.4).

Members of the APP superfamily of proteins are transported by and play a role in the fast anterograde transport system (Koo et al., 1990; Sisodia et al., 1993); they also accumulate in presynaptic membranes. Axonal pathology is reflected by diminished axonal transport (Geinisman et al., 1977). This may involve the role of APP in chaperoning NCAM and sialic acid to the presynapse. Aging is associated with decreased axonal transport, which may be caused by AB (Kasa et al., 2000). The insulin receptor signaling cascade also affects APP trafficking. Reduction in axonal transport (anterograde) would deplete APP at the presynapse and cause accumulation in other cell compartments where AB production may be favored (Golde et al., 1992). Conversely, reducing APP proteolysis to AB is predicted to lead to trophic APP accumulation at the presynapse (see Sections 3.4, 5.3).

#### 5.3 APP Processing Balance

The processing of APP by  $\alpha$  secretase produces soluble/secreted APP that promotes new synapse formation. Decreasing the amount of functional APP or shifting toward  $\beta$  secretase products could contribute to failure of plasticity and elimination of synapses. This situation could be induced by mutations that inhibit any of these trophic activities of APP, or by mutations or other factors that shift the APP processing balance to produce nonfunctional fragments. Mutations in APP and other situations that shift the processing balance away from secreted APP toward AB42 would interfere with APP-induced plasticity. Transgenic mice expressing such mutated APP show decreased synaptic and dendritic density in the hippocampus, impaired LTP, decreased compensatory synaptogenesis in response to injury, and impaired spatial memory (Games et al., 1995; Masliah et al., 1995b; Chapman et al., 1999). Conceivably, any event that promotes AB deposition could do so by shifting the processing toward more AB production (albeit, ignoring clearance effects). With age, the processing of APP also shifts away from producing the neurotrophic secreted APP form (Palmert et al., 1990; van Gool et al., 1994). APP metabolism is also influenced by

laminin (Monning et al., 1995; Narindrasorasak et al., 1995; Bronfman et al., 1996), cholesterol (Jick et al., 2000; Refolo et al., 2000; Wolozin, 2001), and NSAIDs (see Section 10.3). APP-knockout mice show loss of presynaptic markers, reduced CA1 dendritic length, impaired LTP and cognitive performance, and reduced axon and dendrite growth in vitro (reviewed in Arendt, 2001a).

#### 5.4 Amyloid-β and Amyloid Plaques

A major issue facing the amyloid hypothesis of AD is the relative contributions of the various forms of AB, the peptide released from  $\beta$  secretase-processed APP. AB forms a continuum of aggregation species: monomeric AB, soluble AB, ADDLs, insoluble AB, diffuse amyloid, compact amyloid, and neuritic amyloid, the latter two being the pathologic and diagnostic hallmarks of AD. Independent of fibril or plaque formation, however, AB can alter membrane potential and firing, synaptic transmission, synaptic plasticity, and learning (Cullen et al., 1997; Lambert et al., 1998; Hartley et al., 1999; Chen G et al., 2000; Chen QS et al., 2000; Chapman et al., 2001). AB, especially AB1-42, shows neurotoxic, neuriteinhibiting, and LTP-inhibiting properties (Freir et al., 2001; Dewachter et al., 2002). Soluble AB and AB oligomers inhibit LTP but not LDP, resulting in a 'neuroplasticity imbalance' in the competition for synaptic stabilization (Cullen et al., 1997; Wang et al., 2002), as occurs in development (Constantine-Paton, 1990). Oligomeric, but not monomeric AB inhibits LTP in vivo (Lambert et al., 1998; Walsh et al., 2002).

Negative effects of AB on plasticity are also revealed in studies of vaccination/immunization in mouse models. Active immunization reduces both brain AB/amyloid burden and cognitive impairments in APP transgenic mice (Janus et al., 2000; Morgan et al., 2000). Recent studies using the passive immunization approach in APP transgenic mice also indicate such a correlation, but importantly, very rapid behavioral effects are achieved (within days), even without changing global AB levels in the brain (Dodart et al., 2002). Perhaps relatively minor compartments of AB that inhibit learning are reduced rapidly by peripheral immunization, which is consistent with reports that soluble AB is a better correlate of memory impairment (Lue et al., 1999; McLean et al., 1999; Koistinaho et al., 2001). Clinical immunization for AB, despite an early setback, remains a potential therapeutic strategy.

These neurotoxic effects of AB may act to destroy synapses that are no longer required or are underutilized. In contrast to these neurotoxic effects, however, low concentrations of AB can show neurotrophic effects (Calabrese, 2001), and laminin and AB act synergistically in stimulating neurite outgrowth (Koo et al., 1993). Low levels of AB can modulate the activity of the transcription factor CREB (Sato et al., 1997), a factor necessary for neuronal plasticity (Segal and Murphy, 1998; Silva et al., 1998). AB may play a functional role in membrane lipid dynamics (Muller et al., 2001; Chochina et al., 2001). AB enhances the transport, uptake, and oxidative metabolism of lipids, acting much like an apolipoprotein (Wood et al., 1999), a process that takes place in the ER/trans Golgi and endolysosomal pathways, which are also utilized by apoE (Jensen et al., 1994). The possible equilibrium between plaque-bound AB and soluble AB is not understood. Plaques could be a localized source of soluble AB which, when released, could impact plasticity responses, and contribute to the plaque-association of dystrophic neurites, growth-promoting factors, and synaptic proteins (SNAP-25, synaptophysin, synaptotagmin, chromogranins, NT75, spectrin), as reviewed by Arendt (2001a) (see Section 3.5). Numerous growth-promoting factors associated with plaques could contribute to stimulated sprouting, such as S100b, bFGF, HGF, PDGF, Trk receptors, proteoglycans, EGF-R, ICAM, integrins, collagen, laminin (reviewed in Arendt, 2001a). Other plaque constituents like perlecan, agrin, and laminin could also contribute to localized sprouting responses (Phinney et al., 1999).

#### 6. TAU

Tau is a member of the MAP family (reviewed in Maccioni and Cambiazo, 1995). Aggregated, hyperphosphorylated  $\tau$  forms neurofibrillary tangles (NFT), the intracellular pathological hallmark of AD (reviewed in Lovestone and Reynolds 1997; Lovestone et al., 2001). Its expression and phosphorylation is associated with increased neuroplasticity in vivo and in vitro (Busciglio et al., 1987; Viereck et al., 1989; Trojanowski et al., 1993; Brion et al., 1994; Black et al., 1996; Lovestone and Reynolds, 1997). Hyperphosphorylation may also cause deleterious effects on plasticity, however, and may underlie its role in the etiology of AD (Maccioni and Cambiazo, 1995; Mandelkow et al., 1995). Tau modulates cytoskeletal and microtubule dynamics that contribute to growth cone migration and collateral branching (Gallo and Letourneau, 1999). Tau plays a major role in the outgrowth of neurites and axonal development (Maccioni and Cambiazo, 1995). NFT-bearing hippocampal neurons show more extensive dendritic trees, suggesting a concurrent or previous induction of reactive plasticity (Gertz et al., 1990) (see Section 3.4). NFT-bearing neurons contain numerous growth-associated proteins (reviewed in Arendt, 2001a). Antisense to tau mRNA suppresses neurite formation in B103 cells (Lambert et al., 1995). Tau overexpression by PC12 cells induces neurite extension, and NGF-induced extension is associated with large upregulation of  $\tau$  (Drubin et al., 1985; Esmaeli-Azad et al., 1994).

The equilibrium between  $\tau$  phosphorylation and dephosphorylation modulates the stability of the cytoskeleton and thereby the axonal morphology. Tau is phosphorylated by several kinases including GSK3b and Cdk5, and broken down by several phosphatases including A and B. Breakdown of this equilibrium causes structural and conformational changes in  $\tau$ , thus affecting binding with tubulin and the capacity to promote microtubule assembly (Mandelkow et al., 1995; von Bergen et al., 2000). This may promote NFT, particularly in limbic structures (see Section 3.1), leading to cytoskeletal dysfunction. Dentate granule cell mossy fiber axons that undergo deafferentation-induced sprouting (see Sections 3.6, 4.6) display

excessive  $\tau$  phosphorylation (Koudinov and Koudinova, 2001). It is not clear whether neurofibrillary-induced neurodegeneration is a later event in AD or whether its pathology simply cannot be detected in early AD (see Sections 3.4, 3.5).

Tau can link trophic signaling with cytoskeletal rearrangements involved with dendritic sprouting (see Sections 8.2, 8.3). The protein kinase Cdk5 and its neuronspecific activator p35 are essential molecules for neuronal migration and regulate axonal extension through phosphorylation of MAPs including  $\tau$  (Pigino et al., 1997; Paglini et al., 1998). The formation of a stable Cdk5/p35 complex in hippocampal neurons (Alvarez et al., 1999) may lead to constitutive activation of the protein kinase with a consequent increase in  $\tau$  phosphorylation. The complex concentrates at the leading edge of the growth cone (Pigino et al., 1997). Laminin stimulates p35 expression, increasing its redistribution to the growth cone (Ramirez et al., 1999), contributing to the axonoutgrowth activity of laminin (Paglini et al., 1998). Cdk5 may link  $\tau$  hyperphosphorylation and AB (Alvarez et al., 1999; Maccioni et al., 2001). Therefore, the Cdk5 system may provide an important regulatory link between extracellular signals like laminin and the intracellular organization of MAPs and other cytoskeletal proteins involved in axon elongation (Maccioni et al., 2001). Cdk5 interacting proteins include  $\tau$ , synapsin, CK1, b- and g-catenins, N-cadherins, Rac GTPase and Pak1 (which impacts actin cytoskeleton dynamics) (reviewed in Maccioni et al., 2001) (see Sections 3.4, 8.2).

#### 7. PRESENILINS

PS-1 is necessary for normal neurogenesis and survival (Shen et al., 1997; Wong et al., 1997) and localizes to synaptic membranes and neurite growth cones. Presenilins are involved in intracellular trafficking, developmental signaling pathways, and Ca<sup>2+</sup> homeostasis (Shen et al., 1997; Wong et al., 1997; Naruse et al., 1998; Nishimura et al., 1999). Ca<sup>2+</sup> dysregulation could underlie effects of PS-1 on LTP: wild-type PS-1 underexpression impairs LTP in mice (Morton et al., 2002) and rats (Dewachter et al., 2002), mutant PS-1 alters LTP (Parent et al., 1999; Zaman et al., 2000); however, mutant but not wild-type PS-1 and mutant PS-2 facilitate weak-stimulation LTP in brain slices (Schneider et al., 2001). Mutant PS-1 may interfere with metabolism of  $\beta$ - and  $\gamma$ -catenin, which are involved in synapse formation and stabilization (Zhang et al., 1998; Kang et al., 1999) and cell adhesion (Noll et al., 2000) (see Section 6). Wild-type PS-1 stimulates whereas mutant PS1 inhibits NGF-induced neurite outgrowth (Furukawa et al., 1998; Dowjat et al., 1999; reviewed in Arendt, 2001a). PS-1 cleaves Notch, which inhibits neurite outgrowth (Berezovska et al., 1999; Sestan et al., 1999; Figueroa et al., 2002). Presenilins are also considered as a therapeutic target for AD (Golde and Younkin, 2001), however, the negative secondary effects of inhibiting presenilin need further investigation (Dewachter et al., 2002).

## 8. NEURODEGENERATION AND NEUROREGENERATION INTERACTIONS

## 8.1 Degeneration-Regeneration Cross Talk and Combinatorial Signaling

The same trophic signals that control survival can also promote neurite outgrowth (Campenot, 1994; Meyer-Franke et al., 1995). This could allow for a mechanism whereby simply promoting survival stimulates plasticity (Goldberg and Barres, 2000). Trophic responsiveness can be dependent on continuous trophic stimulation, where competition for limited target-derived trophic factors can ultimately decide cell fate, and thus it is possible that continuous availability of trophic stimulation could be limiting for plasticity mechanisms as well (Goldberg and Barres, 2000).

Neurons do not extend axons by default but must be signaled specifically to do so (Goldberg and Barres, 2000). Promotion of plasticity requires both presentation of an extrinsic stimulus and the intrinsic responsiveness of the neuron, which includes states that can be induced transcriptionally (Smith and Skene, 1997). Responsiveness of neurons to intrinsic and extrinsic signals that promote plasticity may come from combinatorial signaling, e.g., simultaneous presentation of electrical activity and growth factors, like BDNF. (McAllister et al., 1996; (reviewed in Goldberg and Barres, 2000). Electrically and biochemically active neurites would therefore survive and grow in response to trophic stimulation; for example, granule cell axon sprouting can result from alteration of the intrinsic neural excitability in the absence of cell death (Stringer et al., 1997). Thus, reduced neuronal activity would reduce its responsiveness to stimulation of plasticity.

Many factors influence both neuronal death and neurite sprouting, for example, c-Jun and GAP-43 (Herdegen et al., 1997; Gagliardini et al., 2000; Wehrle et al., 2001), and neurotrophins that promote neurite growth (Levi-Montalcini, 1987; Campenot, 1994; Meyer-Franke et al., 1995; Henderson, 1996). Other factors include substrate molecules like laminins, although this may be insufficient (Goldberg and Barres, 2000). Retrogradely transported signals like CREB (Silva et al., 1998) and signaling by the ras/raf/MAP pathway may play important roles in intra-neuronal signaling of plasticity (Perron and Bixby, 1999) (see Sections 3.2, 6).

The relationship between plasticity and the classical plaque pathology of AD is unclear. Granule cell dendritic regression is not modified by plaque association (Einstein et al., 1994). Transgenic mice expressing anti-NGF antibody develop amyloid plaques, NFT-like inclusions, neuron losses, and behavioral deficiencies with age (Capsoni et al., 2000) including impaired spatial learning (Van der Zee et al., 1995a,b; Chen KS et al., 1997). Ex-boxers with an increased plasticity burden (injury-induced) have ADlike neuropathological changes (Tokuda et al., 1991; Geddes et al., 1996). The increased risk of AD with head injury and stroke (Salib and Hillier, 1997; Snowdon et al., 1997) may require widespread or chronic injury combined with factors or events that inhibit neuroplasticity responses (see Sections 3.5, 5.4).

#### 8.2 Cell Cycle

If a neuron that is committed to permanent cessation of cell division is forced, through ectopic expression of cell cycle proteins, to reenter the cell cycle, it may die. In AD frontal cortex during the early Braak I/II stage, when there in no  $\tau$  or amyloid pathology or related dementia, mitotic events are activated including increased MAP2 and ERK1/2, which may lead to cell death (Arendt, 2001b). Tau kinases like MAP kinases, Cdk5, and others, are all associated with the cell cycle (reviewed in Arendt, 2001a,b). MAP kinases are activated by cell surface receptors through p21ras (Stokoe et al., 1994), which also plays a role in dendritic proliferation and synaptogenesis (Phillips and Belardo, 1994) (see Section 8.3).

#### 8.3 Nitric Oxide (NO)

NO participates in axonal remodeling at the growth cone and synaptogenesis during development and regeneration (Hess et al., 1993; Wu et al., 1994; Yu, 1994; Luth et al., 1995; Rossiter et al., 1996; Yan et al., 1996; Downen et al., 1999). NO may be the retrograde messenger in LTP and may serve to help maintain normal LTP; however, NO also mediates some excitotoxicity (glutaminergic) mechanisms and has anti-proliferative effects (Arendt, 2001b). The NO synthesizing enzyme nNOS is dynamically regulated in neuronal development, plasticity, and responses to injury (Dawson et al., 1994, 1998; Dawson and Snyder, 1994; Forstermann et al., 1995). Activation of NF-KB in astrocytes increases iNOS expression and NO production. Changes of NOS (nNOS, neuronal) and iNOS (glial) (Srivastava et al., 1997) in AD are inconsistent (Law et al., 2001). Endothelial eNOS mediates neuroprotective actions of NO in ischemia and multi-infarct dementia (Law et al., 2001).

NO and other oxidative stress intermediates activate p21ras, a potential endogenous NO-redox-sensitive effector molecule (Yun et al., 1998). p21ras is highly colocalized with nNOS expression in AD and in NFT-bearing neurons (Luth et al., 2000). P21ras is overexpressed in advanced AD (Gartner et al., 1995), and is upregulated in early AD in affected regions before neurofibrillary degeneration (Gartner et al., 1999) (see Sections 3.4, 10.4), paralleling nNOS (Luth and Arendt, 1998; Luth et al., 2000). This may set up an autocrine loop (Lander et al., 1997) and may exacerbate neurofibrillary degeneration and limit the ability to terminate the vicious cycle. This relationship may switch two potentially neuroprotective mechanisms of NO and p21ras into a chronic neurodegenerative process (Arendt, 2001b).

#### 9.1 GENDER AND ESTROGEN

Estrogen plays a powerful, pleiotropic role in many neurodegenerative conditions including AD (reviewed in Brinton, 2001; Garcia-Segura et al., 2001; McEwen, 2001; Wise et al., 2001a,b). Women have been shown to have increased risk, earlier onset, and more rapid progression of

AD than men, although gender-specific morbidity is an issue (Sanderson et al., 2002). Postmenopausal loss of estrogens leads to generally reversible decreases in memory that respond to ERT (Sherwin and Tulandi, 1996). Besides mechanisms of blocking neurotoxicity directly, estrogen acts at various levels of plasticity: axon sprouting, synaptogenesis, and promoting synaptic transmission (electrophysiologically and biochemically). These effects can be ascribed to either receptor-dependent mechanisms, primarily transcriptional, including direct effects of ER in transcription and indirect effects through other transcription factors like CREB and Akt, as well as their retrograde transport (McEwen, 2001) or receptor-independent (rapid) mechanisms involving activational effects of second messenger systems, coexisting neurotransmission, or coordinated activation of both (Kelly et al., 1977; Nabekura et al., 1986; Wong and Moss, 1992; Garcia-Segura et al., 2001) (see Section 8.1), as well as oxidative effects of the estrogen molecule (see Section 9.3). Other secondary effects could be mediated through effects on AB,  $\tau$ , microtubules, apoE, GAP-43, BDNF, ERK, IGF-1, NF-KB, CREB, gliosis, neurogenesis (Blanco et al., 1990; Gould et al., 1999; Tanapat et al., 1999), differentiation, or many other modulators of plasticity.

#### 9.2 Estrogen Replacement Therapy (ERT)

Estrogen replacement decreases the risk of AD in postmenopausal women (Paganini-Hill, 1996; Kawas et al., 1997), delays the age of onset (Tang et al., 1996), and perhaps slows the decline; however, it remains controversial whether ERT can treat the disease once it has reached the clinical stage (Henderson et al., 2000; Marder and Sano, 2000; Mulnard et al., 2000; Wang et al., 2000). This latter effect is consistent with experimental results indicating that neuroprotective effects of estrogen occur only when administered before or during the neurodegenerative stimulus, but not after (Garcia-Segura et al., 2001). The therapeutic efficacy of ERT depends on its administration protocol: time course, treatment window, endogenous vs. exogenous hormone, and neuron populationspecific effects on promoting survival vs. death. The mechanism of ERT efficacy is unlikely to include antioxidant effects as they require very high hormone concentrations to reduce lipid peroxidation (Vedder et al., 1999) (see Sections 4.4, 10.2); however, whether hormone concentrations are modulated by local aromatase expression are not known (Garcia-Segura et al., 2001). This is critical for considerations of testosterone therapy for men, in terms of its estrogenic actions (Cyr et al., 2000; Goodenough et al., 2000; Twist et al., 2000; Bowen, 2001). Testosterone also has estrogen-independent, potentially beneficial actions on amyloid toxicity (Pike, 2001).

#### 9.3 Estrogen/Plasticity

(Reviewed in Toran-Allerand et al., 1999; Brinton, 2001; McEwen, 2001; Kelly and Levin, 2001).

Estrogen stimulates axon and dendrite plasticity in the limbic neurons of both male and female brain (Ferreira and Caceres, 1991; Lorenzo et al., 1992; Woolley and McEwen, 1992; Woolley et al., 1996; McEwen et al., 1997; Teter et al., 1999a). Estrogens may support neuronal functions and confer resistance to neural damage by their ability to maintain synaptic connections (McEwen, 2001). Many studies demonstrate neuroprotective effects of estrogen in a variety of systems, and these are often associated with changes in gene expression, including genes that effect axonal elongation and synaptogenesis (GAP-43,  $\tau$ , microtubules) (Ferreira and Caceres, 1991; Shughrue and Dorsa, 1993). Estrogen modulates plasticity during development and in adult CNS (Matsumoto, 1991; Garcia-Segura et al., 1994; McEwen, 1996; Woolley, 1998). Estrogen enhances neurite outgrowth by repressing GFAP and reorganizing laminin (Rozovsky et al., 2002). Estrogen can activate neurite mRNA translation (Pierce et al., 2000; Tiedge et al., 2001; Steward and Schuman, 2001).

Granule cell sprouting is stimulated by physiological levels of estrogen (which is blocked by progesterone and tamoxifen) in both wild-type OHSC (Teter et al., 1999a) (see Sections 4.6, 4.11) and in purified neuron cultures (Chawen et al., 1992; Woolley and McEwen, 1993). This effect is hippocampal region-specific, occurring only in the dorsal dentate region, both in OHSC (in the same region that is also dependent on apoE expression) and in vivo in adult, female rats, where short-term estrogen replacement in long-term estrogen-deprived females increases dentate granule cell spine density primarily by the dorsal region (Miranda et al., 1999). Sprouting of hippocampal neurons in response to ECL (see Section 3.6) was reduced by ovariectomy in rats and mice (Stone et al., 1998, 2000), and estrogen replacement rescues sprouting (Morse et al., 1986, 1992; Stone et al., 1998) (see Section 4.11).

Estrogen stimulates cyclic induction of synapses and dendritic spines in the hypothalamus and hippocampus of female rats (reviewed in Woolley and McEwen, 1992; McEwen, 2001). Synapse formation induced by estrogen may differ from that occurring during development, however (see Section 2.2): estrogen increases the number of synapses on multiple synaptic boutons between neurons not connected previously (Yankova et al., 2001). Estrogen induces both pre- and postsynaptic markers where new spines are formed (Brake et al., 2001).

## 10.1 TREATMENTS FOR PLASTICITY (see Sections 3.7, 4.12, 5, 9.2)

The public health impact of AD is predicted to rise at least three-fold in the next 50 years (Sloane et al., 2002). Clearly, all rational therapeutic avenues should be tested, but therapeutic stimulation, stabilization, or recovery of plasticity mechanisms could impact all neurodegenerative diseases. Current AD therapy targets cholinergic dysfunction, which may be linked to effects on plasticity through modulation of APP metabolism (and  $\tau$  phosphorylation) by affecting the coupling of M1 muscarinic ACh receptors to G proteins (Fisher et al., 2000) (see Section 5.3). Other therapies under development focus specifically on AB,  $\tau$ , inflammation, and oxidation (Galasko, 2001); if these pathologic phenotypes contribute to AD indirectly through interaction with an underlying process of plasticity, the effectiveness of interventions targeting these hallmarks may be enhanced if age-related plasticity failure is also treated. New drug strategies that target mediators of either neurodegenerative processes or neuroplastic processes must be considered for their pleiotropic and potentially confounding roles in both, as exemplified by apoE, NF- $\kappa$ B, etc. Of critical importance for the efficacy of plasticity-stimulating therapies is whether they create neural networks that are competent to replace lost function (see Section 10.5).

There are several avenues for stimulating plasticity in the damaged CNS, with targets at all levels of plasticity failure (Horner and Gage, 2000). Enhancing regrowth has been targeted as a therapeutic strategy. Putative neurotrophic agents such as Cerebrolysin have been reported to have positive effects in clinical trials, with sustained improvement after short treatment of AD (Bae et al., 2000; Ruther et al., 2000; Xiao et al., 2000); Cerebrolysin also ameliorates behavioral deficits and neurodegeneration in apoE-ko mice (Masliah et al., 1999). Propentofylline shows neurotrophic effects on glia function (Wilkinson, 2001), and cholesterol inhibitors like statins may reduce the incidence of AD (see Section 3.7). Memory rehabilitation, which targets mechanisms of cognitive reserve and compensatory reorganization to activate alternative, intact brain structures (facilitation of residual explicit memory, or, the 'use it or lose it' phenomenon), can be clinically effective (Grady, 1996), and alternative and innovative techniques are still under refinement (De Vreese et al., 2001).

## 10.2 Oxidation

As a lipid rich organ, the CNS is particularly susceptible to effects of lipid peroxidation in modulating cellular signaling pathways, cell dysfunction, and cell death in the nervous system (Keller and Mattson, 1998). In AD, emerging evidence provides strong support for a role for oxidative stress in neurodegeneration, as multiple indices of oxidative stress have been observed, including protein oxidation, decreased polyunsaturated fatty acids, mitochondrial and nuclear DNA damage, as well lipid peroxidation markers 4HNE (Savre et al., 1997; Markesbery and Carney, 1999), and F2 and F4 isoprostanes (Nourooz-Zadeh et al., 1999), variously detected in brain and CSF. Vitamin E slows cognitive decline in AD (Sano et al., 1997a) and in rat models (Yamada et al., 1999) (see Section 4.4). Although it is not clear what causal relation oxidation has to AD etiology, e.g., whether it is a secondary effect of the stress caused by synaptic or neuronal loss, antioxidant therapies have shown limited but promising efficacy in treating AD (Pitchumoni and Doraiswamy, 1999).

Lipid peroxidation toxicity could inhibit sprouting by the inability to efflux such toxins; an efflux defect is shown by oxidized HDL (Therond et al., 1999). This oxidation-induced lipoprotein aggregation is neurotoxic to primary neurons and is accompanied by cytoskeletal microtubule disruption and inhibition of neuritogenesis (Kivatinitz et al., 1997). Importantly, 4HNE disruption of microtubule organization inhibits neuronal sprouting (Neely et al., 1999). Effects of lipid peroxidation toxicity on inhibiting neuritogenesis could also involve apoE (see Sections 4.4, 8, 8.3).

#### 10.3 Inflammation/NF-κB

NF- $\kappa$ B is directly required for synaptic plasticity, as shown in in vitro hippocampal slices (Albensi and Mattson, 2000). NF- $\kappa$ B is activated in association with LTP (Worley et al., 1993; Meberg et al., 1996). NF- $\kappa$ B is located in synapses at considerable distance from its canonical nuclear site of action, suggesting that it modulates synaptic function locally (O'Neill and Kaltschmidt, 1997).

NF-κB activation (reviewed in Mattson and Camandola, 2001) is implicated in AD (Perez-Otano et al., 1996; Lukiw and Bazan, 1998). NF-KB activity is increased in AD brain, including cholinergic neurons in the basal forebrain (Boissiere et al., 1997; O'Neill and Kaltschmidt, 1997). This could represent a neuroprotective and a cytoprotective response to plaques, because AB and secreted APP can activate NF- $\kappa$ B (Barger et al., 1995), and is associated with neuroprotective response to metabolic/ excitotoxic events (Barger and Mattson, 1996) and mutant PS-1 (Guo et al., 1998). NSAIDs that target NF-κB have been shown to reduce the incidence of AD (Akiyama et al., 2000), even at low NSAID doses (Broe et al., 2000). Some NSAIDs reduce AB production, however, by modulating y-secretase (Weggen et al., 2001) and could thereby influence the mechanism of plasticity involving the processing balance of APP (see Sections 4.1, 5.3, 8.3, 9, 10).

#### **10.4 Growth Factors**

NGF can induce sprouting and outgrowth, particularly after injury (Ramer et al., 2000; reviewed in Sofroniew et al., 2001), consistent with retrograde transport of trkA signaling complexes that alters gene expression in NGF-responsive neurons, including cholinergic neurons (Knipper et al., 1994; Holtzman et al., 1995a), which account for most of the NGF-responsive neurons in the adult CNS. The extent to which NGF is necessary for cholinergic survival of adult cholinergic neurons is controversial (reviewed in Rattray, 2001) (see Section 8). Indeed, AD brain shows increased NGF in the cortex and hippocampus (Jette et al., 1994; Scott et al., 1995; Fahnestock et al., 1996; Hock et al., 1998), which may reflect an increased demand for cholinergic input with a decreased ability of cholinergic neurons for retrograde transport of NGF. Although NGF may not have a classical neurotrophic role in cholinergic survival, i.e., through actions that are independent of retrograde signaling and gene expression, it is an important regulator of neuron morphology and function that would be predicted to maintain or improve cholinergic function in AD by promoting survival of degenerating neurons, promoting sprouting and enhancing neurotransmitter synthesis, and enhancing neuronal firing (reviewed in Rattray, 2001). Abnormal neurite growth might be associated with elevated NGF receptors (Ernfors et al., 1990; Mufson and Kordower, 1992) that precedes neurofibrillary degeneration (Arendt, 1993) (see Sections 3.4, 8.3).

NGF has been considered as a therapeutic target; however, problems with CNS delivery and side effects (pain) limit its clinical application (Eriksdotter Jonhagen et al., 1998). NGF application to the uninjured CNS causes cholinergic neurons to grow, sprout, increase ChAT, and increase choline uptake (Mobley et al., 1985; Higgins et al., 1989; Lapchak et al., 1992; Heisenberg et al., 1994). Experimental methods of delivery include gene transfer, cell grafts, or direct administration of NGF by intracerebroventricular infusion (reviewed in Rattray, 2001). These approaches have been successful in enhancing cholinergic neuron function and restoring some behavioral function in response to deafferenting lesions or impaired cognitive function. Drugs that increase NGF synthesis in astrocytes include propentofylline, a phosphodiesterase inhibitor (Rother et al., 1998), and various quinone derivatives (Takeuchi et al., 1990; Yamaguchi et al., 1993; Yamada et al., 1999). Nicotinic treatment that targets NGF production is also a possibility (Rattray, 2001). NGF mimetic drugs, like Neotrofin and AIT-082, are being tested in clinical trials (Emilien et al., 2000).

Neurotrophins are pivotal regulators of neurite outgrowth (Crutcher, 1986; Kang and Schuman, 1995). For example, BDNF acts at the synaptic level and is altered in AD (Murer et al., 2001). Further, GDNF clinical trials are underway (Maimone et al., 2001) and other neurotrophic strategies are being considered (Siegel and Chauhan, 2000).

#### 10.5 Neurogenesis

Adult hippocampal neurons retain their proliferative capacity (Brewer, 1999, 2000; Seaberg and van der Kooy, 2002), where they provide a continuous replacement of neurons in the dentate gyrus (Seaberg and van der Kooy, 2002), particularly in conditions of enhanced learning (Kempermann et al., 1997, 1998a,b; Huang et al., 1998; Gould et al., 1999). Neurogenesis in the hippocampus declines with age (Cameron and McKay, 1999). An unanswered question is whether neurogenic capacity declines more in AD. Regardless, replacing lost neurons and reversing the age-related decrease in neurogenesis could be a therapeutic target using neural stem cell technology or other neuronal sources. This approach is attractive because neurogenesis occurs naturally and replacement does not suffer from caveats invoked for stimulation of aberrant sprouting, although it is not clear whether neural networks created by these new neurons are competent to replace lost function (see Section 2). Of critical importance is the orchestration of topographically accurate migration, targeted differentiation, and synaptic functionality of transplanted cells (Gage et al., 1995; Flax et al., 1998; Zhou et al., 1998; reviewed in Horner and Gage, 2000). Multipotent cells from the blood lineage, injected peripherally, migrate through the blood brain barrier; these may also provide a source and therapeutic avenue for CNS neuron

replacement (Bartlett, 1982; Bjornson et al., 1999; Brustle et al., 1999; Ono et al., 1999) (see Sections 5.2, 9.1).

#### REFERENCES

- Agnati LF, Benfenati F, Solfrini V, Biagini G, Fuxe K, Guidolin D, Carani C, Zini I. 1992. Brain aging and neuronal plasticity. Ann N Y Acad Sci 673:180–186.
- Akiyama H, Barger S, Barnum S, Bradt B, Bauer J, Cole GM, Cooper NR, Eikelenboom P, Emmerling M, Fiebich BL, Finch CE, Frautschy S, Griffin WS, Hampel H, Hull M, Landreth G, Lue L, Mrak R, MacKenzie IR, McGeer PL, O'Banion MK, Pachter J, Pasinetti G, Plata-Salaman C, Rogers J, Rydel R, Shen Y, Streit W, Strohmeyer R, Tooyoma I, Van Muiswinkel FL, Veerhuis R, Walker D, Webster S, Wegrzyniak B, Wenk G, Wyss-Coray T. 2000. Inflammation and Alzheimer's disease. Neurobiol Aging 21:383–421.
- Albensi BC, Mattson MP. 2000. Evidence for the involvement of TNF and NF-κB in hippocampal synaptic plasticity. Synapse 35:151–159.
- Alexander GE, Chen K, Pietrini P, Rapoport SI, Reiman EM. 2002. Longitudinal PET evaluation of cerebral metabolic decline in dementia: a potential outcome measure in Alzheimer's disease treatment studies. Am J Psychiatry 159:738–745.
- Alvarez A, Toro R, Munoz JP, Maccioni RB. 1999. Inhibition of tau phosphorylating protein kinase Cdk5 prevents A-β induced neuronal death in Alzheimer's neuronal models. Mol Biol Cell 10:61a.
- Alvarez-Maubecin V, Garcia-Hernandez F, Williams JT, Van Bockstaele EJ. 2000. Functional coupling between neurons and glia. J Neurosci 20:4091–4098.
- Alvarez XA, Muozo R, Pichel V, Perez P, Laredo M, Fernandez-Novoa L, Corzo L, Zas R, Alcaraz M, Secades JJ, Lozano R, Cacabelos R. 1999. Double-blind placebo-controlled study with citicoline in APOE genotyped Alzheimer's disease patients. Effects on cognitive performance, brain bioelectrical activity and cerebral perfusion. Methods Find Exp Clin Pharmacol 21:633–644.
- Anderson B. 1996. Axonal length correlates with dementia severity in Alzheimer's disease. Med Sci Res 24:271–273.
- Anderson KJ, Scheff SW, DeKosky ST. 1986. Reactive synaptogenesis in hippocampal area CA1 of aged and young adult rats. J Comp Neurol 252:374–384.
- Anderson R, Barnes JC, Bliss TVP, Cain DP, Cambon K, Davies HA, Errington ML, Fellows LA, Gray RA, Hoh T, Stewart M, Large CH, Higgins GA. 1998. Behavioural, physiological and morphological analysis of a line of Apolipoprotein E knockout mouse. Neuroscience 85:93–110.
- Anderton BH, Callahan L, Coleman P, Davies P, Flood D, Jicha GA, Ohm T, Weaver C.1998. Dendritic changes in Alzheimer's disease and factors that may underlie these changes. Prog Neurobiol 55:595–609.
- Arai A, Black J, Lynch G. 1994a. Origins of the variations in long-term potentiation between synapses in the basal versus apical dendrites of hippocampal neurons. Hippocampus 4:1–9.
- Arai H, Higuchi S, Matsushita S, Yuzuriha T, Trojanowski JQ, Lee VM.1994b. Expression of beta-amyloid precursor protein in the developing human spinal cord. Brain Res 642:132–136.
- Araujo H, Menezes M, Mendez-Otero R.1997. Blockage of 9-O-acetyl gangliosides induces microtubule depolymerization in growth cones and neurites. Eur J Cell Biol 72:202–213.
- Arendt T, Zvegintseva HG, Leontovich TA. 1986. Dendritic changes in the basal nucleus of Meynert and in the diagonal band nucleus in Alzheimer's disease–a quantitative Golgi investigation. Neuroscience 19: 1265–1278.
- Arendt T, Zvegintseva H. 1987. Alzheimer's disease: increase in dendritic branching of reticular neurons in basal nucleus–a sign of regeneration? In: Dowdell MJ, Hawthorne JN, editors. Cellular and molecular basis of cholinergic function. Weinhem: VCH. p 869–873.

- Arendt T. 1993. Neuronal dedifferentiation and degeneration in Alzheimer's disease. Biol Chem 374:911–912.
- Arendt T, Bruckner MK, Bigl V, Marcova L. 1995. Dendritic reorganization in the basal forebrain under degenerative conditions and its defects in Alzheimer's disease. II. Ageing, Korsakoff's disease, Parkinson's disease, Alzheimer's disease. J Comp Neurol 351:189–222.
- Arendt T, Schindler C, Bruckner MK, Eschrich K, Bigl V, Zedlick D, Marcova L. 1997. Plastic neuronal remodeling is impaired in patients with Alzheimer's disease carrying apolipoprotein epsilon 4 allele. J Neurosci 17:516–529.
- Arendt T, Bruckner MK, Gertz HJ, Marcova L. 1998a. Cortical distribution of neurofibrillary tangles in Alzheimer's disease matches the pattern of neurons that retain their capacity of plastic remodeling in the adult brain. Neuroscience 83:991–1002.
- Arendt T, Holzer M, Gartner U. 1998b. Neuronal expression of cycline dependent kinase inhibitors of the INK4 family in Alzheimer's disease. J Neural Transm 105:949–960.
- Arendt T. 2000. Alzheimer's disease as a loss of differentiation control in a subset of neurons that retain immature features in the adult brain. Neurobiol Aging 21:783–796.
- Arendt T. 2001a. Alzheimer's disease as a disorder of mechanisms underlying structural brain self-organization. Neuroscience 102:723–765.
- Arendt T. 2001b. Disturbance of neuronal plasticity is a critical pathogenetic event in Alzheimer's disease. Int J Dev Neurosci 19:231–245.
- Ashford JW, Jarvik L. 1985. Alzheimer's disease: does neuron plasticity predispose to axonal neurofibrillary degeneration? N Engl J Med 313:388–389.
- Ashford JW, Shan M, Butler S, Rajasekar A, Schmitt FA. 1995. Temporal quantification of Alzheimer's disease severity: 'time index' model. Dementia 6:269–280.
- Ashford JW, Mattson M, Kumar V. 1998a. Neurobiological systems disrupted by Alzheimer's disease and molecular biological theories of vulnerability. In: Kumar V, Eisdorer C, editors. Advances in the diagnosis and treatment of Alzheimer's disease. New York: Springer.
- Ashford JW, Soultanian NS, Zhang SX, Geddes JW. 1998b. Neuropil threads are colinear with MAP2 immunostaining in neuronal dendrites of Alzheimer brain. J Neuropathol Exp Neurol 57:972–978.
- Ashford JW, Shih WJ, Coupal J, Shetty R, Schneider A, Cool C, Aleem A, Kiefer VH, Mendiondo MS, Schmitt FA. 2000. Single SPECT measures of cerebral cortical perfusion reflect time-index estimation of dementia severity in Alzheimer's disease. J Nucl Med 41:57–64.
- Ashford JW, Mortimer JA. 2002. Non-familial Alzheimer's disease is mainly due to genetic factors. J Alzheimer's Dis 4:169–177.
- Aubert I, Ridet JL, Gage FH. 1995. Regeneration in the adult mammalian CNS: guided by development. Curr Opin Neurobiol 5:625–635.
- Aubert I, Ridet JL, Schachner M, Rougon G, Gage FH. 1998. Expression of L1 and PSA during sprouting and regeneration in the adult hippocampal formation. J Comp Neurol 399:1–19.
- Bae CY, Cho CY, Cho K, Hoon Oh B, Choi KG, Lee HS, Jung SP, Kim DH, Lee S, Choi GD, Cho H, Lee H. 2000. A double-blind, placebocontrolled, multicenter study of cerebrolysin for Alzheimer's disease. J Am Geriatr Soc 48:1566–1571.
- Bahr B, Vicente JS. 1998. Age-related phosphorylation and fragmentation events influence the distribution profiles of distinct tau isoforms in mouse brain. J Neuropathol Exp Neurol 57:111–121.
- Bales KR, Verina T, Ghetti B, Dodel RC, Du Y, Altstie L, Bender M, Hyslop P, Johnstone EM, Little SP, Cummins DJ, Piccardo P, Paul SM. 1997. Lack of apolipoprotein E dramatically reduces amyloid β-peptide deposition. Nat Genet 17:263–264.
- Banati RB, Gehrmann J, Czech C, Monning U, Jones LL, Konig G, Beyreuther K, Kreutzberg GW. 1993. Early and rapid de novo synthesis of Alzheimer beta A4-amyloid precursor protein (APP) in activated microglia. Glia 9:199–210.

- Barger SW, Horster D, Furukawa K, Goodman Y, Krieglstein J, Mattson MP. 1995. Tumor necrosis factors  $\alpha$  and  $\beta$  protect neurons against amyloid  $\beta$ -peptide toxicity: evidence for involvement of a  $\kappa$ B-binding factor and attenuation of peroxide and Ca<sup>2+</sup> accumulation. Proc Natl Acad Sci USA 92:9328–9332.
- Barger SW, Mattson MP. 1996. Induction of neuroprotective κBdependent transcription by secreted forms of the Alzheimer's betaamyloid precursor. Brain Res Mol Brain Res 40:116–126.
- Baron-Van Evercooren A, Kleinman HK, Ohno S, Marangos P, Schwartz JP, Dubois-Dalcq ME. 1982. Nerve growth factor, laminin, and fibronectin promote neurite growth in human fetal sensory ganglia cultures. J Neurosci Res 8:179–193.
- Barres BA, Smith SJ. 2001. Neurobiology. Cholesterol–making or breaking the synapse. Science 294:1296–1297.
- Bartlett PF. 1982. Pluripotential hemopoietic stem cells in adult mouse brain. Proc Natl Acad Sci USA 79:2722–2725.
- Baskerville KA, Schweitzer JB, Herron P. 1997. Effects of cholinergic depletion on experience-dependent plasticity in the cortex of the rat. Neuroscience 80:1159–1169.
- Bechmann I, Nitsch R. 1997. Astrocytes and microglial cells incorporate degenerating fibers following entorhinal lesion: a light, confocal, electron microscopical study using a phagocytosis-dependent labeling technique. Glia 20:145–154.
- Beeson JG, Shelton ER, Chan HW, Gage FH. 1994. Age and damage induced changes in amyloid protein precursor immunohistochemistry in the rat brain. J Comp Neurol 342:69–77.
- Bellosta S, Nathan BP, Orth M, Dong LM, Mahley RW, Pitas RE. 1995. Stable expression and secretion of apolipoproteins E3 and E4 in mouse neuroblastoma cells produces differential effects on neurite outgrowth. J Biol Chem 270:27063–27071.
- Benowitz LI, Routtenberg A. 1997. GAP-43: an intrinsic determinant of neuronal development and plasticity. Trends Neurosci 20:84–91.
- Benzing WC, Mufson EJ, Armstrong DM. 1993. Alzheimer's disease-like dystrophic neurites characteristically associated with senile plaques are not found within other neurodegenerative diseases unless amyloid betaprotein deposition is present. Brain Res 606:10–18.
- Berezovska O, McLean P, Knowles R, Frosh M, Lu FM, Lux SE, Hyman BT. 1999. Notch1 inhibits neurite outgrowth in postmitotic primary neurons. Neuroscience 93:433–439.
- Bertoni-Freddari C. 1988. Age-dependent deterioration of neuronal membranes and the pathogenesis of Alzheimer's disease: a hypothesis. Med Hypotheses 25:147–149.
- Bertoni-Freddari C, Fattoretti P, Meier-Ruge W, Ulrich J. 1989. Computer-assisted morphometry of synaptic plasticity during aging and dementia. Pathol Res Pract 185:799–802.
- Bi X, Yong AP, Zhou J, Ribak CE, Lynch G. 2001. Rapid induction of intraneuronal neurofibrillary tangles in apolipoprotein E-deficient mice. Proc Natl Acad Sci USA 98:8832–8837.
- Bjornson CR, Rietze RL, Reynolds BA, Magli MC, Vescovi AL. 1999. Turning brain into blood: a hematopoietic fate adopted by adult neural stem cells in vivo. Science 283:534–537.
- Black MM, Slaughter T, Moshiach S, Obrocka M, Fischer I. 1996. Tau is enriched on dynamic microtubules in the distal region of growing axons. J Neurosci 16:3601–3619.
- Blacker D, Haines JL, Rodes L, Terwedow H, Go RCP, Harrell LE, Perry RT, Bassett SS, Chase G, Meyers D, Albert MS, Tanzi R. 1997. ApoE-4 and age at onset of Alzheimer's disease: the NIMH genetics initiative. Neurology 48:139–147.
- Blanco G, Diaz H, Carrer HF, Beauge L. 1990. Differentiation of rat hippocampal neurons induced by estrogen in vitro: effects on neuritogenesis and NaK-ATPase activity. J Neurosci Res 27:47–54.
- Boissiere F, Hunot S, Faucheux B, Duyckaerts C, Hauw JJ, Agid Y, Hirsch EC. 1997. Nuclear translocation of NF-κB in cholinergic neurons of patients with Alzheimer's disease. Neuroreport 8:2849–2852.

- Bondolfi L, Calhoun M, Ermini F, Kuhn HG, Wiederhold KH, Walker L, Staufenbiel M, Jucker M. 2002. Amyloid-associated neuron loss and gliogenesis in the neocortex of amyloid precursor protein transgenic mice. J Neurosci 22:515–522.
- Bowen RL. 2001. Sex hormones, amyloid protein, and Alzheimer disease. JAMA 286:790–791.
- Boyles JK, Zoellner CD, Anderson LJ, Kosik LM, Pitas RE, Weisgraber KH, Hui DY, Mahley RW, Gebicke-Haerter PJ, Ignatius MJ, Shooter EM. 1989. A role for apolipoprotein E, apolipoprotein A-I, and low density lipoprotein receptors in cholesterol transport during regeneration and remyelination of the rat sciatic nerve. J Clin Invest 83:1015–1031.
- Braak H, Braak E. 1988. Chapter 2. In: Histology and histopathology of the aging brain. Basel: Karger.
- Braak H, Braak E. 1991. Neuropathological staging of Alzheimer-related changes. Acta Neuropathol (Berl) 82:239–259.
- Braak H, Braak E. 1996. Development of Alzheimer-related neurofibrillary changes in the neocortex inversely recapitulates cortical myelogenesis. Acta Neuropathol (Berl) 92:197–201.
- Braak H, Braak E. 1997. Diagnostic criteria for neuropathologic assessment of Alzheimer's disease. Neurobiol Aging 18(Suppl):85–88.
- Brake WG, Alves SE, Dunlop JC, Lee SJ, Bulloch K, Allen PB, Greengard P, McEwen BS. 2001. Novel target sites for estrogen action in the dorsal hippocampus: an examination of synaptic proteins. Endocrinology 142: 1284–1289.
- Breitner JCS, Wyse BW, Anthony JC, Welsh-Bohmer KA, Steffens DC, Norton MC, Tschanz JT, Plassman BL, Meyer MR, Skoog I. 1999. APOE-epsilon 4 count predicts age when prevalence of AD increases, then declines–The Cache County Study. Neurology 53:321–323.
- Bretsky PM, Buckwalter JG, Seeman TE, Miller CA, Poirier J, Schellenberg GD, Finch CE, Henderson VW. 1999. Evidence for an interaction between apolipoprotein E genotype, gender, and Alzheimer disease. Alzheimer Dis Assoc Disord 13:216–221.
- Brewer GJ. 1999. Regeneration and proliferation of embryonic and adult rat hippocampal neurons in culture. Exp Neurol 159:237–247.
- Brewer GJ. 2000. Neuronal plasticity and stressor toxicity during aging. Exp Gerontol 35:1165–1183.
- Brewer GJ, Espinosa J, McIlhaney MP, Pencek TP, Kesslak JP, Cotman C, Viel J, McManus DC. 2001. Culture and regeneration of human neurons after brain surgery. J Neurosci Methods 107:15–23.
- Brinton RD. 2001. Cellular and molecular mechanisms of estrogen regulation of memory function and neuroprotection against Alzheimer's disease: recent insights and remaining challenges. Learn Mem 8:121–133.
- Brion JP, Octave JN, Couck AM. 1994. Distribution of the phosphorylated microtubule-associated protein tau in developing cortical neurons. Neuroscience 63:895–909.
- Broe GA, Grayson DA, Creasey HM, Waite LM, Casey BJ, Bennett HP, Brooks WS, Halliday GM. 2000. Anti-inflammatory drugs protect against Alzheimer disease at low doses. Arch Neurol 57:1586–1591.
- Bronfman FC, Soto C, Tapia L, Tapia V, Inestrosa NC. 1996. Extracellular matrix regulates the amount of the beta-amyloid precursor protein and its amyloidogenic fragments. J Cell Physiol 166:360–369.
- Brose N. 1999. Synaptic cell adhesion proteins and synaptogenesis in the mammalian central nervous system. Naturwissenschafen 86:516-524.
- Bruce AJ, Sakhi S, Schreiber SS, Baudry M. 1995. Development of kainic acid and *N*-methyl-D-aspartic acid toxicity in organotypic hippocampal cultures. Exp Neurol 132:209–219.
- Brunelli MP, Kowal NW, Lee JM, McKee AC. 1991. Synaptophysin immunoreactivity is depleted in cortical laminae with dense dystrophic neurites and neurofibrillary tangles. J Neuropathol 50:315.
- Bruses JL, Chauvet N, Rutishauser U. 2001. Membrane lipid rafts are necessary for the maintenance of the  $(\alpha)$ 7 nicotinic acetylcholine receptor in somatic spines of ciliary neurons. J Neurosci 21:504–512.

- Brustle O, Jones KN, Learish RD, Karram K, Choudhary K, Wiestler OD, Duncan ID, McKay RD. 1999. Embryonic stem cell-derived glial precursors: a source of myelinating transplants. Science 285:754–756.
- Buchs PA, Stoppini L, Muller D. 1993. Structural modifications associated with synaptic development in area CA1 of rat hippocampal organotypic cultures. Brain Res Dev Brain Res 71:81–91.
- Bulinski JC, Ohm T, Roder H, Spruston N, Turner DA, Wheal HV. 1998. Changes in dendritic structure and function following hippocampal lesions: correlations with developmental events? Prog Neurobiol 55:641–650.
- Busciglio J, Ferreira A, Steward O, Caceres A. 1987. An immunocytochemical and biochemical study of the microtubule-associated protein Tau during post-lesion afferent reorganization in the hippocampus of adult rats. Brain Res 419:244–252.
- Buttini M, Akeefe H, Lin C, Mahley RW, Pitas RE, Wyss-Coray T, Mucke L. 2000. Dominant negative effects of apolipoprotein E4 revealed in transgenic models of neurodegenerative disease. Neuroscience 97:207–210.
- Buttini M, Orth M, Bellosta S, Akeefe H, Pitas RE, Wyss-Coray T, Mucke L, Mahley RW. 1999. Expression of human apolipoprotein E3 or E4 in the brains of apoE -/- mice: isoform-specific effects on neurodegeneration. J Neurosci 19:4867–4880.
- Buxbaum JD, Cullen EI, Friedhoff LT. 2002. Pharmacological concentrations of the HMG-CoA reductase inhibitor lovastatin decrease the formation of the Alzheimer  $\beta$ -amyloid peptide in vitro and in patients. Front Biosci 7:50–59.
- Cai D, Shen Y, De Bellard M, Tang S, Filbin MT. 1999. Prior exposure to neurotrophins blocks inhibition of axonal regeneration by MAG and myelin via a cAMP-dependent mechanism. Neuron 22:89–101.
- Calabrese EJ. 2001. Amyloid beta-peptide: biphasic dose responses. Crit Rev Toxicol 31:605-606.
- Calhoun ME, Wiederhold KH, Abramowski D, Phinney AL, Probst A, Sturchler-Pierrat C, Staufenbiel M, Sommer B, Jucker M. 1998. Neuron loss in APP transgenic mice. Nature 395:755–756.
- Cambon K, Davies HA, Stewart MG. 2000. Synaptic loss is accompanied by an increase in synaptic area in the dentate gyrus of aged human apolipoprotein E4 transgenic mice. Neuroscience 97:685–692.
- Cameron HA, McKay RD. 1999. Restoring production of hippocampal neurons in old age. Nat Neurosci 2:894–897.
- Campenot RB. 1994. NGF and the local control of nerve terminal growth. J Neurobiol 25:599–611.
- Campenot RB, Eng H. 2000. Protein synthesis in axons and its possible functions. J Neurocytol 29:793–798.
- Canning DR, Hoke A, Malemud CJ, Silver J. 1996. A potent inhibitor of neurite outgrowth that predominates in the extracellular matrix of reactive astrocytes. Int J Dev Neurosci 14:153–175.
- Capsoni S, Ugolini G, Comparini A, Ruberti F, Berardi N, Cattaneo A. 2000. Alzheimer-like neurodegeneration in aged anti nerve growth factor transgenic mice. Proc Natl Acad Sci USA 97:6826–6831.
- Cassell MD, Brown MW. 1984. Distribution of Timm's stain in the nonsulphide-perfused human hippocampal formation. J Comp Neurol 222:461–471.
- Chang FL, Greenough WT. 1984. Transient and enduring morphological correlates of synaptic activity and efficacy change in the rat hippocampal slice. Brain Res 309:35–46.
- Chao CC, Hu S, Peterson PK. 1996. Glia: the not so innocent bystanders. J Neurovirol 2:234–239.
- Chapman PF, Falinska AM, Knevett SG, Ramsay MF. 2001. Genes, models and Alzheimer's disease. Trends Genet 17:254–261.
- Chapman PF, White GL, Jones MW, Cooper-Blacketer D, Marshall VJ, Irizarry M, Younkin L, Good MA, Bliss TV, Hyman BT, Younkin ST, Hsiao K. 1999. Impaired synaptic plasticity and learning in amyloid precursor protein transgenic mice. Nat Neurosci 2:271–276.
- Chauvet N, Apert C, Dumoulin A, Epelbaum J, Alonso G. 1997. Mab22C11 antibody to amyloid precursor protein recognizes a protein associated with specific astroglial cells of the rat central nervous system

characterized by their capacity to support axonal outgrowth. J Comp Neurol 377:550-564.

- Chawen JA, Torres-Aleman I, Garcia-Segura LM. 1992. Trophic effects of estradiol on fetal rat hypothalamic neurons. Neuroendocrinology 56:895–901.
- Chen C, Tonegawa S. 1997. Molecular genetic analysis of synaptic plasticity, activity-dependent neural development, learning, memory in the mammalian brain. Annu Rev Neurosci 20:157–184.
- Chen G, Chen KS, Knox J, Inglis J, Bernard A, Martin SJ, Justice A, McConlogue L, Games D, Freedman SB, Morris R.G. 2000. A learning deficit related to age and beta-amyloid plaques in a mouse model of Alzheimer's disease. Nature 408:975–979.
- Chen KS, Nishimura MC, Armanini MP, Crowley C, Spencer SD, Phillips HS. 1997. Disruption of a single allele of the nerve growth factor gene results in atrophy of basal forebrain cholinergic neurons and memory deficits. J Neurosci 17:7288–7296.
- Chen QS, Kagan BL, Hirakura Y, Xie CW. 2000. Impairment of hippocampal long-term potentiation by Alzheimer amyloid beta-peptides. J Neurosci Res 60:65–72.
- Chen Y, Lomnitski L, Michaelson DM, Shohami E. 1997. Motor and cognitive deficits in apolipoprotein E-deficient mice after closed head injury. Neuroscience 80:1255–1262.
- Chochina SV, Avdulov NA, Igbavboa U, Cleary JP, O'Hare EO, Wood WG. 2001. Amyloid beta-peptide1-40 increases neuronal membrane fluidity: role of cholesterol and brain region. J Lipid Res 42:1292-1297.
- Connolly ES, Winfree CJ, Stern DM, Solomon RA, Pinsky DJ. 1996. Exacerbation of stroke outcome in apolipoprotein E null mice; role of abnormal lipid metabolism in the pathogenesis of tissue injury following cerebral ischemia. Stroke 27:174A.
- Constantine-Paton M. 1990. NMDA receptor as a mediator of activitydependent synaptogenesis in the developing brain. Cold Spring Harb Symp Quant Biol 55:431–443.
- Corbo RM, Scacchi R. 1999. Apolipoprotein E (APOE) allele distribution in the world. Is APOE\*4 a 'thrifty' allele? Ann Hum Genet 63:301–310.
- Corder EH, Saunders AM, Risch NJ, Strittmatter WJ, Schmechel DE, Gaskell PC, Rimmler JB, Locke PA, Conneally PM, Schmader KE, Small GW, Roses AD, Haines JL, Pericak-Vance MA. 1994. Protective effect of apolipoprotein E type 2 allele for late onset Alzheimer disease. Nat Genet 7:180–183.
- Corder EH, Saunders AM, Strittmatter WJ, Schmechel DE, Gaskell PC, Small GW, Roses AD, Haines JL, Pericak-Vance MA. 1993. Gene dose of apolipoprotein E type 4 allele and the risk of Alzheimer's disease in late onset families. Science 261:921–923.
- Cotman CW, Nieto-Sampedro M. 1984. Cell biology of synaptic plasticity. Science 225:1287–1294.
- Cotman CW, Geddes JW, Kahle JS. 1990. Axon sprouting in the rodent and Alzheimer's disease brain: a reactivation of developmental mechanisms? Prog Brain Res 83:427–434.
- Cotman CW, Cummings BJ, Whitson JS. 1991. The role of misdirected plasticity in plaque biogenesis and Alzheimer's disease pathology. In: Hefti F, Brachet B, Christen WY, editors. Growth factors and Alzheimer's disease. Berlin: Springer-Verlag. p 222–233.
- Cotman CW, Cummings BJ, Pike CJ. 1993. Molecular cascades in adaptive versus pathological plasticity. New York: Raven Press.
- Cotman CW, Anderson AJ. 2000. The brain's microenvironment, early functional loss, the conversion to Alzheimer's disease. Ann N Y Acad Sci 924:112–116.
- Cremer H, Chazal G, Goridis C, Represa A. 1997. NCAM is essential for axonal growth and fasciculation in the hippocampus. Mol Cell Neurosci 8:323–335.
- Crino PB, Eberwine J. 1996. Molecular characterization of the dendritic growth cone: regulated mRNA transport and local protein synthesis. Neuron 17:1173–1187.

- Crutcher KA. 1986. The role of growth factors in neuronal development and plasticity. CRC Crit Rev Clin Neurobiol 2:297–333.
- Cullen WK, Suh YH, Anwyl R, Rowan MJ. 1997. Block of LTP in rat hippocampus in vivo by beta-amyloid precursor protein fragments. Neuroreport 8:3213–3217.
- Cyr M, Calon F, Morissette M, Grandbois M, Di Paolo T, Callier S. 2000. Drugs with estrogen-like potency and brain activity: potential therapeutic application for the CNS. Curr Pharmacol Des 6:1287–1312.
- Danik M, Poirier J. 1998. Apolipoprotein E and neuronal plasticity following experimental de-afferentation and in Alzheimer's disease. Biochem Soc Trans 26:262–266.
- Danscher G. 1981. Histochemical demonstration of heavy metals: a revised version of the sulphide silver method suitable for both light and electron microscopy. Histochemistry 71:1–16.
- David JP, Ghozali F, Fallet-Bianco C, Wattez A, Delaine S, Boniface B, Di Menza C, Delacourte A. 1997. Glial reaction in the hippocampal formation is highly correlated with aging in human brain. Neurosci Lett 235:53–56.
- Davies CA, Mann DM, Sumpter PQ, Yates PO. 1987. A quantitative morphometric analysis of the neuronal and synaptic content of the frontal and temporal cortex in patients with Alzheimer's disease. J Neurol Sci 78:151–164.
- Davies SJ, Fitch MT, Memberg SP, Hall AK, Raisman G, Silver J. 1997. Regeneration of adult axons in white matter tracts of the central nervous system. Nature 390:680–683.
- Dawson TM, Dawson VL, Snyder SH. 1992. A novel neuronal messenger in brain: the free radical, nitric oxide. Ann Neurol 32:297–311.
- Dawson TM, Snyder SH. 1994. Gases as biological messengers: nitric oxide and carbon monoxide in the brain. J Neurosci 14:5147–5159.
- Dawson VL, Brahmbhatt HP, Mong JA, Dawson TM. 1994. Expression of inducible nitric oxide synthase causes delayed neurotoxicity in primary mixed neuronal-glial cortical cultures. Neuropharmacology 33:1425– 1430.
- Dawson TM, Sasaki M, Gonzalez-Zulueta M, Dawson VL. 1998. Regulation of neuronal nitric oxide synthase and identification of novel nitric oxide signaling pathways. Prog Brain Res 118:3–11.
- de la Monte SM, Ng SC, Hsu DW. 1995. Aberrant GAP-43 gene expression in Alzheimer's disease. Am J Pathol 147:934–946.
- De Mattos RD, Curtiss LK, Williams DL. 1998. A minimally lipidated form of cell-derived apolipoprotein E exhibits isoform-specific stimulation of neurite outgrowth in the absence of exogenous lipids or lipoproteins. J Biol Chem 273:4206–4212.
- De Vreese LP, Neri M, Fioravanti M, Belloi L, Zanetti O. 2001. Memory rehabilitation in Alzheimer's disease: a review of progress. Int J Geriatr Psychiatry 16:794–809.
- DeKosky ST, Scheff SW. 1990. Synapse loss in frontal cortex biopsies in Alzheimer's disease: correlation with cognitive severity. Ann Neurol 27:457–464.
- DeKosky ST, Scheff SW, Styren SD. 1996. Structural correlates of cognition in dementia: quantification and assessment of synapse change. Neurodegeneration 5:417–421.
- Delacourte A, Sergeant N, Wattez A, Gauvreau D, Robitaille Y. 1998. Vulnerable neuronal subsets in Alzheimer's and Pick's disease are distinguished by their tau isoform distribution and phosphorylation. Ann Neurol 43:193–204.
- Dewachter I, Reverse D, Caluwaerts N, Ris L, Kuiperi C, Van Den Haute C, Spittaels K, Umans L, Serneels L, Thiry E, Moechars D, Mercken M, Godaux E, Van Leuven F. 2002. Neuronal deficiency of presenilin 1 inhibits amyloid plaque formation and corrects hippocampal long-term potentiation but not a cognitive defect of amyloid precursor protein [V717I] transgenic mice. J Neurosci 22:3445–3453.
- Dickson DW, Crystal HA, Bevona C, Honer W, Vincent I, Davies P. 1995. Correlations of synaptic and pathological markers with cognition of the elderly. Neurobiol Aging 16:285–298; discussion 298–304.

- Dietschy JM, Turley SD. 2001. Cholesterol metabolism in the brain. Curr Opin Lipidol 12:105–112.
- Distl R, Meske V. 2000. Microdensitofluorometrically determined free cholesterol is higher in tangled-bearing neurons than in tangle-free neurons. Neurobiol Aging 21(Suppl):21.
- Dodart JC, Bales KR, Gannon KS, Greene SJ, DeMattos RB, Mathis C, DeLong CA, Wu S, Wu X, Holtzman DM, Paul SM. 2002. Immunization reverses memory deficits without reducing brain A $\beta$  burden in Alzheimer's disease model. Nat Neurosci 5:452–457.
- Dowjat WK, Wisniewski T, Efthimiopoulos S, Wisniewski HM. 1999. Inhibition of neurite outgrowth by familial Alzheimer's disease-linked presenilin-1 mutations. Neurosci Lett 267:141–144.
- Downen M, Zhao ML, Lee P, Weidenheim KM, Dickson DW, Lee SC. 1999. Neuronal nitric oxide synthase expression in developing and adult human CNS. J Neuropathol Exp Neurol 58:12–21.
- Drubin DG, Feinstein SC, Shooter EM, Kirschner MW. 1985. Nerve growth factor-induced neurite outgrowth in PC12 cells involves the coordinate induction of microtubule assembly and assembly-promoting factors. J Cell Biol 101:1799–1807.
- Einstein G, Buranosky R, Crain BJ. 1994. Dendritic pathology of granule cells in Alzheimer's disease is unrelated to neuritic plaques. J Neurol Sci 14:5077–5088.
- Emilien G, Beyreuther K, Masters CL, Maloteaux JM. 2000. Prospects for pharmacological intervention in Alzheimer disease. Arch Neurol 57:454– 459.
- Engert F, Bonhoeffer T. 1999. Dendritic spine changes associated with hippocampal long-term synaptic plasticity. Nature 399:66-70.
- Erickson JC, Sewell AK, Jensen LT, Winge DR, Palmiter RD. 1994. Enhanced neurotrophic activity in Alzheimer's disease cortex is not associated with down-regulation of metallothionein-III (GIF). Brain Res 649:297–304.
- Eriksdotter Jonhagen M, Nordberg A, Amberla K, Backman L, Ebendal T, Meyerson B, Olson L, Seiger, Shigeta M, Theodorsson E, Viitanen M, Winblad B, Wahlund LO. 1998. Intracerebroventricular infusion of nerve growth factor in three patients with Alzheimer's disease. Dement Geriatr Cogn Disord 9:246–257.
- Eriksson PS, Perfilieva E, Bjork-Eriksson T, Alborn AM, Nordborg C, Peterson DA, Gage FH. 1998. Neurogenesis in the adult human hippocampus. Nat Med 4:1313–1317.
- Ernfors P, Lindefors N, Chan-Palay V, Persson H. 1990. Cholinergic neurons of the nucleus basalis express elevated levels of nerve growth factor receptor mRNA in senile dementia of the Alzheimer's type. Dementia 1:138–145.
- Esclaire F, Terro F, Yardin C, Hugon J. 1998. Neuronal apoptosis is associated with a decrease in tau mRNA expression. Neuroreport 9:1173–1177.
- Esmaeli-Azad B, McCarty JH, Feinstein SC. 1994. Sense and antisense transfection analysis of tau function: tau influences net microtubule assembly, neurite outgrowth and neuritic stability. J Cell Sci 107:869–879.
- Espinosa B, Zenteno R, Mena R, Robitaille Y, Zenteno E, Guevara J. 2001. O-Glycosylation in sprouting neurons in Alzheimer disease, indicating reactive plasticity. J Neuropathol Exp Neurol 60:441–448.
- Fagan AM, Bu G, Sun Y, Daugherty A, Holtzman DM. 1996. Apolipoprotein E-containing high density lipoprotein promotes neurite outgrowth and is a ligand for the low density lipoprotein receptor-related protein. J Biol Chem 271:30121–30125.
- Fagan AM, Murphy BA, Patel SN, Kilbridge JF, Mobley WC, Bu G, Holtzman DM. 1998. Evidence for normal aging of the septohippocampal cholinergic system in apoE (-/-) mice but impaired clearance of axonal degeneration products following injury. Exp Neurol 151:314–325.
- Fahnestock M, Scott SA, Jette N, Weingartner JA, Crutcher KA. 1996. Nerve growth factor mRNA and protein levels measured in the same tissue from normal and Alzheimer's disease parietal cortex. Brain Res Mol Brain Res 42:175–178.

- Fan QW, Yu W, Senda T, Yanagisawa K, Michikawa M. 2001. Cholesterol-dependent modulation of tau phosphorylation in cultured neurons. J Neurochem 76:391–400.
- Fan QW, Yu W, Gong JS, Zou K, Sawamura N, Senda T, Yanagisawa K, Michikawa M. 2002. Cholesterol-dependent modulation of dendrite outgrowth and microtubule stability in cultured neurons. J Neurochem 80:178–190.
- Farlow MR, Lahiri DK, Poirier J, Davignon J, Schneider L, Hui SL. 1998. Treatment outcome of tacrine therapy depends on apolipoprotein genotype and gender of the subjects with Alzheimer's disease. Neurology 50:669–677.
- Farooqui AA, Horrocks LA. 1994. Involvement of glutamate receptors, lipases, and phospholipases in long-term potentiation and neurodegeneration. J Neurosci Res 38:6–11.
- Farrer LA, Cupples LA, van Duijn CM, Kurz A, Zimmer R, Muller U, Green RC, Clarke V, Shoffner J, Wallace DC, et al. 1995. Apolipoprotein E genotype in patients with Alzheimer's disease: implications for the risk of dementia among relatives. Ann Neurol 38:797–808.
- Farrer LA, Cupples LA, Haines JL, Hyman B, Kukull WA, Myers RH, van Duijin CM. 1997. Effects of age, sex, and ethnicity on the association between apolipoprotein E genotype and Alzheimer disease. A metaanalysis. APOE and Alzheimer Disease Meta Analysis Consortium. JAMA 278:1349–1356.
- Fernandes MA, Proenca MT, Nogueira AJ, Oliveira LM, Santiago B, Santana I, Oliveira CR. 1999. Effects of apolipoprotein E genotype on blood lipid composition and membrane platelet fluidity in Alzheimer's disease. Biochim Biophys Acta 1454:89–96.
- Ferreira A, Caceres A. 1991. Estrogen-enhanced neurite growth: evidence for a selective induction of tau and stable microtubules. J Neurosci 11:392–400.
- Ferrer I, Aymami A, Rovira A, Grau Veciana JM. 1983. Growth of abnormal neurites in atypical Alzheimer's disease. A study with the Golgi method. Acta Neuropathol (Berl) 59:167–170.
- Ferrer I, Guionnet N, Cruz-Sanchez F, Tunon T. 1990. Neuronal alterations in patients with dementia: a Golgi study on biopsy samples. Neurosci Lett 114:11–16.
- Feussner G, Dobmeyer J, Grone HJ, Lohmer S, Wohlfeil S. 1996. A 10-bp deletion in the apolipoprotein epsilon gene causing apolipoprotein E deficiency and severe type III hyperlipoproteinemia. Am J Hum Genet 58:281–291.
- Figueroa DJ, Morris JA, Ma L, Kandpal G, Chen E, Li YM, Austin CP. 2002. Presenilin-dependent gamma-secretase activity modulates neurite outgrowth. Neurobiol Dis 9:49–60.
- Fischer O. 1907. Miliare necrosen mit drusigen wucherungen der neurofibrillen, eine regelmassige veranderung der hirnrinde bei seniler demenz. Psychiatrie Neurol 22:361–372.
- Fisher A, Michaelson DM, Brandeis R, Haring R, Chapman S, Pittel Z. 2000. M1 muscarinic agonists as potential disease-modifying agents in Alzheimer's disease. Rationale and perspectives. Ann N Y Acad Sci 920:315–320.
- Flax JD, Aurora S, Yang C, Simonin C, Wills AM, Billinghurst LL, Jendoubi M, Sidman RL, Wolfe JH, Kim SU, Snyder EY. 1998. Engraftable human neural stem cells respond to developmental cues, replace neurons, and express foreign genes. Nat Biotechnol 16:1033–1039.
- Flood D, Coleman PD. 1990. Hippocampal plasticity in normal aging and decreased plasticity in Alzheimer's disease. Prog Brain Res 83:435–443.
- Flood DG, Buell SJ, Defiore CH, Horwitz GJ, Coleman PD. 1985. Age-related dendritic growth in dentate gyrus of human brain is followed by regression in the 'oldest old.' Brain Res 345:366–368.
- Forstermann U, Gath I, Schwarz P, Closs EI, Kleinert H. 1995. Isoforms of nitric oxide synthase: properties, cellular distribution and expressional control. Biochem Pharmacol 50:1321–1332.
- Foster TC, Norris CM. 1997. Age-associated changes in Ca2+ dependent

processes: relation to hippocampal synaptic plasticity. Hippocampus 7:602-612.

- Francis PT, Palmer AM, Snape M, Wilcock GK. 1999. The cholinergic hypothesis of Alzheimer's disease: a review of progress. J Neurol Neurosurg Psychiatry 66:137–147.
- Frederickson RC. 1992. Astroglia in Alzheimer's disease. Neurobiol Aging 13:239–253.
- Freir DB, Holscher C, Herron CE. 2001. Blockade of long-term potentiation by beta-amyloid peptides in the CA1 region of the rat hippocampus in vivo. J Neurophysiol 85:708–713.
- Frotscher M. 1992. Application of the Golgi/electron microscopy technique for cell identification in immunocytochemical, retrograde labeling, and developmental studies of hippocampal neurons. Microsc Res Tech 23:306–323.
- Furukawa K, Guo Q, Schellenberg GD, Mattson MP. 1998. Presenilin-1 mutation alters NGF-induced neurite outgrowth, calcium homeostasis, and transcription factor (AP-1) activation in PC12 cells. J Neurosci Res 52:618–624.
- Gaarskjaer FB. 1986. The organization and development of the hippocampal mossy fiber system. Brain Res Brain Res Rev 11:335–357.
- Gage FH, Olejnizak P, Armstrong DM. 1988. Astrocytes are important for sprouting in the septohippocampal circuit. Exp Neurol 102:2–13.
- Gage FH, Coates PW, Palmer TD, Kuhn HG, Fisher LJ, Suhonen JO, Peterson DA, Suhr ST, Ray J. 1995. Survival and differentiation of adult neuronal progenitor cells transplanted to the adult brain. Proc Natl Acad Sci USA 92:11879–11883.
- Gagliardini V, Dusart I, Fankhauser C. 2000. Absence of GAP-43 can protect neurons from death. Mol Cell Neurosci 16:27–33.
- Galasko D. 2001. New approaches to diagnose and treat Alzheimer's disease: a glimpse of the future. Clin Geriatr Med 17:393-410.
- Gall C, Rose G, Lynch G. 1979. Proliferative and migratory activity of glial cells in the partially deafferented hippocampus . J Comp Neurol 183: 539–550.
- Gallo G, Letourneau PC. 1999. Different contributions of microtubule dynamics and transport to the growth of axons and collateral sprouts. J Neurosci 19:3860–3873.
- Games D, Adams D, Alessandrini R, Barbour R, Berthelette P, Blackwell C, Carr T, Clemens J, Donaldson T, Gillespie F, Guido T, Hagoplan S, Johnson-Wood K, Khan K, Lee M, Leibowitz P, Lieberburg I, Little S, Masliah E, McConlogue L, Montoya-Zavala M, Mucke L, Paganini L, Penniman E, Power M, Schenk D, Seubert P, Snyder B, Soriano F, Tan H, Vitale J, Wadsworth S, Wolozin B, Zhao J. 1995. Alzheimer-type neuropathology in transgenic mice overexpressing V717F beta-amyloid precursor protein. Nature 373:523–527.
- Gandy S, Stafstrom C, Turner B, Sweeney D, Breslow J, Greengard P, Smith J, Quimet C. 1995. Aged apoE-deficient mice: abnormal learning and glial markers but no obvious neuropathology. Soc Neurosci Abstr 21:5.
- Garcia-Segura LM, Azcoitia I, Don Carlos LL. 2001. Neuroprotection by estradiol. Prog Neurobiol 63:29–60.
- Garcia-Segura LM, Chowen JA, Perdiz A, Naftolin F. 1994. Gonadal hormones as promoters of structural synaptic plasticity: cellular mechanisms. Prog Neurobiol 44:279–307.
- Gartner U, Holzer M, Arendt T. 1999. Elevated expression of p21ras is an early event in Alzheimer's disease and precedes neurofibrillary degeneration. Neuroscience 91:1–5.
- Gartner U, Holzer M, Heumann R, Arendt T. 1995. Induction of p21ras in Alzheimer pathology. Neuroreport 6:1441–1444.
- Gazzaley AH, Benson DL, Handley GW, Morrison JH. 1997. Differential subcellular regulation of NMDAR1 protein and mRNA in dendrites of dentate gyrus granule cells after perforant path transection. J Neurosci 17:2006–2017.
- Geddes JW, Monaghan DT, Cotman CW, Lott IT, Kim RC, Chui HC. 1985. Plasticity of hippocampal circuitry in Alzheimer's disease. Science 230:1179–1181.

- Geddes JW, Anderson KJ, Cotman CW. 1986. Senile plaques as aberrant sprout-stimulating structures. Exp Neurol 94:767–776.
- Geddes JW, Wilson MC, Miller FD, Cotman CW. 1990. Molecular markers of reactive plasticity. Adv Exp Med Biol 268:425-432.
- Geddes JW, Lundgren K, Kim YK. 1991. Aberrant localization of MAP5 immunoreactivity in the hippocampal formation in Alzheimer's disease. J Neurosci Res 30:183–191.
- Geddes JF, Vowels GH, Robinson SF, Sutcliffe JC. 1996. Neurofibrillary tangles, but not Alzheimer-type pathology, in a young boxer. Neuro-pathol Appl Neurobiol 22:12–16.
- Geinisman Y, Bonsdorff W, Teller A. 1977. Diminished axonal transport of glycoproteins in the senescent rat brain. Mech Ageing Dev 6:363–378.
- Gertz HJ, Cervus-Navarro J, Ewald V. 1987. The septo-hippocampal pathway in patients suffering from senile dementia of Alzheimer's type: evidence for neuronal plasticity? Neurosci Lett 76:228–232.
- Gertz HJ, Kruger H, Patt S, Cervos-Navarro J. 1990. Tangle-bearing neurons show more extensive dendritic trees than tangle-free neurons in area CA1 of the hippocampus in Alzheimer's disease. Brain Res 548:260– 266.
- Geula C, Mesulam MM. 1999. Cholinergic systems in Alzheimer's disease. In: Terry RD, Katzman R, Sisodia SS, Bick KL, editors. Alzheimer's disease. Philadelphia: Lippincott, Williams & Wilkins. p 269–292.
- Ghosh A, Greenberg ME. 1995. Calcium signaling in neurons: molecular mechanisms and cellular consequences. Science 268:239–247.
- Gibson PH. 1983. EM study of the numbers of cortical synapses in the brains of ageing people and people with Alzheimer-type dementia. Acta Neuropathol (Berl) 62:127–133.
- Gimpl G, Burger K, Fahrenholz F. 1997. Cholesterol as modulator of receptor function. Biochemistry 36:10959–10974.
- Goldberg JL, Barres BA. 2000. The relationship between neuronal survival and regeneration. Annu Rev Neurosci 23:579–612.
- Golde TE, Estus S, Younkin LH, Selkoe DJ, Younkin SG. 1992. Processing of the amyloid protein precursor to potentially amyloidogenic derivatives. Science 255:728–730.
- Golde TE, Younkin SG. 2001. Presenilins as therapeutic targets for the treatment of Alzheimer's disease. Trends Mol Med 7:264–269.
- Gonatas NK, Anderson W, Evangelista I. 1967. The contribution of altered synapses in the senile plaque: an electron microscopic study in Alzheimer's dementia. J Neuropathol Exp Neurol 26:25–39.
- Good M, Honey RC. 1997. Dissociable effects of selective lesions to hippocampal subsystems on exploratory behavior, contextual learning, and spatial learning. Behav Neurosci 111:487–493.
- Goodenough S, Engert S, Behl C. 2000. Testosterone stimulates rapid secretory amyloid precursor protein release from rat hypothalamic cells via the activation of the mitogen-activated protein kinase pathway. Neurosci Lett 296:49–52.
- Gordon I, Grauer E, Genis I, Sehayek E, Michaelson DM. 1995. Memory deficit and cholinergic impairments in apolipoprotein E-deficient mice. Neurosci Lett 199:1–4.
- Gordon-Weeks PR, Fischer I. 2000. MAP1B expression and microtubule stability in growing and regenerating axons. Microsc Res Tech 48:63–74.
- Gottfries CG, Karlsson I, Svennerholm L. 1996. Membrane components separate early-onset Alzheimer's disease from senile dementia of the Alzheimer type. Int Psychogeriatr 8:365–372.
- Gotthardt M, Trommsdorff M, Nevitt MF, Shelton J, Richardson JA, Stockinger W, Nimpf J, Herz J. 2000. Interactions of the low density lipoprotein receptor gene family with cytosolic adaptor and scaffold proteins suggest diverse biological functions in cellular communication and signal transduction. J Biol Chem 275:25616–25624.
- Gould E, Beylin A, Tanapat P, Reeves A, Shors TJ. 1999. Learning enhances adult neurogenesis in the hippocampal formation. Nat Neurosci 2:260–265.
- Grady CL. 1996. Age-related changes in cortical blood flow activation during perception and memory. Ann N Y Acad Sci 777:14–21.

- Grady MS, Jane JA, Steward O. 1989. Synaptic reorganization within the human central nervous system following injury. J Neurosurg 71:534–537.
- Greenough WT, West RW, DeVoogd TJ. 1978. Subsynaptic plate perforations: changes with age and experience in the rat. Science 202:1096– 1098.
- Gundersen GG, Cook TA. 1999. Microtubules and signal transduction. Curr Opin Cell Biol 11:81–94.
- Guo Q, Robinson N, Mattson MP. 1998. Secreted beta-amyloid precursor protein counteracts the proapoptotic action of mutant presenilin-1 by activation of NF- $\kappa$ B and stabilization of calcium homeostasis. J Biol Chem 273:12341–12351.
- Hamos JE, DeGennaro LJ, Drachman DA. 1989. Synaptic loss in Alzheimer's disease and other dementias. Neurology 39:355–361.
- Handelmann GE, Boyles JK, Weisgraber KH, Mahley RW, Pitas RE. 1992. Effects of apolipoprotein E,  $\beta$ -very low density lipoproteins, and cholesterol on the extension of neurites by rabbit dorsal root ganglion neurons in vitro. J Lipid Res 33:1677–1688.
- Hardman R, Evans DJ, Fellows L, Hayes B, Rupniak HT, Barnes JC, Higgins GA. 1997. Evidence for recovery of spatial learning following entorhinal cortex lesions in mice. Brain Res 758:187–200.
- Hardy J, Duff K, Hardy KG, Perez-Tur J, Hutton M. 1998. Genetic dissection of Alzheimer's disease and related dementias: amyloid and its relationship to tau. Nat Neurosci 1:355–358.
- Hartley DM, Walsh DM, Ye CP, Diehl T, Vasquez S, Vassilev PM, Teplow DB, Selkoe DJ. 1999. Protofibrillar intermediates of amyloid β-protein induce acute electrophysiological changes and progressive neurotoxicity in cortical neurons. J Neurosci 19:8876–8884.
- Hartmann T. 2001. Cholesterol, Aβ, and Alzheimer's disease. Trends Neurosci 24(Suppl):45-48.
- Hasselmo ME. 1997. A computational model of the progression of Alzheimer's disease. MD Comput 14:181–191.
- Hatanpaa K, Isaacs KR, Shirao T, Brady DR, Rapoport SI. 1999. Loss of proteins regulating synaptic plasticity in normal aging of the human brain and in Alzheimer disease. J Neuropathol Exp Neurol 58:637–643.
- Hebb DO. 1949. The organization of behaviour. New York: Wiley.
- Heininger K. 2000. A unifying hypothesis of Alzheimer's disease. IV. Causation and sequence of events. Rev Neurosci 11:213–328.
- Heinonen O, Soininen H, Sorvari H, Kosunen O, Paljarvi L, Koivisto E, Riekkinen PJ Sr. 1995. Loss of synaptophysin-like immunoreactivity in the hippocampal formation is an early phenomenon in Alzheimer's disease. Neuroscience 64:375–384.

Heintz N. 1993. Cell death and the cell cycle: a relationship between transformation and neurodegeneration? Trends Biochem Sci 18:157–159.

- Heisenberg CP, Cooper JD, Berke J, Sofroniew MV. 1994. NMDA potentiates NGF-induced sprouting of septal cholinergic fibres. Neuro-report 12:413–416.
- Helbecque N, Richard F, Cottel D, Neuman E, Guez D, Amouyel P. 1998. The very low density lipoprotein (VLDL) receptor is a genetic susceptibility factor for Alzheimer disease in a European Caucasian population. Alzheimer Dis Assoc Disord 12:368–371.
- Henderson CE. 1996. Role of neurotrophic factors in neuronal development. Curr Opin Neurobiol 6:64–70.
- Henderson VW, Paganini-Hill A, Miller BL, Elble RJ, Reyes PF, Shoupe D, McCleary CA, Klein RA, Hake AM, Farlow MR. 2000. Estrogen for Alzheimer's disease in women: randomized, double-blind, placebocontrolled trial. Neurology 54:295–301.
- Herdegen T, Skene P, Bahr M. 1997. The c-Jun transcription factor: bipotential mediator of neuronal death, survival and regeneration. Trends Neurosci 20:227–231.
- Herz J. 2001a. The LDL receptor gene family: (un)expected signal transducers in the brain. Neuron 29:571–581.
- Herz J. 2001b. Lipoprotein receptors: beacons to neurons? Trends Neurosci 24:193–195.

- Herz J, Strickland DK. 2001. LRP: a multifunctional scavenger and signaling receptor. J Clin Invest 108:779–784.
- Hess DT, Patterson SI, Smith DS, Skene JH. 1993. Neuronal growth cone collapse and inhibition of protein fatty acylation by nitric oxide. Nature 366:562–565.
- Hesse C, Larsson H, Fredman P, Minthon L, Andreasen N, Davidsson P, Blennow K. 2000. Measurement of apolipoprotein E (apoE) in cerebrospinal fluid. Neurochem Res 25:511–517.
- Higgins GA, Koh S, Chen KS, Gage FH. 1989. NGF induction of NGF receptor gene expression and cholinergic neuronal hypertrophy within the basal forebrain of the adult rat. Neuron 3:247–256.
- Ho L, Guo Y, Spielman L, Petrescu O, Haroutunian V, Purohit D, Czernik A, Yemul S, Aisen PS, Mohs R, Pasinetti GM. 2001. Altered expression of a-type but not b-type synapsin isoform in the brain of patients at high risk for Alzheimer's disease assessed by DNA microarray technique. Neurosci Lett 298:191–194.
- Hock C, Heese K, Muller-Spahn F, Hulette C, Rosenberg C, Otten U. 1998. Decreased trkA neurotrophin receptor expression in the parietal cortex of patients with Alzheimer's disease. Neurosci Lett 241:151–154.
- Hof PR, Morrison JH. 1994. The cellular basis of cortical disconnection in Alzheimer's disease and related dementing conditions. In: Terry RD, Katzman R, Bick KL, editors. The cellular basis of cortical disconnection in Alzheimer's disease and related dementing conditions. New York: Raven Press. p 197–229.
- Hoffman PW, Chernak JM. 1994. The rat amyloid precursor protein promoter contains two DNA regulatory elements which influence high level gene expression. Biochem Biophys Res Commun 201:610–617.
- Holcomb L, Gordon MN, McGowan E, Yu X, Benkovic S, Jantzen P, Wright K, Saad I, Mueller R, Morgan D, Sanders S, Zehr C, O'Campo K, Hardy J, Prada CM, Eckman C, Younkin S, Hsiao K, Duff K. 1998. Accelerated Alzheimer-type phenotype in transgenic mice carrying both mutant amyloid precursor protein and presenilin 1 transgenes. Nat Med 4:97–100.
- Holtzman DM, Kilbridge J, Li Y, Cunningham ET Jr, Lenn NJ, Clary DO, Reichardt LF, Mobley WC. 1995a. TrkA expression in the CNS: evidence for the existence of several novel NGF-responsive CNS neurons. J Neurosci 15:1567–1576.
- Holtzman DM, Pitas RE, Kilbridge J, Nathan B, Mahley RW, Bu G, Schwartz AL. 1995b. Low density lipoprotein receptor-related protein mediates apolipoprotein E-dependent neurite outgrowth in a central nervous system-derived neuronal cell line. Proc Natl Acad Sci USA 92:9480–9484.
- Holtzman DM, Fagan AM. 1998. Potential role of apoE in structural plasticity in the nervous system: implications for disorders of the central nervous system. Trends Cardiovasc Med 8:250–255.
- Honer WG, Dickson DW, Gleeson J, Davies P. 1992. Regional synaptic pathology in Alzheimer's disease. Neurobiol Aging 13:375–382.
- Horn D, Levy N, Ruppin E. 1996. Neuronal-based synaptic compensation: a computational study in Alzheimer's disease. Neural Comput 8:1227– 1243.
- Horner CH. 1993. Plasticity of the dendritic spine. Prog Neurobiol 41: 281-321.
- Horner P, Gage FH. 2000. Regenerating the damaged central nervous system. Nature 407:963–970.
- Horsburgh K, Kelly S, McCulloch J, Higgins GA. 1999. Increased neuronal damage in apolipoprotein E-deficient mice following global ischaemia. Neuroreport 17:837–841.
- Hsia AY, Masliah E, McConlogue L, Yu GQ, Tatsuno G, Hu K, Kholodenko D, Malenka RC, Nicoll RA, Mucke L. 1999. Plaque-independent disruption of neural circuits in Alzheimer's disease mouse models. Proc Natl Acad Sci USA 96:3228–3233.
- Huang DY, Weisgraber KH, Strittmatter WJ, Matthew WD. 1995. Interaction of apolipoprotein E with laminin increases neuronal adhesion and alters neurite morphology. Exp Neurol 136:251–257.

- Huang EJ, Reichardt LF. 2001. Neurotrophins: roles in neuronal development and function. Annu Rev Neurosci 24:677–736.
- Huang EP. 1999. Synaptic plasticity: regulated translation in dendrites. Curr Biol 9:168–170.
- Huang L, DeVries GJ, Bittman EL. 1998. Photoperiod regulates neuronal bromodeoxyuridine labeling in the brain of a seasonally breeding mammal. J Neurobiol 36:410–420.
- Huang Y, Liu XQ, Mahley RW. 1999. Differential effects of cytosolic apolipoprotein (APO) E3 and ApoE4 on neurite outgrowth. Soc Neurosci Abstr 16:4.
- Huber G, Bailly Y, Martin JR, Mariani J, Brugg B. 1997. Synaptic betaamyloid precursor proteins increase with learning capacity in rats. Neuroscience 80:313–320.
- Hyman BT, Gomez-Isla T, Rebeck GW, Briggs M, Chung H, West HL, Greenberg S, Mui S, Nichols S, Wallace R. 1996. Epidemiological, clinical, and neuropathological study of apolipoprotein E genotype in Alzheimer's disease. Ann N Y Acad Sci 802:1–5.
- Hyman BT, Kromer LJ, Van Hoesen GW. 1987. Reinnervation of the hippocampal perforant pathway zone in Alzheimer's disease. Ann Neurol 21:259–267.
- Hyman BT, van Hoesen GW, Damasio AR, Barnes CL. 1984. Alzheimer's disease: cell-specific pathology isolates the hippocampal formation. Science 225:1168–1170.
- Hynes RO. 1999. Cell adhesion: old and new questions. Trends Cell Biol 9:33–37.
- Ihara Y. 1988. Massive somatodendritic sprouting of cortical neurons in Alzheimer's disease. Brain Res 459:138-144.
- Inestrosa NC, Marzolo MP, Bonnefont AB. 1998. Cellular and molecular basis of estrogen's neuroprotection: potential relevance for Alzheimer's disease. Mol Neurobiol 17:73–86.
- Ishida A, Furukawa K, Keller JN, Mattson MP. 1997. Secreted form of beta-amyloid precursor protein shifts the frequency dependency for induction of LTD, and enhances LTP in hippocampal slices. Neuroreport 8:2133–2137.
- Janus C, Pearson J, McLaurin J, Mathews PM, Jiang Y, Schmidt SD, Chishti MA, Horne P, Heslin D, French J, Mount HT, Nixon RA, Mercken M, Bergeron C, Fraser PE, St. George-Hyslop P, Westaway D. 2000. A beta peptide immunization reduces behavioral impairment and plaques in a model of Alzheimer's disease. Nature 408:979–982.
- Jensen TG, Roses AD, Jorgensen AL. 1994. Apolipoprotein E uptake and degradation via chloroquine-sensitive pathway in cultivated monkey cells overexpressing low density lipoprotein receptor. Neurosci Lett 180:193– 196.
- Jette N, Cole MS, Fahnestock M. 1994. NGF mRNA is not decreased in frontal cortex from Alzheimer's disease patients. Brain Res Dev Brain Res 25:242–250.
- Ji SR, Wu Y, Sui SF. 2002. Cholesterol is an important factor affecting the membrane insertion of beta-amyloid peptide (A beta 1–40), which may potentially inhibit the fibril formation. J Biol Chem 277:6273–6279.
- Ji ZS, Pitas RE, Mahley RW. 1998. Differential cellular accumulation/ retention of apolipoprotein E mediated by cell surface heparan sulfate proteoglycans. Apolipoproteins E3 and E2 greater than E4. J Biol Chem 273:13452–13460.
- Jick H, Zornberg GL, Jick SS, Seshadri S, Drachman DA. 2000. Statins and the risk of dementia. Lancet 356:1627–1631.
- Jobst KA, Smith AD, Szatmari M, Esiri MM, Jaskowski A, Hindley N, McDonald B, Molyneux AJ. 1994. Rapidly progressing atrophy of medial temporal lobe in Alzheimer's disease. Lancet 343:829–830.
- Johannsson G, Oscarsson J, Rosen T, Wiklund O, Olsson G, Wilhelmsen L, Bengtsson BA. 1995. Effects of growth hormone therapy on serum lipoprotein levels in growth hormone deficient adults. Influence of gender and apo(a) and apoE phenotypes. Arterioscler Thromb Vasc Biol 15: 2142–2150.

- Jordan J, Galindo MF, Miller RJ, Reardon CA, Getz GS, LaDu MJ. 1998. Isoform-specific effect of apolipoprotein E on cell survival and betaamyloid-induced toxicity in rat hippocampal pyramidal neuronal cultures. J Neurosci 18:195–204.
- Jorgensen OS, Brooksbank BW, Balazs R. 1990. Neuronal plasticity and astrocytic reaction in Down syndrome and Alzheimer disease. J Neurol Sci 98:63–79.
- Jorgensen MB. 1993. The role of signal transduction in the delayed necrosis of the hippocampal CA1 pyramidal cells following transient ischemia. Acta Neurol Scand 143(Suppl):1–20.
- Jorgensen OS. 1995. Neural cell adhesion molecule (NCAM) as a quantitative marker in synaptic remodeling. Neurochem Res 20:533–547.
- Jorgensen OS, Hansen LI, Hoffman SW, Fulop Z, Stein DG. 1997. Synaptic remodeling and free radical formation after brain contusion injury in the rat. Exp Neurol 144:326–338.
- Joseph J, Shukitt-Hale B, Denisova NA, Martin A, Perry G, Smith MA. 2001. Copernicus revisited: amyloid beta in Alzheimer's disease. Neurobiol Aging 22:131–146.
- Kang H, Schuman EM. 1995. Long-lasting neurotrophin-induced enhancement of synaptic transmission in the adult hippocampus. Science 267:1658–1662.
- Kang DE, Saitoh T, Chen X, Xia Y, Masliah E, Hansen LA, Thomas RG, Thal LJ, Katzman R. 1997. Genetic association of the low-density lipoprotein receptor-related protein gene (LRP), an apolipoprotein E receptor, with late-onset Alzheimer's disease. Neurology 49:56–61.
- Kang DE, Soriano S, Frosch MP, Collins T, Naruse S, Sisodia SS, Leibowitz G, Levine F, Koo EH. 1999. Presenilin 1 facilitates the constitutive turnover of beta-catenin: differential activity of Alzheimer's diseaselinked PS1 mutants in the beta-catenin-signaling pathway. J Neurosci 19:4229–4237.
- Kang DE, Pietrzik CU, Baum L, Chevallier N, Merriam DE, Kounnas MZ, Wagner SL, Troncoso JC, Kawas CH, Katzman R, Koo EH. 2000. Modulation of amyloid beta-protein clearance and Alzheimer's disease susceptibility by the LDL receptor-related protein pathway. J Clin Invest 106:1159–1166.
- Kasa P, Papp H, Kovacs I, Forgon M, Penke B, Yamaguchi H. 2000. Human amyloid-beta1–42 applied in vivo inhibits the fast axonal transport of proteins in the sciatic nerve of rat. Neurosci Lett 278:117–9.
- Kawas C, Resnick S, Morrison A, Brookmeyer R, Corrada M, Zonderman A, Bacal C, Lingle D, Metter E. 1997. A prospective study of estrogen replacement therapy and the risk of developing Alzheimer's disease. Neurology 48:1517–1521.
- Keller JN, Lauderback CM, Butterfield DA, Kindy MS, Yu J, Markesbery WR. 2000. Amyloid beta-peptide effects on synaptosomes from apolipoprotein E-deficient mice. J Neurochem 74:1579–1586.
- Keller JN, Mattson MP. 1998. Roles of lipid peroxidation in modulation of cellular signaling pathways, cell dysfunction, and death in the nervous system. Rev Neurosci 9:105–116.
- Kelly MJ, Moss RL, Dudley CA. 1977. The effects of microelectrophoretically applied estrogen, cortisol and acetylcholine on medial preoptic-septal unit activity throughout the estrous cycle of the female rat. Exp Brain Res 30:53–64.
- Kelly MJ, Levin ER. 2001. Rapid actions of plasma membrane estrogen receptors. Trends Endocrinol Metab 12:152–156.
- Kempermann G, Kuhn HG, Gage FH. 1997. More hippocampal neurons in adult mice living in an enriched environment. Nature 386:493–495.
- Kempermann G, Brandon EP, Gage FH. 1998a. Environmental stimulation of 129/SvJ mice causes increased cell proliferation and neurogenesis in the adult dentate gyrus. Curr Biol 8:939–942.
- Kempermann G, Kuhn HG, Gage FH. 1998b. Experience-induced neurogenesis in the senescent dentate gyrus. J Neurosci 18:3206–3212.
- Kerr ME, Kraus M. 1998. Genetics and the central nervous system: apolipoprotein E and brain injury. AACN Clinical Issues 9:524–530.

- Kilgard MP, Merzenich MM. 1998. Cortical map reorganization enabled by nucleus basalis activity. Science 279:1714–1718.
- King DL, Arendash GW. 2002. Maintained synaptophysin immunoreactivity in Tg2576 transgenic mice during aging: correlations with cognitive impairment. Brain Res 926:58–68.
- Kittur SD, Endo H, Adler WH, Martin GR, Markesbery WR, Kleinman HK, Weeks BS. 1992. Alzheimer disease brain extract stimulates branching of laminin-mediated neuronal processes. Alzheimer Dis Assoc Disord 6:103–110.
- Kivatinitz SC, Pelsman MA, Alonso AC, Bagatolli L, Quiroga S. 1997. High-density lipoprotein aggregated by oxidation induces degeneration of neuronal cells. J Neurochem 69:2102–2114.
- Knipper M, da Penha Berzaghi M, Blochl A, Breer H, Thoenen H, Lindholm D. 1994. Positive feedback between acetylcholine and the neurotrophins nerve growth factor and brain-derived neurotrophic factor in the rat hippocampus. Eur J Neurosci 6:668–671.
- Koistinaho M, Ort M, Cimadevilla JM, Vondrous R, Cordell B, Koistinaho J, Bures J, Higgins LS. 2001. Specific spatial learning deficits become severe with age in β-amyloid precursor protein transgenic mice that harbor diffuse β-amyloid deposits but do not form plaques. Proc Natl Acad Sci USA 98:14675–14680.
- Kondo T, Shirasawa T, Itoyama Y, Mori H. 1996. Embryonic genes expressed in Alzheimer's disease brains. Neurosci Lett 209:157–160.
- Koo EH, Sisodia SS, Archer DR, Martin LJ, Weidemann A, Beyreuther K, Fischer P, Masters CL, Price DL. 1990. Precursor of amyloid protein in Alzheimer disease undergoes fast anterograde axonal transport. Proc Natl Acad Sci USA 87:1561–1565.
- Koo EH, Park L, Selkoe DJ. 1993. Amyloid  $\beta$ -protein as a substrate interacts with extracellular matrix to promote neurite outgrowth. Proc Natl Acad Sci USA 90:4748–4752.
- Kosik KS. 1991. The neuritic dystrophy of Alzheimer's disease: degeneration or regeneration? In: Hefti F, Brachet B, Christen WY, editors. Growth factors and Alzheimer's disease. New York: Springer-Verlag. p 234–240.
- Kosik K. 1992. Alzheimer's disease: a cell biological perspective. Science 256:780-783.
- Koudinov AR, Koudinova NV. 2001. Essential role for cholesterol in synaptic plasticity and neuronal degeneration. FASEB J 15:1858–1860.
- Krzywkowski P, Gagne J, Chabot C, Momoli F, Rochford J, Massicotte G, Poirier J. 1997. Pharmacology of the cholinergic and glutamatergic systems in behaviorally impaired ApoE knockout mice. Soc Neurosci Abstr 23:1895.
- Krzywkowski P, Ghribi O, Gagne J, Chabot C, Kar S, Rochford J, Massicotte G, Poirier J. 1999. Cholinergic systems and long-term potentiation in memory-impaired apolipoprotein E-deficient mice. Neuroscience 92:1273–1286.
- Kushwaha RS, Foster DM, Barrett PH, Carey KD, Bernard MG. 1991. Metabolic regulation of plasma apolipoprotein E by estrogen and progesterone in the baboon (*Papio sp*). Metabolism 40:93–100.
- LaDu MJ, Reardon C, Van Eldik L, Fagan AM, Bu G, Holtzman D, Getz GS. 2000. Lipoproteins in the central nervous system. Ann N Y Acad Sci 903:167–175.
- Lambert MP, Sabo S, Zhang C, Enam SA, Klein WL. 1995. Constitutive Alzheimer's-type tau epitopes in a neurotogenic rat CNS cell line. Neurobiol Aging 16:583–589.
- Lambert MP, Barlow AK, Chromy BA, Edwards C, Freed R, Liosatos M, Morgan TE, Rozovsky I, Trommer B, Viola KL, Wals P, Zhang C, Finch CE, Krafft GA, Klein WL. 1998. Diffusible, nonfibrillar ligands derived from Abeta1–42 are potent central nervous system neurotoxins. Proc Natl Acad Sci USA 95:6448–6453.
- Laming PR, Kimelberg H, Robinson S, Salm A, Hawrylak N, Muller C, Roots B, Ng K. 2000. Neuronal-glial interactions and behaviour. Neurosci Biobehav Rev 24:295–340.

- Lanahan A, Lyford G, Stevenson GS, Worley PF, Barnes CA. 1997. Selective alteration of long-term potentiation-induced transcriptional response in hippocampus of aged, memory-impaired rats. J Neurosci 17: 2876–2885.
- Lander HM, Hajjar DP, Hempstead BL, Mirza UA, Chait BT, Campbell S, Quilliam LA. 1997. A molecular redox switch on p21(ras). Structural basis for the nitric oxide-p21(ras) interaction. J Biol Chem 272:4323–4326.
- Lang T, Bruns D, Wenzel D, Riedel D, Holroyd P, Thiele C, Jahn R. 2001. SNAREs are concentrated in cholesterol-dependent clusters that define docking and fusion sites for exocytosis. EMBO J 20:2202–2213.
- Lanier LM, Gertler FB. 2000. From Abl to actin: Abl tyrosine kinase and associated proteins in growth cone motility. Curr Opin Neurobiol 10: 80–87.
- Lapchak PA, Jenden DJ, Hefti F. 1992. Pharmacological stimulation reveals recombinant human nerve growth factor-induced increases of in vivo hippocampal cholinergic function measured in rats with partial fimbrial transections. Neuroscience 50:847–856.
- Laskowitz DT, Horsburgh K, Roses AD. 1998. Apolipoprotein E and the CNS response to injury. J Cereb Blood Flow Metab 18:465–471.
- Laskowitz DT, Sheng H, Bart R, Joyner KA, Roses AD, Warner DS. 1997. Apolipoprotein E-deficient mice have increased susceptibility to focal ischemia. J Cereb Blood Flow Metab 17:753–758.
- Lassmann H, Weiler R, Fischer P, Bancher C, Jellinger K, Floor E, Danielczyk W, Seitelberger F, Winkler H. 1992. Synaptic pathology in Alzheimer's disease: immunological data for markers of synaptic and large dense-core vesicles. Neuroscience 46:1–8.
- Lassmann H, Fischer P, Jellinger K. 1993. Synaptic pathology of Alzheimer's disease. Ann N Y Acad Sci 695:59-64.
- Law A, Gauthier S, Quirion R. 2001. Say NO to Alzheimer's disease: the putative links between nitric oxide and dementia of the Alzheimer's type. Brain Res Brain Res Rev 35:73–96.
- Le Bars PL, Katz MM, Berman N, Itil TM, Freedman AM, Schatzberg AF. 1997. A placebo-controlled, double-blind, randomized trial of an extract of Ginkgo biloba for dementia. JAMA 278:1327–1332.
- Lee MK, Slunt HH, Martin LJ, Thinakaran G, Kim G, Gandy SE, Seeger M, Koo E, Price DL, Sisodia SS. 1996. Expression of presenilin 1 and 2 (PS1 and PS2) in human and murine tissues. J Neurosci 16:7513–7525.
- Lemke G. 2001. Glial control of neuronal development. Annu Rev Neurosci 24:87–105.
- Levi-Montalcini R. 1987. The nerve growth factor 35 years later. Science 237:1154–1162.
- Li D, Field PM, Raisman G. 1995. Failure of axon regeneration in postnatal rat entorhino-hippocampal slice coculture is due to maturation of the axon, not that of the pathway or target. Eur J Neurosci 7:1164–1171.
- Lin LH, Bock S, Carpenter K, Rose M, Norden JJ. 1992. Synthesis and transport of GAP-43 in entorhinal cortex neurons and perforant pathway during lesion-induced sprouting and reactive synaptogenesis. Brain Res Dev Brain Res 14:147–153.
- Lino M, Goto K, Kakegawa W, Okado H, Sudo M, Ishiuchi S, Miwa A, Takayasu Y, Saito I, Tsuzuki K, Ozawa S. 2001. Glia-synapse interaction through Ca<sup>2+</sup>-permeable AMPA receptors in Bergmann glia. Science 292:926–929.
- Liuzzi FJ, Lasek RJ. 1987. Astrocytes block axonal regeneration in mammals by activating the physiological stop pathway. Science 237:642–645.
- Locatelli S, Lutjohann D, Schmidt HH, Otto C, Beisiegel U, von Bergmann K. 2002. Reduction of plasma <sup>24</sup>S-hydroxycholesterol (cerebrosterol) levels using high-dosage simvastatin in patients with hypercholesterolemia: evidence that simvastatin affects cholesterol metabolism in the human brain. Arch Neurol 59:213–216.
- Lorenzo A, Diaz H, Carrer H, Caceres A. 1992. Amygdala neurons in vitro: neurite growth and effects of estradiol. J Neurosci Res 33:418–435.
- Lovestone S, Anderton B, Betts J, Dayanandan R, Gibb G, Ljungberg C, Pearce J. 2001. Apolipoprotein E gene and Alzheimer's disease: is tau the link? Biochem Soc Symp 111–120.

- Lovestone S, Reynolds CH. 1997. The phosphorylation of tau: a critical stage in neurodevelopment and neurodegenerative processes. Neuro-science 78:309–324.
- Lubec G, Nonaka M, Krapfenbauer K, Gratzer M, Cairns N, Fountoulakis M. 1999. Expression of the dihydropyrimidinase related protein 2 (DRP-2) in Down's syndrome and Alzheimer's disease brain is down-regulated at the mRNA and dysregulated at the protein level. J Neural Transm 57(Suppl):161–177.
- Lue LF, Kuo YM, Roher AE, Brachova L, Shen Y, Sue L, Beach T, Kurth JH, Rydel RE, Rogers J. 1999. Soluble amyloid beta peptide concentration as a predictor of synaptic change in Alzheimer's disease. Am J Pathol 155:853–862.
- Lukiw WJ, Bazan NG. 1998. Strong nuclear factor-κB-DNA binding parallels cyclooxygenase-2 gene transcription in aging and in sporadic Alzheimer's disease superior temporal lobe neocortex. J Neurosci Res 53:583–592.
- Luo L, Tully T, White K. 1992. Human amyloid precursor protein ameliorates behavioral deficit of flies deleted for Appl gene. Neuron 9:595– 605.
- Luscher C, Nicoll RA, Malenka RC, Muller D. 2000. Synaptic plasticity and dynamic modulation of the postsynaptic membrane. Nat Neurosci 3:545–550.
- Luth HJ, Hedlich A, Winkmann E. 1995. Postnatal development of NADPH-diaphorase positive nerve cells in the visual cortex of the rat. Ann Anat 177:573–577.
- Luth HJ, Arendt T. 1998. Nitric oxide and Alzheimer's disease. J Hirnforsch 39:245–251.
- Luth HJ, Holzer M, Gertz HJ, Arendt T. 2000. Aberrant expression of nNOS in pyramidal neurons in Alzheimer's disease is highly co-localized with p21ras and p16INK4a. Brain Res 852:45–55.
- Lyckman AW, Confaloni AM, Thinakaran G, Sisodia SS, Moya KL. 1998. Post-translational processing and turnover kinetics of presynaptically targeted amyloid precursor superfamily proteins in the central nervous system. J Biol Chem 273:11100–11106.
- Maccioni RB, Cambiazo V. 1995. Role of microtubule associated proteins in the control of microtubule assembly. Physiol Rev 23:1–27.
- Maccioni RB, Otth C, Concha II, Munoz JP. 2001. The protein kinase Cdk5. Structural aspects, roles in neurogenesis and involvement in Alzheimer's pathology. Eur J Biochem 268:1518–1527.
- MacGowan SH, Wilcock GK, Scott M. 1998. Effect of gender and apolipoprotein E genotype on response to anticholinesterase therapy in Alzheimer's disease. Int J Geriatr Psychiatry 13:625–630.
- Mack TG, Koester MP, Pollerberg GE. 2000. The microtubule-associated protein MAP1B is involved in local stabilization of turning growth cones. Mol Cell Neurosci 15:51–65.
- Mahley RW. 1988. Apolipoprotein E: cholesterol transport protein with expanding role in cell biology. Science 240:622–630.
- Mahley R, Huang Y. 1999. Apolipoprotein E: from atherosclerosis to Alzheimer's disease and beyond. Curr Opin Lipidol 10:207–217.
- Mahley RW, Rall SC Jr. 2000. Apolipoprotein E: far more than a lipid transport protein. Annu Rev Genomics Hum Genet 1:507–537.
- Maimone D, Dominici R, Grimaldi LM. 2001. Pharmacogenomics of neurodegenerative diseases. Eur J Pharmacol 413:11–29.
- Majocha RE, Jungalwala FB, Rodenrys A, Marotta CA. 1989. Monoclonal antibody to embryonic CNS antigen A2B5 provides evidence for the involvement of membrane components at sites of Alzheimer degeneration and detects sulfatides as well as gangliosides. J Neurochem 53:953–961.
- Maletic-Savatic M, Malinow R, Svoboda K. 1999. Rapid dendritic morphogenesis in CA1 hippocampal dendrites induced by synaptic activity. Science 283:1923–1927.
- Mandelkow EM, Biernat J, Drewes G, Gustke N, Trinczek B, Mandelkow E. 1995. Tau domains, phosphorylation, and interactions with microtubules. Neurobiol Aging 16:355–362; discussion 362–633.

- Marder K, Sano M. 2000. Estrogen to treat Alzheimer's disease: too little, too late? So what's a woman to do? Neurology 54:2035–2037.
- Markesbery W, Carney JM. 1999. Oxidative alterations in Alzheimer's disease. Brain Pathol 9:133–146.
- Marques MA, Harmony JA, Crutcher KA. 1996. A thrombin cleavage fragment of apolipoprotein E and related synthetic peptides is receptor-mediated. Neuroreport 7:2529–2532.
- Martens JR, Navarro-Polanco R, Coppock EA, Nishiyama A, Parshley L, Grobaski TD, Tamkun MM. 2000. Differential targeting of Shaker-like potassium channels to lipid rafts. J Biol Chem 275:7443–7446.
- Martin KC, Barad M, Kandel ER. 2000. Local protein synthesis and its role in synapse-specific plasticity. Curr Opin Neurobiol 10:587–592.
- Martin LJ, Pardo CA, Cork LC, Price DL. 1994. Synaptic pathology and glial responses to neuronal injury precede the formation of senile plaques and amyloid deposits in the aging cerebral cortex. Am J Pathol 145:1358–1381.
- Martin SJ, Grimwood PD, Morris RG. 2000. Synaptic plasticity and memory: an evaluation of the hypothesis. Annu Rev Neurosci 23:649– 711.
- Masliah E, Terry RD, DeTeresa RM, Hansen LA. 1989. Immunohistochemical quantification of the synapse-related protein synaptophysin in Alzheimer disease. Neurosci Lett 103:234–239.
- Masliah E, Iimoto DS, Saitoh T, Hansen LA, Terry RD. 1990. Increased immunoreactivity of brain spectrin in Alzheimer disease: a marker for synapse loss? Brain Res 531:36–44.
- Masliah E, Fagan AM, Terry RD, DeTeresa R, Mallory M, Gage FH. 1991a. Reactive synaptogenesis assessed by synaptophysin immunoreactivity is associated with GAP-43 in the dentate gyrus of the adult rat. Exp Neurol 113:131–142.
- Masliah E, Hansen L, Mallory M, Albright T, Terry RD. 1991b. Abnormal brain spectrin immunoreactivity in sprouting neurons in Alzheimer disease. Neurosci Lett 129:1–5.
- Masliah E, Mallory M, Hansen L, Alford M, Albright T, DeTeresa R, Terry R, Baudier J, Saitoh T. 1991c. Patterns of aberrant sprouting in Alzheimer's disease. Neuron 6:729–739.
- Masliah E, Terry RD, Alford M, DeTeresa R, Hansen LA. 1991d. Cortical and subcortical patterns of synaptophysin like immunoreactivity in Alzheimer's disease. Am J Pathol 138:235–246.
- Masliah E, Mallory M, Hansen L, Alford M, DeTeresa R, Terry R, Baudier J, Saitoh T. 1992. Localization of amyloid precursor protein in GAP43immunoreactive aberrant sprouting neurites in Alzheimer's disease. Brain Res 574:312–316.
- Masliah E, Mallory M, DeTeresa R, Alford M, Hansen L. 1993a. Differing patterns of aberrant neuronal sprouting in Alzheimer's disease with and without Lewy bodies. Brain Res 617:258–266.
- Masliah E, Miller A, Terry RD. 1993b. The synaptic organization of the neocortex in Alzheimer's disease. Med Hypotheses 41:334–340.
- Masliah E, Mallory M, Hansen L, DeTeresa R, Alford M, Terry R. 1994. Synaptic and neuritic alterations during the progression of Alzheimer's disease. Neurosci Lett 174:67–72.
- Masliah E, Mallory M, Alford M, Deteresa R, Saitoh T. 1995a. PDGF is associated with neuronal and glial alterations of Alzheimer's disease. Neurobiol Aging 16:549–556.
- Masliah E, Mallory M, Alford M, Ge N, Mucke L. 1995b. Abnormal synaptic regeneration in hAPP695 transgenic and ApoE knockout mice. In: Iqbal K, Mortimer J, Winblad B, Wisniewski HM, editors. Research advances in Alzheimer's disease and related disorders. Chichester: Wiley. p 405–414.
- Masliah E, Mallory M, Ge N, Alford M, Veinbergs I, Roses AD. 1995c. Neurodegeneration in the central nervous system of ApoE-deficient mice. Exp Neurol 136:107–122.
- Masliah E, Mallory M, Alford M, Veinbergs I, Roses AD. 1995d. Apolipoprotein E role in maintaining the integrity of the aging central nervous

system. In: Christian Y, Roses AD, editors. Apolipoprotein E and Alzheimer's disease. Berlin Heidelberg: Springer-Verlag.

- Masliah E, Mallory M, Veinbergs I, Miller A, Samuel W. 1996. Alterations in apolipoprotein E expression during aging and neurodegeneration. Prog Neurobiol 50:493–503.
- Masliah E, Samuel W, Veinbergs I, Mallory M, Mante M, Saitoh T. 1997. Neurodegeneration and cognitive impairment in apoE-deficient mice is ameliorated by infusion of recombinant apoE. Brain Res 751:307–314.
- Masliah E, Armasolo F, Veinbergs I, Mallory M, Samuel W. 1999. Cerebrolysin ameliorates performance deficits, and neuronal damage in apolipoprotein E-deficient mice. Pharmacol Biochem Behav 62:239–245.
- Mason RP, Shoemaker WJ, Shajenko L, Chambers TE, Herbette LG. 1992. Evidence for changes in the Alzheimer's disease brain cortical membrane structure mediated by cholesterol. Neurobiol Aging 13:413– 419.
- Matsumoto A. 1991. Synaptogenic action of sex steroids in developing and adult neuroendocrine brain. Psychoneuroendocrinology 16:25–40.
- Matthies H Jr, Schulz S, Hollt V, Krug M. 1997. Inhibition by compactin demonstrates a requirement of isoprenoid metabolism for long-term potentiation in rat hippocampal slices. Neuroscience 79:341–346.
- Mattila KM, Axelman K, Rinne JO, Blomberg M, Lehtimaki T, Laippala P, Roytta M, Viitanen M, Wahlund LO, Winblad B, Lannfelt L. 2000. Interaction between estrogen receptor 1 and the e4 allele of apolipoprotein E increases the risk of familial Alzheimer's disease in women. Neurosci Lett 282:45–48.
- Mattson MP, Cheng B, Davis D, Bryant K, Lieberburg I, Rydel RE. 1992. Beta-amyloid peptides destabilize calcium homeostasis and render human cortical neurons vulnerable to excitotoxicity. J Neurosci 12:376–389.
- Mattson MP. 1994. Secreted forms of β-amyloid precursor protein modulate dendrite outgrowth and calcium responses to glutamate in cultured embryonic hippocampal neurons. J Neurobiol 25:439–450.
- Mattson MP. 1997. Cellular actions of  $\beta$ -amyloid precursor protein and its soluble and fibrillogenic derivatives. Physiol Rev 77:1081–1132.
- Mattson MP, Camandola S. 2001. NF-κB in neuronal plasticity and neurodegenerative disorders. J Clin Invest 107:247–254.
- Mattson MP, Gary DS, Chan SL, Duan W. 2001. Perturbed endoplasmic reticulum function, synaptic apoptosis and the pathogenesis of Alzheimer's disease. Biochem Soc Symp 67:151–162.
- Matus A. 1999. Postsynaptic actin and neuronal plasticity. Curr Opin Neurobiol 9:561–565.
- Mauch DH, Nagler K, Schumacher S, Gorits C, Muller E-C, Otto A, Pfrieger FW. 2001. CNS synaptogenesis promoted by glial-derived cholesterol. Science 294:1354–1357.
- McAllister AK, Katz LC, Lo DC. 1996. Neurotrophin regulation of cortical dendritic growth requires activity. Neuron 17:1057–1064.
- McAllister AK, Katz LC, Lo DC. 1999. Neurotrophins and synaptic plasticity. Annu Rev Neurosci 22:295–318.
- McEwen BS. 1996. Gonadal and adrenal steroids regulate neurochemical and structural plasticity of the hippocampus via cellular mechanisms involving NMDA receptors. Cell Mol Neurobiol 16:103–116.
- McEwen BS, Alves SE, Bulloch K, Weiland NG. 1997. Ovarian steroids and the brain: implications for cognition and aging. Neurology 48(Suppl): 8–15.
- McEwen BS. 2001. Invited review: Estrogens effects on the brain: multiple sites and molecular mechanisms. J Appl Physiol 91:2785–2801.
- McGowan KA, Marinkovich MP. 2000. Laminins and human disease. Microsc Res Tech 51:262–279.
- McKee AC, Kosik KS, Kowall NW. 1991. Neuritic pathology and dementia in Alzheimer's disease. Ann Neurol 30:156–165.
- McKee AC, Kowall NW, Kosik KS. 1989. Microtubular reorganization and dendritic growth response in Alzheimer's disease. Ann Neurol 26: 652–659.

- McKeon RJ, Schreiber RC, Rudge JS, Silver J. 1991. Reduction of neurite outgrowth in a model of glial scarring following CNS injury is correlated with the expression of inhibitory molecules on reactive astrocytes. J Neurosci 11:3398–3411.
- McLean CA, Cherny RA, Fraser FW, Fuller SJ, Smith MJ, Beyreuther K, Bush AI, Masters CL. 1999. Soluble pool of A $\beta$  amyloid as a determinant of severity of neurodegeneration in Alzheimer's disease. Ann Neurol 46:860–866.
- McRae A, Dahlstrom A, Ling EA. 1997. Microglial in neurodegenerative disorders: emphasis on Alzheimer's disease. Gerontology 43:95–108.
- McWilliams JR, Lynch G. 1984. Synaptic density and axonal sprouting in rat hippocampus: stability in adulthood and decline in late adulthood. Brain Res 294:152–156.
- Meberg PJ, Kinney WR, Valcourt EG, Routtenberg A. 1996. Gene expression of the transcription factor NF-κB in hippocampus: regulation by synaptic activity. Brain Res Dev Brain Res 38:179–190.
- Mesulam MM. 1996. The systems-level organization of cholinergic innervation in the human cerebral cortex and its alterations in Alzheimer's disease. Prog Brain Res 109:285–297.
- Mesulam MM. 1999. Neuroplasticity failure in Alzheimer's disease: bridging the gap between plaques and tangles. Neuron 24:521–529.
- Mesulam MM. 2000. A plasticity-based theory of the pathogenesis of Alzheimer's disease. Ann N Y Acad Sci 924:42–52.
- Meyer-Franke A, Kaplan MR, Pfrieger FW, Barres BA. 1995. Characterization of the signaling interactions that promote the survival and growth of developing retinal ganglion cells in culture. Neuron 15:805–819.
- Meyer MR. 1998. APOE genotype predicts when-not whether-one is predisposed to develop Alzheimer disease. Nat Genet 19:321–322.
- Mikkonen M, Soininen H, Tapiola T, Alafuzoff I, Miettinen R. 1999. Hippocampal plasticity in Alzheimer's disease: changes in highly polysialylated NCAM immunoreactivity in the hippocampal formation. Eur J Neurosci 11:1754–1764.
- Milward EA, Papadopoulos R, Fuller SJ, Moir RD, Small D, Beyreuther K, Masters CL. 1992. The amyloid protein precursor of Alzheimer's disease is a mediator of the effects of nerve growth factor on neurite outgrowth. Neuron 9:129–137.
- Miranda P, Williams CL, Einstein G. 1999. Granule cells are sexually dimorphic in their response to estradiol. J Neurosci 19:3316–3325.
- Miwa C, Ueki A. 1996. Effects of entorhinal cortex lesion on learning behavior and on hippocampus in the rat. Psychiatry Clin Neurosci 50: 223–230.
- Mobley WC, Rutkowski JL, Tennekoon GI, Buchanan K, Johnston MV. 1985. Choline acetyltransferase activity in striatum of neonatal rats increased by nerve growth factor. Science 229:284–287.
- Moechars D, Dewachter I, Lorent K, Reverse D, Baekelandt V, Naidu A, Tesseur I, Spittaels K, Haute CV, Checler F, Godaux E, Cordell B, Van Leuven F. 1999. Early phenotypic changes in transgenic mice that overexpress different mutants of amyloid precursor protein in brain. J Biol Chem 274:6483–6492.
- Monning U, Sandbrink R, Weidemann A, Banati RB, Masters CL, Beyreuther K. 1995. Extracellular matrix influences the biogenesis of amyloid precursor protein in microglial cells. J Biol Chem 270:7104– 7110.
- Montine T, Montine KS, Olson SJ, Graham DG, Roberts JL II, Morrow JD, Linton MF, Fazio S, Swift LL. 1999. Increased cerebral cortical lipid peroxidation and abnormal phospholipids in aged homozygous apoE-deficient C57BL/6J mice. Exp Neurol 158:234–241.
- Morgan D, Diamond DM, Gottschall PE, Ugen KE, Dickey C, Hardy J, Duff K, Jantzen P, DiCarlo G, Wilcock D, Connor K, Hatcher J, Hope C, Gordon M, Arendash GW. 2000. A beta peptide vaccination prevents memory loss in an animal model of Alzheimer's disease. Nature 408:982– 985.
- Mori N. 1993. Toward understanding of the molecular basis of loss of neuronal plasticity in ageing. Age Ageing 22(Suppl):5–18.

- Morrison BM, Hof PR, Morrison JH. 1998. Determinants of neuronal vulnerability in neurodegenerative diseases. Ann Neurol 44(Suppl):32–44.
- Morrison JH, Hof PR. 1997. Life and death of neurons in the aging brain. Science 278:412–419.
- Morse JK, Scheff SW, DeKosky ST. 1986. Gonadal steroids influence axon sprouting in the hippocampal dentate gyrus: a sexually dimorphic response. Exp Neurol 94:649–658.
- Morse JK, DeKosky ST, Scheff SW. 1992. Neurotrophic effects of steroids on lesion-induced growth in the hippocampus. II. Hormone replacement. Exp Neurol 118:47–52.
- Morton RA, Kuenzi FM, Fitzjohn SM, Rosahl TW, Smith D, Zheng H, Shearman M, Collingridge GL, Seabrook GR. 2002. Impairment in hippocampal long-term potentiation in mice under-expressing the Alzheimer's disease related gene presenilin-1. Neurosci Lett 319:37–40.
- Moya KL, Benowitz LI, Schneider GE, Allinquant B. 1994. The amyloid precursor protein is developmentally regulated and correlated with synaptogenesis. Dev Biol 161:597–603.
- Mrak RE, Griffin ST, Graham DI. 1997. Aging-associated changes in human brain. J Neuropathol Exp Neurol 56:1269–1275.
- Mucke L, Masliah E, Johnson WB, Ruppe MD, Alford M, Rockenstein EM, Forss-Petter S, Pietropaolo M, Mallory M, Abraham CR. 1994. Synaptotrophic effects of human amyloid β protein precursors in the cortex of transgenic mice. Brain Res 666:151–167.
- Mucke L, Masliah E, Yu GQ, Mallory M, Rockenstein EM, Tatsuno G, Hu K, Kholodenko D, Johnson-Wood K, McConlogue L. 2000. Highlevel neuronal expression of A $\beta$  1–42 in wild-type human amyloid protein precursor transgenic mice: synaptotoxicity without plaque formation. J Neurosci 20:4050–4058.
- Muesing RA, Miller VT, LaRosa JC, Stoy DB, Phillips EA. 1992. Effects of unopposed conjugated equine estrogen on lipoprotein composition and apolipoprotein-E distribution. J Clin Endocrinol Metab 75:1250–1254.
- Mufson EJ, Kordower JH. 1992. Cortical neurons express nerve growth factor receptors in advanced age and Alzheimer disease. Proc Natl Acad Sci USA 89:569–573.
- Mulder M, Ravid R, Swaab DF, de Kloet ER, Haasdijk ED, Julk J, van der Boom JJ, Havekes LM. 1998. Reduced levels of cholesterol, phospholipids, and fatty acids in cerebrospinal fluid of Alzheimer disease patients are not related to apolipoprotein E4. Alzheimer Dis Assoc Disord 12: 198–203.
- Muller D, Buchs PA, Stoppini L. 1993. Time course of synaptic development in hippocampal organotypic cultures. Brain Res Dev Brain Res 71:93–100.
- Muller U, Cristina N, Li ZW, Wolfer DP, Lipp HP, Rulicke T, Brandner S, Aguzzi A, Weissmann C. 1994. Behavioral and anatomical deficits in mice homozygous for a modified  $\beta$ -amyloid precursor protein gene. Cell 79:755–765.
- Muller D, Wang C, Skibo G, Toni N, Cremer H, Calaora V, Rougon G, Kiss JZ. 1996. PSA-NCAM is required for activity-induced synaptic plasticity. Neuron 17:413–422.
- Muller HW, Gebicke-Harter PJ, Hangen DH, Shooter EM. 1997. A specific 37,000-Dalton protein that accumulates in regenerating but not in nongenerating mammalian nerves. Science 228:499–501.
- Muller WE, Kirsch C, Eckert GP. 2001. Membrane-disordering effects of beta-amyloid peptides. Biochem Soc Transm 29:617–623.
- Mulnard RA, Cotman CW, Kawas C, van Dyck CH, Sano M, Doody R, Koss E, Pfeiffer E, Jin S, Gamst A, Grundman M, Thomas R, Thal LJ. 2000. Estrogen replacement therapy for treatment of mild to moderate Alzheimer disease: a randomized controlled trial. Alzheimer's Disease Cooperative Study. JAMA 283:1007–1015.
- Murer MG, Yan Q, Raisman-Vozari R. 2001. Brain-derived neurotrophic factor in the control human brain, and in Alzheimer's disease and Parkinson's disease. Prog Neurobiol 63:71–124.

- Murtomaki S, Risteli J, Risteli L, Koivisto UM, Johansson S, Liesi P. 1992. Laminin and its neurite outgrowth-promoting domain in the brain in Alzheimer's disease and Down's syndrome patients. J Neurosci Res 32: 261–273.
- Myoken Y, Kan M, Sato GH, McKeehan WL, Sato JD. 1990. Bifunctional effects of transforming growth factor-beta (TGF- $\beta$ ) on endothelial cell growth correlate with phenotypes of TGF- $\beta$  binding sites. Exp Cell Res 191:299–304.
- Nabekura J, Oomura Y, Minami T, Mizuno Y, Fukuda A. 1986. Mechanism of the rapid effect of  $17\beta$ -estradiol on medial amygdala neurons. Science 233:226–228.
- Nagy Z, Esiri MM, Cato AM, Smith AD. 1997. Cell cycle markers in the hippocampus in Alzheimer's disease. Acta Neuropathol (Berl) 94:6–15.
- Nalbantoglu J, Tirado-Santiago G, Lahsaini A, Poirier J, Goncalves O, Verge G, Momoli F, Welner SA, Massicotte G, Julien JP, Shapiro ML. 1997. Impaired learning and LTP in mice expressing the carboxy terminus of the Alzheimer amyloid precursor protein. Nature 387:500–505.
- Narindrasorasak S, Altman RA, Gonzalez-DeWhitt P, Greenberg BD, Kisilevsky R. 1995. An interaction between basement membrane and Alzheimer amyloid precursor proteins suggests a role in the pathogenesis of Alzheimer's disease. Lab Invest 72:272–282.
- Naruse S, Thinakaran G, Luo JJ, Kusiak JW, Tomita T, Iwatsubo T, Qian X, Ginty DD, Price DL, Borchelt DR, Wong PC, Sisodia SS. 1998. Effects of PS1 deficiency on membrane protein trafficking in neurons. Neuron 21:1213–1221.
- Nathan BP, Bellosta S, Sanan DA, Weisgraber KH, Mahley RW, Pitas RE. 1994. Differential effects of apolipoproteins E3 and E4 on neuronal growth in vitro. Science 264:850–852.
- Nathan BP, Chang K, Bellosta S, Brisch E, Ge N, Mahley RW, Pitas RE. 1995. The inhibitory effect of apolipoprotein E4 on neurite outgrowth is associated with microtubule depolymerization. J Biol Chem 270:19791–19799.
- Nathan BP, Jiang Y, Wong GK, Shen F, Brewer G, Struble RG. 2002. Apolipoprotein E4 inhibits, and apolipoprotein E3 promotes neurite sprouting in cultured adult mouse cortical neurons through a low-density lipoprotein receptor-related protein. Brain Res 928:96–105.
- Neely MD, Sidell KR, Graham DG, Montine TJ. 1999. The lipid peroxidation product 4-hydroxynonenal inhibits neurite outgrowth, disrupts neuronal microtubules and modifies cellular tubulin. J Neurochem 72: 2323–2333.
- Neill D. 1995. Alzheimer's disease: maladaptive synaptoplasticity hypothesis. Neurodegeneration 4:217–232.
- Neumann S, Woolf CJ. 1999. Regeneration of dorsal column fibers into and beyond the lesion site following adult spinal cord injury. Neuron 23:83–91.
- Neve RL, Finch EA, Bird ED, Benowitz LI. 1988. Growth-associated protein GAP-43 is expressed selectively in associative regions of the adult human brain. Proc Natl Acad Sci USA 85:3638–3642.
- Neve RL, McPhie DL, Chen Y. 2000. Alzheimer's disease: a dysfunction of the amyloid precursor protein. Brain Res 886:54–66.
- Neve RL. 2001. A $\beta$  may be a planet, but APP is central. Neurobiol Aging 22:151–154; discussion 161–163.
- Neve RL, McPhie DL, Chen Y. 2001. Alzheimer's disease: dysfunction of a signaling pathway mediated by the amyloid precursor protein? Biochem Soc Symp 67:37–50.
- Nichols NR, Day JR, Laping NJ, Johnson SA, Finch CE. 1993. GFAP mRNA increases with age in rat and human brain. Neurobiol Aging 14:421–429.
- Nishimura M, Yu G, Levesque G, Zhang DM, Ruel L, Chen F, Milman P, Holmes E, Liang Y, Kawarai T, Jo E, Supala A, Rogaeva E, Xu DM, Janus C, Levesque L, Bi Q, Duthie M, Rozmahel R, Mattila K, Lannfelt L, Westaway D, Mount HT, Woodgett J, St. George-Hyslop P, et al. 1999. Presenilin mutations associated with Alzheimer disease cause de-

fective intracellular trafficking of  $\beta$ -catenin, a component of the presenilin protein complex. Nat Med 5:164–169.

- Nitsch RM, Rebeck GW, Deng M, Richardson UI, Tennis M, Schenk DB, Vigo-Pelfrey C, Lieberburg I, Wurtman RJ, Hyman BT, Growdon JH. 1995. Cerebrospinal fluid levels of amyloid β-protein in Alzheimer's disease: inverse correlation with severity of dementia and effect of apolipoprotein E genotype. Ann Neurol 37:512–518.
- Nitsch RM, Slack BE, Wurtman RJ, Growdon JH. 1992. Release of Alzheimer amyloid precursor derivatives stimulated by activation of muscarinic acetylcholine receptors. Science 258:304–307.
- Noll E, Medina M, Hartley D, Zhou J, Perrimon N, Kosik KS. 2000. Presenilin affects arm/beta-catenin localization and function in Drosophila. Dev Biol 227:450–464.
- Norenberg MD. 1994. Astrocyte responses to CNS injury. J Neuropathol Exp Neurol 53:213–220.
- Nothias F, Fischer I, Murray M, Mirman S, Vincent JD. 1996. Expression of a phosphorylated isoform of MAP1B is maintained in adult central nervous system areas that retain capacity for structural plasticity. J Comp Neurol 368:317–334.
- Nourooz-Zadeh J, Liu EHC, Yhlen B, Anggard EE, Halliwell B. 1999. F4-Isoprostanes as specific marker of docosahexaenoic acid peroxidation in Alzheimer's disease. J Neurochem 72:734–740.
- O'Neill C, Cowburn RF, Bonkale WL, Ohm TG, Fastbom J, Carmody M, Kelliher M. 2001. Dysfunctional intracellular calcium homoeostasis: a central cause of neurodegeneration in Alzheimer's disease. Biochem Soc Symp177–194.
- O'Neill LA, Kaltschmidt C. 1997. NF-κB: a crucial transcription factor for glial and neuronal cell function. Trends Neurosci 20:252–258.
- Ohkubo N, Mitsuda N, Tamatani M, Yamaguchi A, Lee YD, Ogihara T, Vitek MP, Tohyama M. 2001. Apolipoprotein E4 stimulates cAMP response element-binding protein transcriptional activity through the extracellular signal-regulated kinase pathway. J Biol Chem 276:3046– 3053.
- Ohm TG, Hamker U, Cedazo-Minguez A, Rockl W, Scharnagl H, Marz W, Cowburn R, Muller W, Meske V. 2001. Apolipoprotein E and β A4-amyloid: signals and effects. Biochem Soc Symp 67:121–129.
- Ohsawa I, Takamura C, Morimoto T, Ishiguro M, Kohsaka S. 1999. Amino-terminal region of secreted form of amyloid precursor protein stimulates proliferation of neural stem cells. Eur J Neurosci 11:1907–1913.
- Ohta M, Kitamoto T, Iwaki T, Ohgami T, Fukui M, Tateishi J. 1993. Immunohistochemical distribution of amyloid precursor protein during normal rat development. Brain Res Dev Brain Res 75:151–161.
- Ono K, Takii T, Onozaki K, Ikawa M, Okabe M, Sawada M. 1999. Migration of exogenous immature hematopoietic cells into adult mouse brain parenchyma under GFP-expressing bone marrow chimera. Biochem Biophys Res Commun 262:610–614.
- Paganini-Hill A. 1996. Estrogen replacement therapy and Alzheimer's disease. Br J Obstet Gynaecol 103:80-86.
- Paglini G, Pigino G, Kunda P, Morfini G, Maccioni R, Quiroga S, Ferreira A, Caceres A. 1998. Evidence for the participation of the neuron-specific CDK5 activator P35 during laminin-enhanced axonal growth. J Neurosci 18:9858–9869.
- Palmert MR, Usiak M, Mayeux R, Raskind M, Tourtellotte WW, Younkin SG. 1990. Soluble derivatives of the beta amyloid protein precursor in cerebrospinal fluid: alterations in normal aging and in Alzheimer's disease. Neurology 40:1028–1034.
- Parent A, Linden DJ, Sisodia SS, Borchelt DR. 1999. Synaptic transmission and hippocampal long-term potentiation in transgenic mice expressing FAD-linked presenilin 1. Neurobiol Dis 6:56–62.
- Pasinetti GM. 2001. Use of cDNA microarray in the search for molecular markers involved in the onset of Alzheimer's disease dementia. J Neurosci Res 65:471–476.

- Paula-Barbosa MM, Cardoso RM, Guimaraes ML, Cruz C. 1980. Dendritic degeneration and regrowth in the cerebral cortex of patients with Alzheimer's disease. J Neurol Sci 45:129–134.
- Pauwels PJ, van Assouw HP, De Ryck M, Leysen JE, Dom R, Van Gool D. 1993. Towards an improved survival of rat brain neurons in culture by cerebrospinal fluid of patients with senile dementia of Alzheimer's type. Brain Res 610:8–15.
- Perez-Otano I, McMillan MK, Chen J, Bing G, Hong JS, Pennypacker KR. 1996. Induction of NF-κB-like transcription factors in brain areas susceptible to kainate toxicity. Glia 4:306–315.
- Perron JC, Bixby JL. 1999. Distinct neurite outgrowth signaling pathways converge on ERK activation. Mol Cell Neurosci 13:362–378.
- Pfenninger KH, de la Houssaye BA, Helmke SM, Quiroga S. 1991. Growth-regulated proteins and neuronal plasticity. A commentary. Mol Neurobiol 5:143–151.
- Pfrieger FW, Barres BA. 1997. Synaptic efficacy enhanced by glial cells in vitro. Science 277:1684–1687.
- Phillips LL, Belardo ET. 1994. Increase of c-fos and ras oncoproteins in the denervated neuropil of the rat dentate gyrus. Neuroscience 58:503–514.
- Phinney AL, Deller T, Stalder M, Calhoun ME, Frotscher M, Sommer B, Staufenbiel M, Jucker M. 1999. Cerebral amyloid induces aberrant axonal sprouting and ectopic terminal formation in amyloid precursor protein transgenic mice. J Neurosci 19:8552–8559.
- Piedrahita JA, Zhang SH, Hagaman JR, Oliver PM, Maeda N. 1992. Generation of mice carrying a mutant apolipoprotein E gene inactivated by gene targeting in embryonic stem cells. Proc Natl Acad Sci USA 89:4471–4475.
- Pierce JP, van Leyen K, McCarthy JB. 2000. Translocation machinery for synthesis of integral membrane and secretory proteins in dendritic spines. Nat Neurosci 3:311–333.
- Pigino G, Paglini G, Ulloa L, Avila J, Caceres A. 1997. Analysis of the expression, distribution and function of cyclin dependent kinase 5 (cdk5) in developing cerebellar macroneurons. J Cell Sci 110:257–270.
- Pike CJ. 2001. Testosterone attenuates beta-amyloid toxicity in cultured hippocampal neurons. Brain Res 919:160–165.
- Pitas RE. 1996. Microtubule formation and neurite extension are blocked by apolipoprotein E4. Semin Cell Biol 7:725–731.
- Pitas RE, Ji ZS, Weisgraber KH, Mahley RW. 1998. Role of apolipoprotein E in modulating neurite outgrowth: potential effect of intracellular apolipoprotein E. Biochem Soc Transm 26:257–262.
- Pitchumoni SS, Doraiswamy PM. 1999. Current status of antioxidant therapy for Alzheimer's disease. J Am Geriatr Soc 46:1566–1572.
- Poirier J, Hess M, May PC, Finch CE. 1991a. Astrocytic apolipoprotein E mRNA and GFAP mRNA in hippocampus after entorhinal cortex lesioning. Brain Res Dev Brain Res 11:97–106.
- Poirier J, Hess M, May PC, Finch CE. 1991b. Cloning of hippocampal poly(A) RNA sequences that increase after entorhinal cortex lesion in adult rat. Brain Res Dev Brain Res 9:191–195.
- Poirier J, Baccichet DD, Gauthier S. 1993a. Cholesterol synthesis and lipoprotein reuptake during synaptic remodeling in hippocampus in adult rats. Neuroscience 55:81–90.
- Poirier J, Davignon J, Bouthillier D, Kogan S, Bertrand P, Gauthier S. 1993b. Apolipoprotein E polymorphism and Alzheimer's disease. Lancet 342:697–699.
- Poirier J. 1994. Apolipoprotein E in animal models of CNS injury and in Alzheimer's disease. Trends Neurosci 17:525–530.
- Poirier J, Delisle M-C, Quirion R, Aubert I, Farlow M, Lahiri D, Hui S, Bertrand P, Nalbantoglu J, Gilfix BM. 1995. Apolipoprotein E4 allele as a predictor of cholinergic deficits and treatment outcome in Alzheimer disease. Proc Natl Acad Sci USA 92:12260–12264.
- Poirier J, Minnich A, Davignon J. 1995. Apolipoprotein E, synaptic plasticity and Alzheimer's disease. Ann Med 27:663–670.
- Poirier J. 1999. Apolipoprotein E: a pharmacogenetic target for the treatment of Alzheimer's disease. Mol Diagn 4:335–341.

- Popko B, Goodrum JF, Bouldin TW, Zhang SH, Maeda N. 1993. Nerve regeneration occurs in the absence of apolipoprotein E in mice. J Neurochem 60:1155–1158.
- Posse de Chaves EI, Rusinol AE, Vance DE, Campenot RB, Vance JE. 1997. Role of lipoproteins in the delivery of lipids to axons during axonal regeneration. J Biol Chem 272:30766–30773.
- Pratico D, Rokach J, Tangirala RK. 1999. Brains of aged apolipoprotein E-deficient mice have increased levels of F2-Isoprostanes, in vivo markers of lipid peroxidation. J Neurochem 73:736–741.
- Pratico D, Tangirala RK, Rader DJ, Rokach J, Fitzgerald GA. 1998. Vitamin E suppresses isoprostane generation in vivo and reduces atherosclerosis in ApoE-deficient mice. Nat Med 4:1189–1192.
- Probst A, Basler V, Bron B, Ulrich J. 1983. Neuritic plaques in senile dementia of Alzheimer type: a Golgi analysis in the hippocampal region. Brain Res 268:249–254.
- Puttfarcken PS, Manelli AM, Falduto MT, Getz GS, LaDu M-J. 1997. Effect of apolipoprotein E on neurite outgrowth and β-amyloid-induced toxicity in developing rat primary hippocampal cultures. J Neurochem 68:760–769.
- Qiu WQ, Ferreira A, Miller C, Koo EH, Selkoe DJ. 1995. Cell-surface beta-amyloid precursor protein stimulates neurite outgrowth of hippocampal neurons in an isoform-dependent manner. J Neurosci 15:2157– 2167.
- Rabchevsky AG. 2002. Influences of activated microglia/brain macrophages on spinal cord injury and regeneration. In: Streit WJ, editor. Microglia in the regenerating and degenerating central nervous system. New York: Springer-Verlag. p 209–226.
- Raber J, Wong D, Buttini M, Orth M, Bellosta S, Pitas RE, Mahley RW, Mucke L. 1998. Isoform-specific effects of human apolipoprotein E on brain function revealed in apoE knockout mice: increased susceptibility of females. Proc Natl Acad Sci USA 95:10914–10919.
- Raber J, Wong D, Yu G-Q, Buttini M, Mahley RW, Pitas RE, Mucke L. 2000. Apolipoprotein E and cognitive performance. Nature 404:352– 253.
- Raber J, Bongers G, LeFevour A, Buttini M, Mucke L. 2002. Androgens protect against apolipoprotein E4-induced cognitive deficits. J Neurosci 22:5204–5209.
- Ramassamy C, Averill D, Beffert U, Theroux L, Lussier-Cacan S, Cohn JS. 2000. Oxidative insults are associated with apolipoprotein E genotype in Alzheimer's disease brain. Neurobiol Dis 7:23–37.
- Ramer MS, Priestley JV, McMahon SB. 2000. Functional regeneration of sensory axons into the adult spinal cord. Nature 403:312–316.
- Ramirez G, Alvarez A, Garcia-Abreu J, Gomes FC, Moura Neto V, Maccioni RB. 1999. Regulatory roles of microtubule-associated proteins in neuronal morphogenesis. Involvement of the extracellular matrix. Braz J Med Biol Res 32:611–618.
- Ramon y Cajal S. 1928. Degeneration and regeneration of the nervous system. May R, translator. New York: Hafner.
- Rapoport SI. 1990. Integrated phylogeny of the primate brain with special reference to humans and their disease. Brain Res Brain Res Rev 15:267–294.
- Rapoport SI. 1999. In vivo PET imaging and postmortem studies suggest potentially reversible and irreversible stages of brain metabolic failure in Alzheimer's disease. Eur Arch Psychiatry Clin Neurosci 249(Suppl):46–55.
- Rapoport SI. 2001. In vivo fatty acid incorporation into brain phospholipids in relation to plasma availability, signal transduction and membrane remodeling. J Mol Neurosci 16:243–261; discussion 279–284.
- Rattray M. 2001. Is there nicotinic modulation of nerve growth factor? Implications for cholinergic therapies in Alzheimer's disease. Biol Psychiatry 49:185–193.
- Rebeck GW, Reiter JS, Strickland DK, Hyman BT. 1993. Apolipoprotein E in sporadic Alzheimer's disease: allelic variation and receptor interactions. Neuron 11:575–580.

- Refolo LM, Malester B, LaFrancois J, Bryant-Thomas T, Wang R, Tint GS, Sambamurti K, Duff K, Pappolla MA. 2000. Hypercholesterolemia accelerates the Alzheimer's amyloid pathology in a transgenic mouse model. Neurobiol Dis 7:321–331.
- Reier PJ, Stensaas LJ, Guth L. 1983. The astrocytes scar as an impediment of regeneration in central nervous system. In: Kan CG, Bunge RP, Reier PJ, editors. Spinal cord reconstruction. New York: Raven Press. p 163–195.
- Rice DS, Nusinowitz S, Azimi AM, Martinez A, Soriano E, Curran T. 2001. The reelin pathway modulates the structure and function of retinal synaptic circuitry. Neuron 31:929–941.
- Richard F, Helbecque N, Neuman E, Guez D, Levy R, Amouyel P. 1997. ApoE genotyping and response to drug treatment in Alzheimer's disease. Lancet 349:539.
- Roch JM, Masliah E, Roch-Levecq AC, Sundsmo MP, Otero DA, Veinbergs I, Saitoh T. 1994. Increase of synaptic density and memory retention by a peptide representing the trophic domain of the amyloid beta/A4 protein precursor. Proc Natl Acad Sci USA 91:7450–7454.
- Rockwood K, Kirkland S, Hogan DB, MacKnight C, Merry H, Verreault R, Wolfson C, McDowell I. 2002. Use of lipid-lowering agents, indication bias, and the risk of dementia in community-dwelling elderly people. Arch Neurol 59:223–227.
- Ronn LC, Hartz BP, Bock E. 1998. The neural cell adhesion molecule (NCAM) in development and plasticity of the nervous system. Exp Gerontol 33:853–864.
- Ronn LC, Olsen M, Ostergaard S, Kiselyov V, Berezin V, Mortensen MT, Lerche MH, Jensen PH, Soroka V, Saffell JL, Doherty P, Poulsen FM, Bock E, Holm A, Saffells JL. 1999. Identification of a neurotogenic ligand of the neural cell adhesion molecule using a combinatorial library of synthetic peptides. Nat Biotechnol 17:1000–1005.
- Roses AD, Saunders AM, Alberts MA, Strittmatter WJ, Schmechel D, Gorder E, Pericak-Vance MA. 1995. Apolipoprotein E E4 allele and risk of dementia. JAMA 273:374–375.
- Roses AD, Einstein G, Gilbert J, Goedert M, Han SH, Huang D, Hulette C, Masliah E, Pericak-Vance MA, Saunders AM, Schmechel DE, Strittmatter WJ, Weisgraber KH, Xi PT. 1996. Morphological, biochemical, and genetic support for an apolipoprotein E effect on microtubular metabolism. Ann N Y Acad Sci 777:147–157.
- Roses AD, Saunders AM. 1997. Lipid metabolism in the brain: a truly undeveloped science. In: de la Torre JC, Hachinski V, editors. Cerebrovascular pathology in Alzheimer's disease. New York: New York Academy of Sciences. p 208.
- Rosier JP, Riopelle RJ, Bisby MA. 1996. Axotomy-induced apoptotic cell death of neonatal rat facial motoneurons: time course analysis and relation to NADPH-diaphorase activity. Exp Neurol 138:33–44.
- Rother M, Erkinjuntti T, Roessner M, Kittner B, Marcusson J, Karlsson I. 1998. Propentofylline in the treatment of Alzheimer's disease and vascular dementia: a review of phase III trials. Dement Geriatr Cogn Disord 9(Suppl):36–43.
- Rozovsky I, Wei M, Stone DJ, Zanjani H, Anderson CP, Morgan TE, Finch CE. 2002. Estradiol (E2) enhances neurite outgrowth by repressing glial fibrillary acidic protein expression and reorganizing laminin. Endocrinology 143:636–646.
- Rudling M, Angelin B. 1993. Stimulation of rat hepatic low density lipoprotein receptors by glucagon: evidence of a novel regulatory mechanism in vivo. J Clin Invest 91:2796–2805.
- Runz H, Rietdorf J, Tomic I, de Bernard M, Beyreuther K, Pepperkok R, Hartmann T. 2002. Inhibition of intracellular cholesterol transport alters presenilin localization and amyloid precursor protein processing in neuronal cells. J Neurosci 22:1679–1689.
- Ruther E, Ritter R, Apecechea M, Freytag S, Gmeinbaer R, Windisch M. 2000. Sustained improvements in patients with dementia of Alzheimer's type (DAT) 6 months after termination of Cerebrolysin therapy. J Neural Transm 107:815–829.
- Saada A, Dunaevsky-Hutt A, Aamar A, Reichert F, Rotshenker S. 1995.

Fibroblasts that reside in mouse and frog injured peripheral nerves produce apolipoproteins. J Neurochem 64:1996–2003.

- Saitoh T, Horsburgh K, Masliah E. 1993. Hyperactivation of signal transduction systems in Alzheimer's disease. Ann N Y Acad Sci 695:34–41.
- Salib E, Hillier V. Head injury and the risk of Alzheimer's disease: a case control study. 1997. Int J Geriatr Psychiatry 12:363–368.
- Sanderson M, Wang J, Davis DR, Lane MJ, Cornman CB, Fadden MK. 2002. Co-morbidity associated with dementia. Am J Alzheimers Dis Other Demen 17:73–78.
- Sano M, Ernesto C, Thomas RG, Klauber MR, Schafer K, Grundman M, Woodbury P, Growdon J, Cotman CW, Pfeiffer E, Schneider LS, Thal LJ. 1997a. A controlled trial of selegiline, alpha-tocopherol, or both as treatment for Alzheimer's disease. The Alzheimer's Disease Cooperative Study. N Engl J Med 336:1216–1222.
- Sano M, Ernesto C, Thomas RG, Klauber MR, Schafer K, Grundman M, Woodbury P, Growdon J, Cotman CW, Pfeiffer E, Schneider LS, Thal LJ. 1997b. A controlled trial of selegiline, α-tocopherol, or both as treatment for Alzheimer's disease. The Alzheimer's Disease Cooperative Study. N Eng J Med 336:1216–1222.
- Sato N, Kamino K, Tateishi K, Satoh T, Nishiwaki Y, Yoshiiwa A, Miki T, Ogihara T. 1997. Elevated amyloid  $\beta$  protein (1–40) level induces CREB phosphorylation at serine-133 via p44/42 MAP kinase (Erk1/2)-dependent pathway in rat pheochromocytoma PC12 cells. Biochem Biophys Res Commun 232:637–642.
- Saunders AM, Trowers MK, Shimkets RA, Blakemore S, Crowther DJ, Mansfield TA, Wallace DM, Strittmatter WJ, Roses AD. 2000. The role of apolipoprotein E in Alzheimer's disease: pharmacogenomic target selection. Biochim Biophys Acta 1502:85–94.
- Sayre LM, Zelasko DA, Harris PLR, Perry G. 1997. 4-Hydroxynonenalderived advanced lipid peroxidation end products are increased in Alzheimer's disease. J Neurochem 68:2092–2097.
- Schacher S, Wu F. 2002. Synapse formation in the absence of cell bodies requires protein synthesis. J Neurosci 22:1831–1839.
- Scheff SW, Benardo LS, Cotman CW. 1980. Decline in reactive fiber growth in the dentate gyrus of aged rats compared to young adult rats following entorhinal cortex removal. Brain Res 199:21–38.
- Scheff SW, DeKosky ST, Price DA. 1990. Quantitative assessment of cortical synaptic density in Alzheimer's disease. Neurobiol Aging 11:29–37.
- Scheibel AB. 1982. Age-related changes in the human forebrain. Neurosci Res Program Bull 20:577–583.
- Scheibel AB, Tomiyasu U. 1978. Dendritic sprouting in Alzheimer's presenile dementia. Exp Neurol 60:1–8.
- Schneider LS, Farlow M. 1997. Combined tacrine and estrogen replacement therapy in patients with Alzheimer's disease. Ann N Y Acad Sci 826:317–322.
- Schneider I, Reverse D, Dewachter I, Ris L, Caluwaerts N, Kuiperi C, Gilis M, Geerts H, Kretzschmar H, Godaux E, Moechars D, Van Leuven F, Herms J. 2001. Mutant presenilins disturb neuronal calcium homeostasis in the brain of transgenic mice, decreasing the threshold for excitotoxicity and facilitating long-term potentiation. J Biol Chem 276:11539– 11544.
- Schubert D. 1991. The possible role of adhesion in synaptic modification. Trends Neurosci 14:127–130.
- Schuman EM. 1999. mRNA trafficking and local protein synthesis at the synapse. Neuron 23:645–648.
- Scott RB, Collins JM, Hunt PA. 1994. Alzheimer's disease and Down syndrome: leukocyte membrane fluidity alterations. Mech Ageing Dev 75:1–10.
- Scott SA. 1993. Dendritic atrophy and remodeling of amygdaloid neurons in Alzheimer's disease. Dementia 4:264–272.
- Scott SA, Mufson EJ, Weingartner JA, Skau KA, Crutcher KA. 1995. Nerve growth factor in Alzheimer's disease: increased levels throughout the brain coupled with declines in nucleus basalis. J Neurosci 15:6213– 6221.

- Seaberg RM, van der Kooy D. 2002. Adult rodent neurogenic regions: the ventricular subependyma contains neural stem cells, but the dentate gyrus contains restricted progenitors. J Neurosci 22:1784–1793.
- Segal M, Murphy DD. 1998. CREB activation mediates plasticity in cultured hippocampal neurons. Neural Plast 6:1–7.
- Seki T, Arai Y. 1993. Distribution and possible roles of the highly polysialylated neural cell adhesion molecule (NCAM-H) in the developing and adult central nervous system. Neurosci Res 17:265–290.
- Seki T, Rutishauser U. 1998. Removal of polysialic acid-neural cell adhesion molecule induces aberrant mossy fiber innervation and ectopic synaptogenesis in the hippocampus. J Neurosci 18:3757–3766.
- Sekiguchi M, Abe H, Nagato Y, Tanaka O, Guo H, Nowakowski RS. 1996. The abnormal distribution of mossy fiber bundles and morphological abnormalities in hippocampal formation of dreher-J (dr-J/dr-J) mouse. Brain Res Dev Brain Res 92:31–38.
- Sestan N, Artavanis-Tsakonas S, Rakic P. 1999. Contact-dependent inhibition of cortical neurite growth mediated by notch signaling. Science 286:741–746.
- Shen J, Bronson RT, Chen DF, Xia W, Selkoe DJ, Tonegawa S. 1997. Skeletal and CNS defects in Presenilin-1-deficient mice. Cell 89:629– 639.
- Sherwin BB, Tulandi T. 1996. "Add-back" estrogen reverses cognitive deficits induced by a gonadotropin-releasing hormone agonist in women with leiomyomata uteri. J Clin Endocrinol Metab 81:2545–2549.
- Shirazi SK, Wood JG. 1993. The protein tyrosine kinase, fyn, in Alzheimer's disease pathology. Neuroreport 4:435–437.
- Shughrue PJ, Dorsa DM. 1993. Estrogen modulates the growth-associated protein GAP-43 (Neuromodulin) mRNA in the rat preoptic area and basal hypothalamus. Neuroendocrinology 57:439–447.
- Siegel GJ, Chauhan NB. 2000. Neurotrophic factors in Alzheimer's and Parkinson's disease brain. Brain Res Brain Res Rev 33:199–227.
- Sievers J, Bamberger C, Debus OM, Lucius R. 1995. Regeneration in the optic nerve of adult rats: influences of cultured astrocytes and optic nerve grafts of different ontogenetic stages. J Neurocytol 24:783–793.
- Silva AJ, Kogan JH, Frankland PW, Kida S. 1998. CREB and memory. Annu Rev Neurosci 21:127–148.
- Simchowicz T. 1911. Histologic studien uber die senile demenz. In: Nissl F, Alzheimer A, editors. Histologische und histopathologische arbeiten uber die grosshirnrinde mit besonderer berucksichtigung der pathologischen anatomie der geistekrankheiten. Jena: Fischer. p 267–444.
- Sisodia SS, Koo EH, Hoffman PN, Perry G, Price DL. 1993. Identification and transport of full-length amyloid precursor proteins in rat peripheral nervous system. J Neurosci 13:3136–3142.
- Sjoberg A, Oscarsson J, Eden S, Olofsson SO. 1994. Continuous but not intermittent administration of growth hormone to hypophysectomized rats increases apolipoprotein-E secretion from cultured hepatocytes. Endocrinology 134:790–798.
- Sloane PD, Zimmerman S, Suchindran C, Reed P, Wang L, Boustani M, Sudha S. 2002. The public health impact of Alzheimer's disease, 2000– 2050: potential implication of treatment advances. Annu Rev Public Health 23:213–231.
- Slomianka L, Geneser FA. 1997. Postnatal development of zinc-containing cells and neuropil in the hippocampal region of the mouse. Hippocampus 7:321–340.
- Smith DS, Skene JH. 1997. A transcription-dependent switch controls competence of adult neurons for distinct modes of axon growth. J Neurosci 17:646–658.
- Smith JD, Miyata M, Poulin SE, Neveux LM, Craig WY. 1998. The relationship between apolipoprotein E and serum oxidation-related variables is apolipoprotein E phenotype dependent. Int J Clin Lab Res 286:116–121.
- Smith MZ, Nagy Z, Esiri MM. 1999. Cell cycle-related protein expression in vascular dementia and Alzheimer's disease. Neurosci Lett 271:45–48. Snow DM, Lemmon V, Carrino DA, Caplan AI, Silver J. 1990. Sulfated

proteoglycans in astroglial barriers inhibit neurite outgrowth in vitro. Exp Neurol 109:111–130.

- Snowdon DA, Greiner LH, Mortimer JA, Riley KP, Greiner PA, Markesbery WR. 1997. Brain infarction and the clinical expression of Alzheimer disease. The Nun Study. JAMA 277:813–817.
- Sofroniew MV, Howe CL, Mobley WC. 2001. Nerve growth factor signaling, neuroprotection, and neural repair. Annu Rev Neurosci 24: 1217–1281.
- Srivastava RA, Krul ES, Lin RC, Schonfeld G. 1997. Regulation of lipoprotein metabolism by estrogen in inbred strains of mice occurs primarily by posttranscriptional mechanisms. Mol Cell Biochem 173:161–168.
- Steward O, Schuman EM. 2001. Protein synthesis at synaptic sites on dendrites. Annu Rev Neurosci 24:299-355.
- Stokoe D, Macdonald SG, Cadwallader K, Symons M, Hancock JF. 1994. Activation of Raf as a result of recruitment to the plasma membrane. Science 264:1463–1467.
- Stone DJ, Rozovsky I, Morgan TE, Anderson CP, Hajian H, Finch CE. 1997. Astrocytes and microglia respond to estrogen with increased apoE mRNA in vivo and in vitro. Exp Neurol 143:313–318.
- Stone DJ, Rozovsky I, Morgan TE, Anderson CP, Finch CE. 1998. Increased synaptic sprouting in response to estrogen via an apolipoprotein E-dependent mechanism: implications for Alzheimer's disease. J Neurosci 18:3180–3185.
- Stone DJ, Rozovsky I, Morgan TE, Anderson CP, Lopez LM, Shick J, Finch CE. 2000. Effects of age on gene expression during estrogeninduced synaptic sprouting in the female rat. Exp Neurol 165:46–57.
- Stoppini L, Buchs P-A, Muller D. 1993. Lesion-induced neurite sprouting and synapse formation in hippocampal organotypic cultures. Neuroscience 57:985–994.
- Stringer JL, Agarwal KS, Dure LS. 1997. Is cell death necessary for hippocampal mossy fiber sprouting? Epilepsy Res 27:67–76.
- Styren SD, Bowser R, Dekosky ST. 1999. Expression of fetal ALZ-50 reactive clone 1 (FAC1) in dentate gyrus following entorhinal cortex lesion. J Comp Neurol 386:555–561.
- Su JH, Cummings BJ, Cotman CW. 1993. Identification and distribution of axonal dystrophic neurites in Alzheimer's disease. Brain Res 625:228– 237.
- Su JH, Deng GM, Cotman CW. 1997. Neuronal DNA damage precedes tangle formation and is associated with up-regulation of nitrotyrosine in Alzheimer's disease brain. Brain Res 774:193–199.
- Sun Y, Wu S, Bu G, Onifade MK, Patel SN, LaDu MJ, Fagan AM, Holtzman DM. 1998. Glial fibrillary acidic protein-apolipoprotein E transgenic mice: astrocyte-specific expression and differing biological effects of astrocyte-secreted apoE3 and apoE4 lipoproteins. J Neurosci 18:3261–3272.
- Suter DM, Forscher P. 1998. An emerging link between cytoskeletal dynamics and cell adhesion molecules in growth cone guidance. Curr Opin Neurobiol 8:106–116.
- Sutula T, Cascino G, Cavazos J, Parada I, Ramires L. 1989. Mossy fiber synaptic reorganization in the epileptic human temporal lobe. Ann Neurol 26:321–330.
- Svennerholm L, Gottfries CG. 1994. Membrane lipids, selectively diminished in Alzheimer brains, suggest synapse loss as a primary event in early-onset form (type I) and demyelination in late-onset form (type II). J Neurochem 62:1039–1047.
- Sze CI, Troncoso JC, Kawas C, Mouton P, Price DL, Martin LJ. 1997. Loss of the presynaptic vesicle protein synaptophysin in hippocampus correlates with cognitive decline in Alzheimer disease. J Neuropathol Exp Neurol 56:933–944.
- Szklarczyk A, Lapinska J, Rylski M, McKay RD, Kaczmarek L. 2002. Matrix metalloproteinase-9 undergoes expression and activation during dendritic remodeling in adult hippocampus. J Neurosci 22:920– 930.

- Takeuchi R, Murase K, Furukawa Y, Furukawa S, Hayashi K. 1990. Stimulation of nerve growth factor synthesis/secretion by 1,4benzoquinone and its derivatives in cultured mouse astroglial cells. FEBS Lett 261:63–66.
- Tam S-P, Archer TK, Deeley RG. 1986. Biphasic effects of estrogen on apolipoprotein synthesis in human hepatoma cells: mechanisms of antagonism by testosterone. Proc Natl Acad Sci USA 83:3111–3115.
- Tanapat P, Hastings NB, Reeves AJ, Gould E. 1999. Estrogen stimulates a transient increase in the number of new neurons in the dentate gyrus of the adult female rat. J Neurosci 19:5792–5801.
- Tang M-X, Jacobs D, Stern Y, Marder K, Schofield P, Gurland B, Andrews H, Mayeux R. 1996. Effect of oestrogen during menopause on risk and age at onset of Alzheimer's disease. Lancet 348:429–432.
- Tanzi RE. 1999. A genetic dichotomy model for the inheritance of Alzheimer's disease and common age-related disorders. J Clin Invest 104:1175–1179.
- Temple K, Malouf AT. 2000. Regeneration of hippocampal mossy fibers in cultured hippocampal slices from P21 rat. Soc Neurosci Abstr 30:323.
- Terry RD, Masliah E, Salmon DP, Butters N, DeTeresa R, Hill R, Hansen LA, Katzman R. 1991. Physical basis of cognitive alterations in Alzheimer's disease: synapse loss is the major correlate of cognitive impairment. Ann Neurol 30:572–580.
- Terry RD, Masliah E, Hansen LA. 1994. Structural basis of the cognitive alterations in Alzheimer's disease. In: Terry RD, Katzman R, Bick K, editors. Alzheimer disease. New York: Raven Press. p 179–196.
- Tesseur I, Van Dorpe J, Spittaels K, Van den Haute C, Moechars D, van Leuven F. 2000. Expression of human apolipoprotein E4 in neurons causes hyperphosphorylation of protein tau in the brains of transgenic mice. Am J Pathol 156:951–964.
- Teter B, Harris-White M, Frautschy SA, Cole GM. 1999a. Role of Apolipoprotein E and estrogen in mossy fiber sprouting in hippocampal slice cultures. Neuroscience 91:1009–1016.
- Teter B, Xu P-T, Gilbert JR, Roses AD, Galasko D, Cole GM. 1999b. Human apolipoprotein E isoform-specific differences in neuronal sprouting in organotypic hippocampal culture. J Neurochem 73:2613–2616.
- Teter B. 2000. Apolipoprotein E isotype-specific effects in neurodegeneration. Alzheimer's Reports 3:199–212.
- Teter B, Raber J, Nathan B, Crutcher KA. 2002. The presence of apoE4, not the absence of apoE3, contributes to AD pathology. J Alzheimer's Dis 4:155–163.
- Teter B, Xu P-T, Gilbert JR, Roses AD, Galasko D, Cole GM. 2002. Defective neuronal sprouting supported by human Apolipoprotein E4 represents a gain-of-deleterious function. J Neurosci Res 68:331–336.
- Teyler TJ, Cavus I, Coussens C, DiScenna P, Grover L, Lee YP, Little Z. 1994. Multideterminant role of calcium in hippocampal synaptic plasticity. Hippocampus 4:623–634.
- Tezuka T, Umemori HAT, Nakanishi SYT. 2001. PSD-95 promotes fyn-mediated tyrosine phosphorylation of the *N*-methyl-D-aspartate receptor subunit NR2A. Proc Natl Acad Sci USA 96:435–440.
- Therond P, Bonnefont-Rousselot D, Laureaux C, Vasson MP, Motta C, Legrand A, Delattre J. 1999. Copper oxidation of in vitro dioleolylphosphatidylcholine-enriched high-density lipoproteins: physicochemical features and cholesterol effluxing capacity. Arch Biochem Biophys 362:139–147.
- Thiele C, Hannah MJ, Fahrenholz F, Huttner WB. 2000. Cholesterol binds to synaptophysin and is required for biogenesis of synaptic vesicles. Nat Cell Biol 2:42–49.
- Thinakaran G, Kitt CA, Roskams AJ, Slunt HH, Masliah E, von Koch C, Ginsberg SD, Ronnett GV, Reed RR, Price DL, et al. 1995. Distribution of an APP homolog, APLP2, in the mouse olfactory system: a potential role for APLP2 in axogenesis. J Neurosci 15:6314–6326.
- Tiedge H, Bloom FE, Richter D. 2001. Molecular kinesis in cellular function and plasticity. Proc Natl Acad Sci USA 98:6997–6998.
- Tokuda T, Ikeda S, Yanagisawa N, Ihara Y, Glenner GG. 1991. Re-

examination of ex-boxers' brains using immunohistochemistry with antibodies to amyloid  $\beta$ -protein and  $\tau$  protein. Acta Neuropathol (Berl) 82:280–285.

- Toni N, Buchs PA, Nikonenko I, Bron CR, Muller D. 1999. LTP promotes formation of multiple spine synapses between a single axon terminal and a dendrite. Nature 402:421–425.
- Toran-Allerand CD, Singh M, Setalo G Jr. 1999. Novel mechanisms of estrogen action in the brain: new players in an old story. Front Neuroen-docrinol 20:97–121.
- Trojanowski JQ, Schmidt ML, Shin R-W, Bramblett GT, Rao D, Lee VMY. 1993. Altered tau and neurofilament proteins in neurodegenerative diseases: diagnostic implications for Alzheimer's disease and Lewy body dementias. Brain Pathol 3:45–54.
- Tuttle R, O'Leary DD. 1998. Neurotrophins rapidly modulate growth cone response to the axon guidance molecule, collapsin-1. Mol Cell Neurosci 11:1–8.
- Twist SJ, Taylor GA, Weddell A, Weightman DR, Edwardson JA, Morris CM. 2000. Brain oestradiol and testosterone levels in Alzheimer's disease. Neurosci Lett 286:1–4.
- Uchida Y, Ihara Y, Tomonaga M. 1988. Alzheimer's disease brain extract stimulates the survival of cerebral cortical neurons from neonatal rats. Biochem Biophys Res Commun 150:1263–1267.
- Uchida Y, Tomonaga M. 1989. Neurotrophic action of Alzheimer's disease brain extract is due to the loss of inhibitory factors for survival and neurite formation of cerebral cortical neurons. Brain Res 481:190–193.
- Ullian EM, Sapperstein SK, Christopherson KS, Barres BA. 2001. Control of synapse number by glia. Science 291:657–660.
- Van der Zee CE, Lourenssen S, Stanisz J, Diamond J. 1995a. NGF deprivation of adult rat brain results in cholinergic hypofunction and selective impairments in spatial learning. Eur J Neurosci 7:160–168.
- Van Der Zee CEEM, Rashid K, Le K, Moore K-A, Stanisz J, Diamond J, Racine RJ, Fahnestock M. 1995b. Intraventricular administration of antibodies to nerve growth factor retards kindling and blocks mossy fiber sprouting in adult rats. J Neurosci 15:5316–5323.
- van Gool WA, Schenk DB, Bolhuis PA. 1994. Concentrations of amyloidbeta protein in cerebrospinal fluid increase with age in patients free from neurodegenerative disease. Neurosci Lett 172:122–124.
- Vance JE, Campenot RB, Vance DE. 2000. The synthesis and transport of lipids for axonal growth and nerve regeneration. Biochim Biophys Acta 1486:84–96.
- Vedder H, Anthes N, Stumm G, Wurz C, Behl C, Krieg JC. 1999. Estrogen hormones reduce lipid peroxidation in cells and tissues of the central nervous system. J Neurochem 72:2531–2538.
- Veinbergs I, Juang M, Young SJ, Van Uden E, Samuel W, Grooves PM, Masliah E. 1997. Synaptic, behavioral, electrophysiological alterations in apolipoprotein E deficient mice. Soc Neurosci Abstr 23:1646.
- Veinbergs I, Mallory M, Mante M, Rockenstein E, Gilbert JR, Masliah E. 1999. Differential neurotrophic effects of apolipoprotein E in aged transgenic mice. Neurosci Lett 265:218–222.
- Veinbergs I, Masliah E. 1999. Synaptic alterations in apolipoprotein E knockout mice. Neuroscience 91:401–403.
- Veinbergs I, Mallory M, Sagara Y, Masliah E. 2000. Vitamin E supplementation prevents spatial learning deficits and dendritic alterations in aged apolipoprotein E-deficient mice. Eur J Neurosci 12:4541–4546.
- Veinbergs I, Van Uden E, Mallory M, Alford M, McGiffert C, DeTeresa R, Orlando R, Masliah E. 2001. Role of apolipoprotein E receptors in regulating the differential in vivo neurotrophic effects of apolipoprotein E. Exp Neurol 170:15–26.
- Verdi JM, Groves AK, Farinas I, Jones K, Marchionni MA, Reichardt LF, Anderson DJ. 1996. A reciprocal cell–cell interaction mediated by NT-3 and neuregulins controls the early survival and development of sympathetic neuroblasts. Neuron 16:515–527.
- Vernadakis A. 1996. Glia-neuron intercommunications and synaptic plasticity. Prog Neurobiol 49:185–214.

- Viereck C, Tucker RP, Matus A. 1989. The adult rat olfactory system expresses microtubule-associated proteins found in the developing brain. J Neurosci 9:3547–3557.
- von Bergen M, Friedhoff P, Biernat J, Heberle J, Mandelkow EM, Mandelkow E. 2000. Assembly of tau protein into Alzheimer paired helical filaments depends on a local sequence motif ((306)VQIVYK(311)) forming  $\beta$  structure. Proc Natl Acad Sci USA 97:5129–5134.
- Wahrle S, Das P, Nyborg AC, McLendon C, Shoji M, Kawarabayashi T, Younkin LH, Younkin SG, Golde TE. 2002. Cholesterol-dependent  $\gamma$ -secretase activity in buoyant cholesterol-rich membrane microdomains. Neurobiol Dis 9:11–23.
- Wallace W, Ahlers ST, Gotlib J, Bragin V, Sugar J, Gluck R, Shea PA, Davis KL, Haroutunian V. 1993. Amyloid precursor protein in the cerebral cortex is rapidly and persistently induced by loss of subcortical innervation. Proc Natl Acad Sci USA 90:8712–8716.
- Walsh DM, Klyubin I, Fadeeva JV, Cullen WK, Anwyl R, Wolfe MS, Rowan MJ, Selkoe DJ. 2002. Naturally secreted oligomers of amyloid beta protein potently inhibit hippocampal long-term potentiation in vivo. Nature 416:535–539.
- Wang HW, Pasternak JF, Kuo H, Ristic H, Lambert MP, Chromy B, Viola KL, Klein WL, Stine WB, Krafft GA, Trommer BL. 2002. Soluble oligomers of β-amyloid (1–42) inhibit long-term potentiation but not long-term depression in rat dentate gyrus. Brain Res 924:133–140.
- Wang PN, Liao SQ, Liu RS, Liu CY, Chao HT, Lu SR, Yu HY, Wang SJ, Liu HC. 2000. Effects of estrogen on cognition, mood, and cerebral blood flow in AD: a controlled study. Neurology 54:2061–2066.
- Weggen S, Eriksen JL, Das P, Sagi SA, Wang R, Pietrzik CU, Findlay KA, Smith TE, Murphy MP, Bulter T, Kang DE, Marquez-Sterling N, Golde TE, Koo EH. 2001. A subset of NSAIDs lower amyloidogenic Aβ42 independently of cyclooxygenase activity. Nature 414:212–216.
- Wehrle R, Caroni P, Sotelo C, Dusart I. 2001. Role of GAP-43 in mediating the responsiveness of cerebellar and precerebellar neurons to axotomy. Eur J Neurosci 13:857–870.
- Weidner N, Blesch A, Grill RJ, Tuszynski MH. 1999. Nerve growth factor-hypersecreting Schwann cell grafts augment and guide spinal cord axonal growth and remyelinate central nervous system axons in a phenotypically appropriate manner that correlates with expression of L1. J Comp Neurol 413:495–506.
- Weiler R, Lassmann H, Fischer P, Jellinger K, Winkler H. 1990. A high ratio of chromogranin A to synaptin/synaptophysin is a common feature of brains in Alzheimer and Pick disease. FEBS Lett 263:337–339.
- Weisgraber KH. 2001. Apolipoprotein E: structure-function relationships. Adv Protein Chem 45:249–302.
- Wells DG, Richter JD, Fallon JR. 2000. Molecular mechanisms for activity-regulated protein synthesis in the synapto-dendritic compartment. Curr Opin Neurobiol 10:132–137.
- Wheal HV, Chen Y, Mitchell J, Schachner M, Maerz W, Wieland H, Van Rossum D, Kirsch J. 1998. Molecular mechanisms that underlie structural and functional changes at the postsynaptic membrane during synaptic plasticity. Prog Neurobiol 55:611–640.
- Whitaker-Azmitia PM, Wingate M, Borella A, Gerlai R, Roder J, Azmitia EC. 1997. Transgenic mice overexpressing the neurotrophic factor S-100  $\beta$  show neuronal cytoskeletal and behavioral signs of altered aging processes: implications for Alzheimer's disease and Down's syndrome. Brain Res 776:51–60.
- White F, Nicoll JA, Roses AD, Horsburgh K. 2001. Impaired neuronal plasticity in transgenic mice expressing human apolipoprotein E4 compared to E3 in a model of entorhinal cortex lesion. Neurobiol Dis 8:611–625.
- Wilkinson D. 2001. Drugs for treatment of Alzheimer's disease. Int J Clin Pract 55:129–134.
- Winckler B, Forscher P, Mellman I. 1999. A diffusion barrier maintains distribution of membrane proteins in polarized neurons. Nature 397:698– 701.

- Wise PM, Dubal DB, Wilson ME, Rau SW, Bottner M. 2001a. Mini review: neuroprotective effects of estrogen-new insights into mechanisms of action. Endocrinology 142:969–973.
- Wise PM, Dubal DB, Wilson ME, Rau SW, Liu Y. 2001b. Estrogens: trophic and protective factors in the adult brain. Front Neuroendocrinol 22:33–66.
- Wolozin B. 2001. A fluid connection: cholesterol and Aβ. Proc Natl Acad Sci USA 98:5371–5373.
- Wolozin B, Kellman W, Ruosseau P, Celesia GG, Siegel G. 2000. Decreased prevalence of Alzheimer disease associated with 3-hydroxy-3methyglutaryl coenzyme A reductase inhibitors. Arch Neurol 57:1439– 1443.
- Wong M, Moss RL. 1992. Long-term and short-term electrophysiological effects of estrogen on the synaptic properties of hippocampal CA1 neurons. J Neurosci 12:3217–3225.
- Wong PC, Zheng H, Chen H, Becher MW, Sirinathsinghji DJ, Trumbauer ME, Chen HY, Price DL, Van der Ploeg LH, Sisodia SS. 1997. Presenilin 1 is required for Notch1 and DII1 expression in the paraxial mesoderm. Nature 387:288–292.
- Wood WG, Schroeder F, Avdulov NA, Chochina SV, Igbavboa U. 1999. Recent advances in brain cholesterol dynamics: transport, domains, and Alzheimer's disease. Lipids 34:225–234.
- Woolley CS, McEwen BS. 1992. Estradiol mediates fluctuation in hippocampal synapse density during the estrous cycle in the adult rat. J Neurosci 12:2549–2554.
- Woolley CS, McEwen BS. 1993. Roles of estradiol and progesterone in regulation of hippocampal dendritic spine density during the estrous cycle in the rat. J Comp Neurol 336:293–306.
- Woolley CS, Wenzel HJ, Schwartzkroin PA. 1996. Estradiol increases the frequency of multiple synapse boutons in the hippocampal CA1 region of the adult female rat. J Comp Neurol 373:108–117.
- Woolley CS. 1998. Estrogen-mediated structural and functional synaptic plasticity in the female rat hippocampus. Horm Behav 34:140–148.
- Worley PF, Bhat RV, Baraban JM, Erickson CA, McNaughton BL, Barnes CA. 1993. Thresholds for synaptic activation of transcription factors in hippocampus: correlation with long-term enhancement. J Neurosci 13: 4776–4786.
- Wu W, Liuzzi FJ, Schinco FP, Depto AS, Li Y, Mong JA, Dawson TM, Snyder SH. 1994. Neuronal nitric oxide synthase is induced in spinal neurons by traumatic injury. Neuroscience 61:719–726.
- Xiang Z-M, Moskowitz SI, Hrabetova S, Bergold PJ. 2000. Long-term cultures of adult rat hippocampus. J Neurosci Methods 98:145–154.
- Xiao SF, Yan HQ, Yao PF, Cerebrolysin Study Group. 2000. Efficacy of FPF 1070 (cerebrolysin) in patients with Alzheimer's disease. A multicenter, randomized, double-blind, placebo controlled trial. Clin Drug Invest 19:43–53.
- Xu P-T, Schmechel D, Rothrock T, Qiu H-L, Popko B, Sullavan P, Maeda N, Saunders A, Roses AD, Gilbert JR. 1996. Human apolipoprotein E2, E3 and E4 isoform specific transgenic mice: human-like pattern of glial and neuronal immunoreactivity in central nervous system not observed in wild type mice. Neurobiol Dis 3:229–245.
- Xu P-T, Gilbert JR, Qiu HL, Rothrock-Christian T, Settles DL, Roses AD, Schmechel DE. 1998. Regionally specific neuronal expression of human APOE gene in transgenic mice. Neurosci Lett 246:65–68.
- Xu P-T, Gilbert JR, Qiu H-L, Ervin J, Rothrock-Christian TR, Hulette C, Schmechel DE. 1999. Specific regional transcription of apolipoprotein E in human brain neurons. Am J Pathol 154:601–611.
- Yaffe K, Cauley J, Sands L, Browner W. 1997. Apolipoprotein E phenotype and cognitive decline in a prospective study of elderly community women. Arch Neurol 54:1110–1114.
- Yaffe K, Haan M, Byers A, Tangen C. 2000. Estrogen use, APOE, and cognitive decline. Evidence of gene-environment interaction. Neurology 54:1949–1953.

- Yaffe K, Barrett-Connor E, Lin F, Grady D. 2002. Serum lipoprotein levels, statin use, and cognitive function in older women. Arch Neurol 59:378–384.
- Yamada K, Tanaka T, Han D, Senzaki K, Kameyama T, Nabeshima T. 1999. Protective effects of idebenone and  $\alpha$ -tocopherol on  $\beta$ -amyloid-(1–42)-induced learning and memory deficits in rats: implication of oxidative stress in  $\beta$ -amyloid-induced neurotoxicity in vivo. Eur J Neurosci 11:83–90.
- Yamaguchi K, Sasano A, Urakami T, Tsuji T, Kondo K. 1993. Stimulation of nerve growth factor production by pyrroloquinoline quinone and its derivatives in vitro and in vivo. Biosci Biotechnol Biochem 57:1231– 1233.
- Yan XX, Garey LJ, Jen LS. 1996. Prenatal development of NADPHdiaphorase-reactive neurons in human frontal cortex. Cereb Cortex 6:737–745.
- Yankova M, Hart SA, Woolley CS. 2001. Estrogen increases synaptic connectivity between single presynaptic inputs and multiple postsynaptic CA1 pyramidal cells: a serial electron-microscopic study. Proc Natl Acad Sci USA 98:3525–3530.
- Yu WH. 1994. Nitric oxide synthase in motor neurons after axotomy. J Histochem Cytochem 42:451-457.
- Yun HY, Gonzalez-Zulueta M, Dawson VL, Dawson TM. 1998. Nitric oxide mediates N-methyl-D-aspartate receptor-induced activation of p21ras. Proc Natl Acad Sci USA 95:5773–5778.
- Zaman SH, Parent A, Laskey A, Lee MK, Borchelt DR, Sisodia SS, Malinow R. 2000. Enhanced synaptic potentiation in transgenic mice expressing presenilin 1 familial Alzheimer's disease mutation is normalized with a benzodiazepine. Neurobiol Dis 7:54–63.

Zhan SS, Beyreuther K, Schmitt HP. 1993. Quantitative assessment of the

synaptophysin immuno-reactivity of the cortical neuropil in various neurodegenerative disorders with dementia. Dementia 4:66–74.

- Zhang C, Lambert MP, Bunch C, Barber K, Wade WS, Krafft GA, Klein WL. 1994. Focal adhesion kinase expressed by nerve cell lines shows increased tyrosine phosphorylation in response to Alzheimer's A beta peptide. J Biol Chem 269:25247–25250.
- Zhang Z, Hartmann H, Do VM, Abramowski D, Sturchler-Pierrat C, Staufenbiel M, Sommer B, van de Wetering M, Clevers H, Saftig P, De Strooper B, He X, Yankner BA. 1998. Destabilization of β-catenin by mutations in presenilin-1 potentiates neuronal apoptosis. Nature 395: 698–702.
- Zhou W, Raisman G, Zhou C. 1998. Transplanted embryonic entorhinal neurons make functional synapses in adult host hippocampus. Brain Res 788:202–206.
- Zhou J, Zhang F, Zhang Y. 2000. Corticosterone inhibits generation of long-term potentiation in rat hippocampal slice: involvement of brainderived neurotrophic factor. Brain Res 885:182–191.
- Zhu XO, Waite PM. 1998. Cholinergic depletion reduces plasticity of barrel field cortex. Cereb Cortex 8:63–72.
- Zimmer J, Gahwiler BH. 1987. Growth of hippocampal mossy fibers: A lesion and coculture study of organotypic slice cultures. J Comp Neurol 264:1–13.
- Zimmer J, Kristensen BW, Jakobsen B, Gramsbergen JB, Noraberg J. 1999. Organotypic brain slice cultures used for neurotoxicity testing. Soc Neurosci Abstr 25:442.16.
- Zubenko GS, Kopp U, Seto T, Firestone LL. 1999. Platelet membrane fluidity individuals at risk for Alzheimer's disease: a comparison of results from fluorescence spectroscopy and electron spin resonance spectroscopy. Psychopharmacology (Berl) 145:175–180.